# University of Miami Scholarly Repository

**Open Access Dissertations** 

**Electronic Theses and Dissertations** 

2013-07-11

# Partial Identification of Average Treatment Effects in Program Evaluation: Theory and Applications

Xuan Chen University of Miami, buer0615@gmail.com

Follow this and additional works at: https://scholarlyrepository.miami.edu/oa\_dissertations

#### **Recommended** Citation

Chen, Xuan, "Partial Identification of Average Treatment Effects in Program Evaluation: Theory and Applications" (2013). *Open Access Dissertations*. 1050. https://scholarlyrepository.miami.edu/oa\_dissertations/1050

This Open access is brought to you for free and open access by the Electronic Theses and Dissertations at Scholarly Repository. It has been accepted for inclusion in Open Access Dissertations by an authorized administrator of Scholarly Repository. For more information, please contact repository.library@miami.edu.

# UNIVERSITY OF MIAMI

# PARTIAL IDENTIFICATION OF AVERAGE TREATMENT EFFECTS IN PROGRAM EVALUATION: THEORY AND APPLICATIONS

By

Xuan Chen

A DISSERTATION

Submitted to the Faculty of the University of Miami in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Coral Gables, Florida

August 2013

©2013 Xuan Chen All Rights Reserved

### UNIVERSITY OF MIAMI

# A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy

# PARTIAL IDENTIFICATION OF AVERAGE TREATMENT EFFECTS IN PROGRAM EVALUATION: THEORY AND APPLICATIONS

Xuan Chen

Approved:

Carlos A. Flores, Ph.D. Assistant Professor of Economics

M. Brian Blake, Ph.D. Dean of the Graduate School

Christopher F. Parmeter, Ph.D. Associate Professor of Economics Laura Giuliano, Ph.D. Associate Professor of Economics

C. Hendricks Brown, Ph.D. Professor of Psychiatry and Behavioral Sciences Northwestern University

# CHEN, XUAN. <u>Partial Identification of Average Treatment Effects</u> in Program Evaluation: Theory and Applications

(Ph.D., Economics) (August 2013)

Abstract of a dissertation at the University of Miami.

Dissertation supervised by Professor Carlos A. Flores. No. of pages in text. (116)

There has been a recent increase on research focusing on partial identification of average treatment effects in the program evaluation literature. In contrast with traditional point identification, partial identification approaches derive bounds on parameters of interest based on relatively weak assumptions. Thus, they deliver more credible results in empirical applications. This dissertation extends Instrumental Variable (IV) methods in the program evaluation literature by partially identifying treatment effects of interest when evaluating a program or intervention.

An influential approach for studying causality within the IV framework was developed by Imbens and Angrist (1994) and Angrist, Imbens and Rubin (1996). They show that, when allowing for heterogeneous effects, IV estimators point identify the local average treatment effect (*LATE*) for compliers, whose treatment status is affected by the instrument. This dissertation advances the current IV literature in two important ways. First, inspired by the common criticism that *LATE* lacks external validity, this dissertation derives sharp nonparametric bounds for population average treatment effects (*ATE*) within the *LATE* framework. Second, the dissertation extends the *LATE* framework to bound treatment effects in the presence of both sample selection and noncompliance. Even when employing randomized experiments to evaluate programs -- as is now

common in economics and other social science fields -- assessing the impact of the treatment on outcomes of interest is often made difficult by those two critical identification problems. The sample selection issue arises when outcomes of interest are only observed for a selected group. The noncompliance problem appears because some treatment group individuals do not receive the treatment while some control individuals do. The dissertation addresses both of these identification problems simultaneously and derives nonparametric bounds for average treatment effects within a principal stratification framework. More generally, these bounds can be employed in settings where two identification problems are present and there is a valid instrument to address one of them. The bounds derived in this dissertation are based on two sets of relatively weak assumptions: monotonicity assumptions on potential outcomes within specified subpopulations, and mean dominance assumptions across subpopulations.

The dissertation employs the derived bounds to evaluate the effectiveness of the Job Corps (JC) program, which is the largest federally-funded job training program for disadvantaged youth in the United States, with the focus on labor market outcomes and welfare dependence. The dissertation uses data from an experimental evaluation of JC. Individuals were randomly assigned to a treatment group (whose members were allowed to enroll in JC) or to a control group (whose members were denied access to JC for three years). However, there was noncompliance: some individuals who were assigned to participate in JC did not enroll, while some individuals assigned to the control group did. The dissertation addresses this noncompliance issue using random assignment as an IV for enrollment into JC. Concentrating on the population *ATE*, JC enrollment increases weekly earnings by at least \$24.61 and employment by at least 4.3 percentage points four

years after randomization, and decreases yearly dependence on public welfare benefits by at least \$84.29. These bounds are significantly narrower than the ones derived in the current IV literature. The dissertation also evaluates the effect of JC on wages, which are observed only for those who are employed. Hence, the sample selection issue has to be addressed when evaluating this effect. In the presence of sample selection and noncompliance, the average treatment effect of JC enrollment on wages for the alwaysemployed compliers, who would comply with their assigned treatment and who would be always employed regardless of their assignment statuses, is bounded between 5.7 percent and 13.9 percent four years after random assignment. The results suggest greater positive average effects of JC on wages than those found in the literature evaluating JC without adjusting for noncompliance.

The dissertation closes by pointing out that a similar analytic strategy to the one used in this dissertation can be used to address other problems, for example, to bound the *ATE* when the instrument does not satisfy the exclusion restriction, and to derive bounds on the part of the effect of a treatment on an outcome that works through a given mechanism (i.e., direct or net effects) in the presence of one identification issue (e.g., noncompliance). To Deng

#### ACKNOWLEDGEMENTS

I owe my deepest gratitude to my advisor, Professor Carlos A. Flores. This dissertation would not have been possible without his sharp insight and excellent guidance. His generous help and extreme patience made me progress in my area of specialization. He always supported me throughout my graduate studies. His kindness and encouragement also helped me through the hard times of the job market.

I would also like to express my gratitude to the other three members of my dissertation committee, Christopher F. Parmeter, Laura Giuliano, and C. Hendricks Brown, for their sustained interest and valuable suggestions in the process of completing this dissertation.

I also want to express my sincere appreciation to Professors Alfonso Flores-Lagunes and Oscar Mitnik for their encouragement and generous help at various stages of my study.

Finally, I would like to extend my thanks to participants at the Seminar Series at the University of Miami for their useful comments and constant support.

# **TABLE OF CONTENTS**

LIST (	LIST OF FIGURES vi			
LIST (	OF TABLES	viii		
Chapte	ers			
1	INTRODUCTION	1		
2	LITERATURE REVIEW	8		
_	2.1 Partial Identification in the IV Literature	8		
	2.2 Partial Identification Addressing Sample Selection	13		
3	THEORY: BOUNDS ON THE PARAMETERS OF INTEREST	19		
	3.1 The <i>LATE</i> Framework	19		
	3.2 Bounds on the Population <i>ATE</i>	20		
	3.2.1 Bounds under Monotonicity	22		
	3.2.2 Bounds under Mean Dominance	23		
	3.3 Bounds Addressing Sample Selection and Noncompliance	28		
	3.3.1 Setup, Principal Strata, and Parameter of Interest	28		
	3.3.2 Bounds under Monotonicity	32		
	3.3.3 Bounds under Mean Dominance	38		
	3.3.4 Remarks on Assumptions	40		
	3.4 Estimation and Inference	45		
4	APPLICATION: EVALUATION OF JOB CORPS	48		
	4.1 Job Corps and the National Job Corps Study	48		
	4.2 The Population <i>ATE</i> of JC Enrollment	50		
	4.2.1 Specific Sample	50		
	4.2.2 Assessment of Assumptions	51		
	4.2.3 Empirical Results	56		
	4.3 The Wage Effect of JC Enrollment	63		
	4.3.1 Specific Sample	63		
	4.3.2 Assessment of Assumptions	66		
	4.3.3 Empirical Results	69		
	4.3.4 Simulation Exercises	77		
5	CONCLUSION	88		
Biblio	ogranhy	93		
210110	rJ ·····	22		
Appen	ndix A Derivation of the Bounds	99		
A.	1 Derivation of the Bounds in Section 3.2	99		

A.2	Derivation of the Bounds in Section 3.3	101
	A.2.1 Proof of Proposition 3.1	101
	A.2.2 Proof of Proposition 3.2	108
Appendix	B Procedure for Estimation and Inference	111
Appendix	C GMM Moment Functions	114
C.1	GMM Moment Functions for Assumption 2.3	114
C.2	GMM Moment Functions for Assumption 3.6	114
C.3	Baseline Characteristic Estimates	116

# LIST OF FIGURES

4.1	Bounds When Monotonicity Fails in Simulation 1	80
4.2	Percentage of Times True Effect Falls within the Bounds in Simulation 1	81
4.3	Bounds When Mean Dominance Fails in Simulation 2	82
4.4	Percentage of Times True Effect Falls within the Bounds in Simulation 2	82
4.5	Bounds When Both Assumptions Fail with $\pi_{cNE}$ =.05 in Simulation 3	84
4.6	Percentage of Times True Effect Falls within the Bounds in Simulation 3	84
4.7	Bounds When Both Assumptions Fail with $\pi_{cNE}$ =.10 in Simulation 4	85
4.8	Percentage of Times True Effect Falls within the Bounds in Simulation 4	85
4.9	Bounds When Both Assumptions Fail with $\pi_{cNE}$ =.15 in Simulation 5	86
4.10	Percentage of Times True Effect Falls within the Bounds in Simulation 5	86

# LIST OF TABLES

3.1	Observed Groups and Principal Strata	34
4.1	Summary Statistics of Baseline Variables (Population ATE of JC)	52
4.2	Point Identified Average Outcomes after Random Assignment	54
4.3	Average Baseline Characteristics in the Sample for Labor Market Outcomes	57
4.4	Average Baseline Characteristics in the Sample for Public Assistance Benefits	58
4.5	Bounds on the Population Average Treatment Effects	59
4.6	Summary Statistics of Baseline Variables (Wage Effect of JC)	65
4.7	Summary Statistics of Post-treatment Variables by Random Assignment	66
4.8	Bounds for the ITT Effects on ln(wage) at Week 208	71
4.9	Bounds for the Effects of JC on ln(wage) for the <i>cEE</i> Stratum	73
4.10	Bounds on the Effects of JC on ln(wage) for <i>cEE</i> Adjusting for Non-Response	78
C.1	Average Baseline Characteristics for the <i>cEE</i> and <i>cNE</i> Strata	116

#### CHAPTER 1

#### INTRODUCTION

There has been a recent increase on research focusing on partial identification of average treatment effects in the program evaluation literature. In contrast with traditional point identification, partial identification approaches derive bounds on parameters of interest instead of estimating a single value. Their main advantage is their dependence on weaker assumptions. Thus, partial identification approaches deliver more reliable results in empirical applications. This dissertation extends instrumental variable (IV) methods in the program evaluation literature by partially identifying treatment effects of interest when evaluating a program or intervention.

This dissertation is motivated by an influential approach for studying causality in IV contexts. Imbens and Angrist (1994) and Angrist, Imbens and Rubin (1996) (hereafter IA and AIR, respectively) show that, when allowing for heterogeneous effects, IV estimators point identify the local average treatment effect (*LATE*) for compliers, whose treatment status is affected by the instrument. However, a common criticism of their approach is the focus on the effect for a subpopulation, which cannot be identified from the population of interest (e.g., Heckman, 1996; Robins and Greenland, 1996; Deaton, 2010; Heckman and Urzua, 2010). Policy makers may be more interested in the population average treatment effect (*ATE*). AIR (1996) and Imbens (2009) responded to the criticism that the discussion is limited to compliers because it is the only subpopulation about which the data are directly informative.

The dissertation advances the current IV literature in two import ways. First, it derives sharp nonparametric bounds for the population *ATE* within the *LATE* framework. It improves the bounds on the ATE in the current IV literature by combining two sets of assumptions that may be useful in practice. The first is monotonicity in the treatment of the average outcomes of subpopulations. This assumption infers the sign of average treatment effects for other subpopulations from that for compliers. In contrast to the existing literature (e.g., Manski and Pepper, 2000; Bhattacharya, Shaikh and Vytlacil, 2008, hereafter BSV; Shaikh and Vytlacil, 2011), monotonicity is imposed on the average outcomes within the same subpopulation rather than on the individuals' outcomes. This makes the assumption more plausible in practice by allowing some individuals to experience a treatment effect that has the opposite sign to the LATE. The second set of assumptions is mean dominance that compares average potential outcomes across different subpopulations. Unlike the monotonicity, which informs the unobserved outcomes in the *LATE* framework by inferring the sign of treatment effects, the mean dominance infers the unobserved terms from the identified average outcomes across subpopulations. Different from the literature on partially identifying the ATE within an IV, some of the bounds in the present research do not require a bounded-outcome assumption once combining the monotonicity and mean dominance assumptions. Moreover, the direction of the mean dominance can be informed by comparing average baseline characteristics of strata that are likely to be highly correlated with the outcome of interest, and these average characteristics can be obtained by estimating an overidentified nonparametric GMM problem.

The second way this dissertation advances the current IV literature is to derive nonparametric bounds for treatment effects in the presence of both sample selection and noncompliance. Randomized experiments are now commonly used to evaluate programs or interventions in economics and other social science fields. Despite the fact that randomized experiments are reliable approaches to establishing causality, assessing treatment effects of interest in randomized experiments is often made complicated by two critical identification issues: sample selection and noncompliance with the assigned treatment. The sample selection issue arises when outcomes are only observed for a selected group. For example, future health status is only observed for individuals who are alive at the time of following waves of surveys. The noncompliance problem appears because individuals can still decide whether or not to take the treatment in most of randomized experiments, especially in the ones with encouragement designs. As a result, it is common that in social experiments some treatment group individuals do not take the treatment, while some control individuals do.

The dissertation extends the partial identification results in Zhang, Rubin, and Mealli (2008) (hereafter ZRM) and Lee (2009), who construct bounds in the presence of sample selection, to also account for noncompliance using the *LATE* framework proposed by IA (1994) and AIR (1996). The *LATE* framework is a special case of principal stratification, which partitions the population based on the joint potential values of post-treatment variables under the two treatment assignment arms. Within the principal stratification framework, this dissertation derives nonparametric bounds for the always-selected compliers, who would comply with their assigned treatment and whose outcomes would be always observable regardless of treatment assignment. Analogous to the cases analyzed in IA (1994) and AIR (1996), ZRM (2008) and Lee (2009), among others, this is the only group of individuals whose outcomes can be observed in both treatment receipt arms in settings where both sample selection and noncompliance are present.

Additional assumptions are necessary to derive bounds on treatment effects for other subpopulations.

Principal stratification has often been used to address a single post-treatment complication. While there are a few papers that employ principal stratification to point identify treatment effects in the presence of more than one complication, it is my understanding that this is one of the first studies deriving bounds for treatment effects within this framework accounting for more than one identification problem. More generally, these bounds can be employed in settings where two identification problems are present and there is a valid instrument to address one of them. Some of these complications may include sample selection, noncompliance, endogeneity of the treatment variable, missing outcomes, among others. For example, when assessing the effect of military service on future health using the Vietnam-era draft lottery as an instrument to address endogeneity (e.g., Angrist, Chen, and Frandsen, 2009), the results could be used to bound the average effect on those who enrolled in the military because of the draft lottery (compliers) and who were alive when the outcome was measured regardless of their veteran status.

This dissertation employs the derived bounds to evaluate the effectiveness of the Job Corps (JC) program, which is the largest and most comprehensive federally-funded job training program for disadvantaged youth in the United States. It provides academic, vocational, and social skills training, among many other services, at over 120 centers throughout the country. Assessing the effect of this and other programs is of great importance to policy makers (e.g., Tennessee class size project STAR, Greater Avenues for Independence (GAIN) program, Tax Deferred Account (TDA) retirement plan). This dissertation uses data from the National Job Corps Study (NJCS), which is a randomized social experiment undertaken in the mid-to-late nineties to evaluate the effectiveness of JC. Individuals were randomly assigned to a treatment group (whose members were allowed to enroll in JC) or to a control group (whose members were denied access to JC for three years). However, there was noncompliance: some individuals in the treatment group did not enroll, while some individuals in the control group did.

Previous studies evaluating the JC program (Schochet, Burghardt, and Glazerman, 2001; Schochet, Burghardt, and McConnell, 2008; Lee, 2009; Flores-Lagunes, Gonzalez, and Neumann, 2010; Flores et al., 2012; Blanco, Flores, and Flores-Lagunes, 2012) usually focus on intention-to-treat (ITT) effects. An ITT effect compares potential outcomes according to the assigned treatment and ignores possible noncompliance, and thus would dilute the effect of actual JC enrollment. Schochet, Burghardt, and Glazerman (2001) and Schochet, Burghardt, and McConnell (2008) also address noncompliance by using IV estimators, hence focusing on the *LATE* for compliers. Using random assignment as an IV, this dissertation addresses this noncompliance issue and looks at the population ATEs of actual JC enrollment on participants' labor market outcomes and welfare. It also examines wage effects of JC enrollment by focusing on a specified subpopulation (i.e., the always-employed compliers). Thus, the dissertation also contributes to the empirical literature on the evaluation of the JC program by providing credible bounds based on relatively weak assumptions for treatment effects of interest other than *LATE* and the *ITT* effect.

Focusing on the *ATE* on labor market outcomes and welfare, JC enrollment increases weekly earnings by at least \$24.61 and employment by at least 4.3 percentage points four years after randomization, and decreases yearly dependence on public wel-

fare benefits by at least \$84.29. More specifically, the preferred bounds on the ATE under the monotonicity and mean dominance assumptions are [24.61, 201.04] for weekly earnings, [.042, .163] for employment, and [-142.76, -84.29] for public benefits. These bounds are significantly narrower than the IV bounds proposed by Manski (1990), Heckman and Vytlacil (2000), Kitagawa (2009). The enhanced identification power comes from the mean dominance assumption, which is not considered in the current literature. This assumption provides new information to the bounds of the ATE by using identified average outcomes across strata. Accordingly, combining it with the monotonicity assumption sharpens the bounds. The width of the bounds is also smaller than that under both the IV and Monotone Treatment Response (MTR) assumptions of Manski and Pepper (2000), especially for public benefits. The bounds on employment are also narrower than those proposed by Balke and Pearl (1997), BSV (2008), Chesher (2010), Chiburis (2010b) and Shaikh and Vytlacil (2011) for the case of a binary outcome. The lower bounds for weekly earnings and employment are 10 percent higher than their respective *ITT* effects (22.19 and .038), while the upper bound for public benefits is equal to its *ITT* effect. The *LATE*s for compliers on the three outcomes also fall within these bounds.

The dissertation also evaluates the effect of JC enrollment on wages, which are observed only for those who are employed. Ignoring this sample selection issue would give biased results. Hence, it is necessary and important to address sample selection when evaluating wage effects. In the presence of sample selection and noncompliance, the bounds on the wage effect of JC enrollment are derived for the always-employed compliers, who would comply with their assigned treatment and who would always be employed regardless of assignment status. This is the only stratum for which wages are observed for individuals who enrolled and for individuals who did not enroll in the JC program. It is also the largest stratum in the sample, accounting for about 40 percent of the population. The wage effect of JC enrollment for the always-employed compliers is between 5.7 percent and 13.9 percent four years after random assignment, and between 7.7 and 17.5 percent for Non-Hispanics. The results suggest greater positive average effects of JC on wages than those found without adjusting for noncompliance in Lee (2009) and Blanco, Flores, and Flores-Lagunes (2012). This evidence suggests that the JC training program has positive effects not only on the employability of its participants but also on their wages, implying that JC participation is likely to increase their human capital. By mimicking the characteristics of the sample, this section also presents simulation exercises done to analyze the sensitivity of the empirical results to violations of the two main assumptions employed (monotonicity and mean dominance). The simulation results suggest that the estimated bounds are robust to small departures from these assumptions.

The remainder of the dissertation is organized as follows. Chapter 2 presents a review of the partial identification literature. Chapter 3 presents the econometric framework and the partial identification results for the parameters of interest, and then discusses estimation and inference. Chapter 4 briefly describes the JC program and the NJCS, and empirically analyzes the effects of JC enrollment on participants' labor market outcomes and welfare dependence. Chapter 5 concludes.

#### CHAPTER 2

#### LITERATURE REVIEW

# 2.1 Partial Identification in the IV Literature

Instrumental variable (IV) approaches have been widely used in the literature of program evaluation due to their high internal validity. IA (1994) and AIR (1996) develop an influential approach for studying causality within the IV framework. Using a randomized instrument, they partition the population into always-takers, never-takers, compliers, and defiers according to the joint potential values of treatment status. Always-takers (never-takers) are individuals who would always (never) take the treatment irrespective of instrument status. Compliers behave consistently with the assigned instrument, while defiers do the opposite of the assigned instrument. Under the monotonicity of the treatment in the instrument, defiers are ruled out. In the absence of strong homogeneity or distributional assumptions, data never reveals information on the outcome of always-takers under the inactive treatment or the outcome of never-takers under the active treatment. As a result, IV estimators point identify the local average treatment effect (*LATE*) for compliers in heterogenous treatment effect models.

A common criticism of their approach, however, is the focus on the effect for a subpopulation (e.g., Deaton, 2010; Heckman and Urzua, 2010). Heckman (1996) and Robins and Greenland (1996) state that the *LATE* is defined for a latent subpopulation in the sense that compliers cannot be identified from the population of interest. The latter suggests that attention should be on the population average treatment effect (*ATE*). AIR (1996) and Imbens (2009) respond that one may also be interested in averages

for the entire population from the point of view of policy makers, but they stress that compliers is the only subpopulation about which the data are directly informative, and that extension to treatment effects for other subpopulations has to be extrapolations in IV contexts.

Point identification of the population treatment effects by the IV methods usually requires parametric or structural assumptions. Heckman (2010) proposes a method to nonparametric identify the ATE, but the instrument in his approach is required that is strong enough to drive the probability of being treated from zero to one, which is hard to satisfy in practice. In contrast to traditional point identification, Manski (1990) pioneered partial identification of the population ATE under the assumption of mean independence of the instrument.

There has been a growing literature on partial identification of the ATE with IV methods since Manski (1990). One strand of this literature endeavors to improve Manski's bounds by assuming different versions of monotonicity of the outcome. Manski and Pepper (2000) introduce the assumptions of monotonicity of the treatment response (MTR) and the monotonicity of the treatment selection (MTS). Combined with the mean independence assumption, Chiburis (2010a) derives the bounds for the ATE under both MTR and MTS assumptions without specifying the direction of the monotonicity a priori. Instead of the monotonicity assumptions employed in the above papers, another strand of the literature assumes structural models on the treatment or the outcome. Under the statistical independence assumption of the instrument, Heckman and Vytlacil (2000) impose a threshold crossing model with a separable error on the treatment. Focusing on a binary outcome, Shaikh and Vytlacil (2011) impose threshold crossing model.

els on both the treatment and the outcome, while Chiburis (2010b) considers a threshold crossing model on the outcome. Rather than assuming the threshold crossing model with separable errors, Chesher (2010) imposes a non-separable structural model on the outcome and assumes the structural function is weakly increasing in the non-separable error.

Comparison of identification power among these assumptions are also discussed in the existing literature on partial identification with IV methods. First, the monotonicity assumption on the treatment (e.g., Balke and Pearl, 1997; Huber and Mellace, 2010) and the structural model assumptions on the treatment (e.g., Heckman and Vytlacil, 2000) do not improve Manski's bounds derived under the mean independence assumption. Second, monotonicity assumptions on the outcome (e.g., Manski and Pepper, 2000) and the structural model assumptions on the outcome do improve Manski's bounds (e.g., BSV, 2008; Chiburis, 2010a; 2010b; Chesher 2010; Shaikh and Vytlacil, 2011). Third, because the bounds for the *ATE* involve counterfactual potential outcomes, partial identification with IV methods usually requires bounded support of the outcome. This might also be the reason why quite a few papers focus on binary outcomes (e.g., Balke and Pearl, 1997; BSV, 2008; Hahn, 2010; Chiburis, 2010b; Shaikh and Vytlacil, 2011).

It's worth noting that for a binary dependent variable, the monotonicity assumptions and the structural model assumptions are equivalent. Vytlacil (2002) shows the equivalence between the monotonicity assumption and the threshold crossing model on the treatment. Machado et al. (2011) notice the equivalence between the MTR assumption and the threshold crossing model on the outcome. In the absence of covariates, Chiburis (2010b) observe the equivalence between the threshold crossing model with a separable error and the non-separable structural function being weakly increasing in the non-separable error. BSV (2008) show that in the absence of covariates, the bounds for a binary outcome under MTR and the mean independence assumptions are equal to the ones derived from the threshold crossing models on both the treatment and the outcome. Chiburis (2010b) noticed that his bounds obtained by imposing the threshold crossing model on the outcome are equal to the ones under MTR and the mean independence assumptions.

This dissertation improves Manski's nonparametric bounds on the population ATEby extending the work of IA (1994) and AIR (1996). The setup of a binary treatment and a binary instrument is used in most of the program evaluation literature (e.g., Imbens and Wooldridge, 2009) and common in empirical applications (e.g., Angrist, 1990; Oreopoulous, 2006). And the LATE framework allows nonparametric identification of the ATE within the context of heterogeneous treatment effects. This dissertation adds to the literature by considering two different sets of assumptions. The first is monotonicity in the treatment of the average outcomes of subpopulations defined by the joint potential values of the treatment status under each value of the instrument. As in BSV (2008) and Shaikh and Vytlacil (2011), prior knowledge about the direction of the monotonicity is not required. This assumption infers the sign of average treatment effects for other subpopulations from that for compliers. In contrast to the existing literature (e.g., Manski and Pepper, 2000; BSV, 2008; Shaikh and Vytlacil, 2011), monotonicity is imposed on the average outcomes of the strata rather than on the individuals' outcomes. This makes the assumption more plausible in practice by allowing some individuals to experience a treatment effect that has the opposite sign to the ATE. The second set of assumptions is

mean dominance that compares average potential outcomes across different subpopulations. Assumptions similar to the mean dominance have been shown to have significant identifying power in other settings (e.g., ZRM, 2008; Flores and Flores-Lagunes, 2010). Different from the monotonicity, which informs the unobserved outcomes by inferring the sign of treatment effects, the mean dominance infers the unobserved terms from the identified average outcomes across subpopulations. Once combining the two sets of assumptions, the sharp bounds of the ATE in the present study are significantly narrower than the ones obtained in the current IV literature, and some of the bounds do not require a bounded-outcome assumption. Moreover, the directions of the mean dominance can be informed by comparing average baseline characteristics of strata that are likely to be highly correlated with the outcome of interest. These average characteristics are estimated by solving an overidentified nonparametric GMM problem.

A recent paper by Huber and Mellace (2010) also derives bounds on the *ATE* within the IV framework. The main difference between this research and theirs is that the monotonicity assumption is imposed on the average outcomes of the strata, which results in narrower bounds and can be justified by economic theory in many applications. Also, a priori direction of the monotonicity is avoided, while its direction can be inferred from data. In addition, the mean dominance assumptions not only differ from theirs, but the direction of the mean dominance can be informed by comparing the average baseline characteristics across strata, which are estimated by solving an overidentified nonparametric GMM problem.

#### 2.2 Partial Identification Addressing Sample Selection

One of the leading examples of sample selection in the program evaluation literature is to evaluate the effect of a training program on participants' wages. The sample selection issue arises from the fact that wages are only observed for those who are employed, with the employment decision itself being potentially affected by the program. Assessing the effect of training programs on wages is of great importance to policy makers. Most of the econometric evaluations of training programs, however, focus on the impact on total earnings, which are the product of the hourly wage and the hours worked. As discussed by Lee (2009), focusing only on total earnings fails to answer the relevant question of whether the programs lead to an increase in participants' wages (e.g., through human capital accumulation), or to an increase in the probability of being employed (e.g., through counseling and job search assistance services) without any increase in wages.

Standard approaches for point identification of treatment effects in the presence of sample selection require strong parametric assumptions or the availability of a valid instrument (e.g., Heckman, 1979). In settings where an instrument is unavailable, an alternative strategy is to partially identify the effects under relatively mild assumptions (Zhang and Rubin, 2003; ZRM, 2008; Imai, 2008; Lee, 2009; Lechner and Melly, 2010; Huber and Mellace, 2010). Part of this literature uses principal stratification (Frangakis and Rubin, 2002), which provides a framework for studying causal effects when controlling for a variable that has been affected by the treatment (in this example, the employment decision). The basic idea behind principal stratification is to compare individuals within common principal strata (subpopulations of individuals who share the

same potential values of the employment variable under both treatment arms). Since membership to a particular principal stratum is not affected by treatment assignment, sample selection is not an issue within the principal strata, and the estimated effects are causal effects.

ZRM (2008) and Lee (2009) derive bounds for the average effect of a training program on wages for a particular stratum: the "always-employed" (those individuals who would be employed whether or not they were assigned to enroll in the training program). They focus on this stratum because it is the only one for which the individuals' outcomes are observed under both treatment assignments. Following Zhang and Rubin (2003), ZRM (2008) consider two assumptions and derive bounds for this effect under each assumption and when both are imposed. The first assumption is a monotonicity assumption on the effect of the treatment (training program) on the selection (employment), and the second is a stochastic dominance assumption comparing the potential outcomes of the always-employed to those of other strata. Lee (2009) uses an alternative approach to that in ZRM (2008) to derive bounds under the monotonicity assumption. Importantly, the bounds derived in these papers do not impose the assumption that the support of the outcome is bounded. Lee (2009) uses his bounds to evaluate the wage effects of JC. More recently, Blanco, Flores, and Flores-Lagunes (2012) employ the bounds used by ZRM (2008) and Lee (2009), and their extension to quantile treatment effects by Imai (2008) to study the wage effects of JC for different demographic groups without adjusting for noncompliance.

Huber and Mellace (2010) and Lechner and Melly (2010) derive bounds for subpopulations other than the always-employed. Huber and Mellace (2010) use a principal stratification approach to construct bounds on the effects for two other strata (those who would be employed only if assigned to the treatment group, and those who would be employed only if assigned to the control group), as well as for the "selected" subpopulation (those whose wages are observed and are a mixture of different strata). While their assumptions are similar to those in ZRM (2008) and Lee (2009), additional assumptions are required (e.g., bounded support of the outcome), since bounds are constructed for strata and subpopulations for which the outcome is never observed under one of the treatment states. Lechner and Melly (2010) derive bounds for mean and quantile treatment effects for the "treated and selected" subpopulation (employed individuals who received training and are also a mixture of different strata). Contrary to the previously described literature, they do not follow a principal stratification approach to derive their results. The assumptions they consider involve monotonicity assumptions on the training program's effect on employment (conditional on covariates), as well as stochastic dominance assumptions involving observed subpopulations (e.g., employed versus unemployed) rather than involving different strata. Similar to Huber and Mellace (2010), they require an outcome with bounded support to partially identify the mean effects.

The previously discussed literature, with the exception of Lechner and Melly (2010), focuses on the intention-to-treat (*ITT*) effect of being offered to participate in the training program. The *ITT* effect compares potential outcomes according to the assigned treatment and ignores possible noncompliance. The popularity of the *ITT* effect in the program evaluation literature is partly stimulated by the fact that randomized and natural experiments are now commonly used in economics and other social science fields to deliver causal effects. However, though individuals are randomly assigned to the treatment

group and the control group, it is usually the case that they can still choose whether or or not to actually take the treatment. In the case of training programs, the noncompliance problem appears when some treatment group individuals did not enroll in the program, while some of the control individuals did enroll. For example, in the data set of the NJCS employed in the dissertation, only 73 percent of the individuals assigned to the treatment group enrolled in JC, while 4 percent of the individuals assigned to the control group enrolled in JC in the four years after random assignment. The noncompliance issue dilutes the effect of actual participation in the program.

This dissertation derives nonparametric bounds for treatment effects in settings where both sample selection and noncompliance are present. It extends the partial identification results in ZRM (2008) and Lee (2009) to account for noncompliance. Thus, this part bounds the wage effect of actual *enrollment* in the program, rather than the effect of *being allowed* to enroll in the program. The approach to account for noncompliance is based on the works by IA (1994) and AIR (1996), who use the IV approach to address noncompliance in the absence of sample selection. Their approach is also a special case of principal stratification. Based on the individuals' potential compliance behavior under the two treatment assignments, they stratify the population into four strata: the so-called always-takers, never-takers, compliers, and defiers. The dissertation employs principal stratification to address the sample selection and noncompliance problems simultaneously, and derive bounds for the average effect of participating in a training program on wages for the stratum of always-employed compliers. This stratum consists of those who comply with their treatment assignment and would be employed whether or not they enrolled in the training program. Analogous to the cases analyzed

in IA (1994), AIR (1996), ZRM (2008), and Lee (2009), among others, this is the only stratum for which wages are observed for individuals who enrolled and did not enroll in the training program. In the context of analyzing the effects of JC on wages, this is also the largest stratum (about 40 percent of the population).

Principal stratification has often been used to address a single post-treatment complication. While there are a few papers that employ principal stratification to point identify treatment effects in the presence of more than one complication, it is my understanding that this is one of the first papers deriving bounds for treatment effects within this framework accounting for more than one identification problem. A particularly relevant paper in a similar setting is the one by Frumento et al. (2012), who analyze the effects of JC on employment and wages using data from the NJCS. They perform a likelihood-based analysis to simultaneously address three problems: sample selection, noncompliance and missing outcomes due to non-response. They stratify the population based on the potential values of the compliance behavior and employment status to address the noncompliance and sample selection issues, and they employ a "missing at random" assumption (Rubin, 1976) to address the missing-outcome problem.<sup>1</sup> Under some parametric assumptions, Frumento et al. (2012) point identify the effect of JC on wages for the always-employed compliers.<sup>2</sup> The thesis complements the work by Frumento et al. (2012) by constructing nonparametric bounds for the effect of JC on wages based on an alternative set of assumptions. In the empirical part, it also presents results

<sup>&</sup>lt;sup>1</sup>The missing at random assumption states that the probability that the outcome is missing for a given individual is random conditional on a set of observable characteristics.

<sup>&</sup>lt;sup>2</sup>Another assumption in Frumento et al. (2012) is that the individuals in the control group never enroll in JC, which rules out the existence of "always-takers". This assumption may not be plausible in applications in the dissertation, especially when looking at outcomes four years after random assignment.

that account for missing values due to non-response by using weights constructed by the NJCS using non-public use data that account for sample design and non-response.

#### CHAPTER 3

#### THEORY: BOUNDS ON THE PARAMETERS OF INTEREST

# **3.1 The** *LATE* **Framework**

Consider a random sample of size *n* from a population. Let  $D_i \in \{0, 1\}$  indicate whether unit *i* is treated ( $D_i = 1$ ) or not ( $D_i = 0$ ), and let  $Z_i \in \{0, 1\}$  be an instrument for treatment. Let  $D_i(1)$  and  $D_i(0)$  denote the treatment individual *i* would receive if  $Z_i = 1$ or  $Z_i = 0$ , respectively. The outcome of interest is *Y*. Denote by  $Y_i(1)$  and  $Y_i(0)$  the potential outcomes as a function of *D*, i.e., the outcomes individual *i* would experience if she received the treatment or not, respectively. Finally, let  $Y_i(z,d)$  be the potential outcome as a function of the instrument and the treatment. For each unit, econometricians observe  $\{Z_i, D_i(Z_i), Y_i(Z_i, D_i(Z_i))\}$ . The notation implicitly imposes the stable unit treatment value assumption (SUTVA) (Rubin 1978, 1980, 1990), which is common in the literature and implies that the potential outcomes for each unit are unrelated to the treatment assignment and treatment receipt of the other individuals. For the sake of simplicity, the subscript *i* is omitted unless deemed necessary for clarity. This setting has received considerable attention in the literature (e.g., AIR, 1996; BSV, 2008).

AIR (1996) partition the population into four strata based on the joint potential values of  $\{D_i(0), D_i(1)\}$ :  $\{1, 1\}, \{0, 0\}, \{0, 1\}$  and  $\{1, 0\}$ . AIR (1996) and the subsequent literature refer to these strata as always-takers (*a*), never-takers (*n*), compliers (*c*), and defiers (*d*), respectively. AIR (1996) impose the following assumptions:

**Assumption 1.1** (*Randomized Instrument*).  $\{Y(z,d),D(z)\}$  is independent of Z for all  $z, d \in \{0,1\}$ .

**Assumption 1.2** (Exclusion Restriction).  $Y_i(0,d) = Y_i(1,d) = Y_i(d), d \in \{0,1\}$  for all *i*.

Assumption 1.3 (Nonzero First Stage).  $E[D(1) - D(0)] \neq 0$ .

Assumption 1.4 (Individual-Level Monotonicity of D in Z). Either  $D_i(1) \ge D_i(0)$  for all i or  $D_i(1) \le D_i(0)$  for all i.

Assumptions 1.1 through 1.4 are standard assumptions in the IV literature (e.g., IA, 1994; AIR, 1996; Huber and Mellace, 2010; Blanco, Flores, and Flores-Lagunes, 2012). Assumption 1.4 rules out the existence of defiers (compliers) when the monotonicity is non-decreasing (non-increasing). The direction of the monotonicity can be inferred from the data given the independence of Z. Following BSV (2008), Z is ordered so that  $E[D|Z=1] \ge E[D|Z=0]$  to simplify notation.

Let  $LATE_k = E[Y(1) - Y(0)|k]$  and  $\pi_k$  denote, respectively, the local average treatment effect and the stratum proportion in the population, for stratum k, with k = a, n, c. Let  $\overline{Y}^{zd} = E[Y|Z = z, D = d]$  and  $p_{d|z} = \Pr(D = d|Z = z)$ . Under Assumptions 1.1 through 1.4, the following quantities are point identified:  $\pi_a = p_{1|0}, \pi_n = p_{0|1},$  $\pi_c = p_{1|1} - p_{1|0}, E[Y(1)|a] = \overline{Y}^{01}, E[Y(0)|n] = \overline{Y}^{10}$  and  $LATE_c = (E[Y|Z = 1] - E[Y|Z = 0])/(p_{1|1} - p_{1|0})$ . As shown in IA (1994) and AIR (1996),  $LATE_c$  is point identified for compliers whose treatment status is affected by the instrument, and equals the conventional IV estimand in the absence of covariates.

### **3.2 Bounds on the Population** *ATE*

The parameter of interest in this section is the population average treatment effect,  $ATE = E[Y_i(1) - Y_i(0)]$ . To derive the bounds, ATE is decomposed as a weighted average of the LATEs for always-takers, never-takers, and compliers:

$$ATE = LATE_a \pi_a + LATE_n \pi_n + LATE_c \pi_c \tag{3.1}$$

$$= \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + E[Y(1)|n] p_{0|1} - E[Y(0)|a] p_{1|0}, \qquad (3.2)$$

where E[Y|Z = z] = E[E[Y|Z = z, D = d]|Z = z] is used in the second equality. By equation (3.2), since Y(1) for never-takers and Y(0) for always-takers are never observed in the data, additional assumptions are needed to bound *ATE*. The most basic assumption considered in the previous literature is the bounded support of the outcome.

# Assumption 2.1 (Bounded Outcome). $Y(0), Y(1) \in [y^l, y^u]$ .

This assumption states that the potential outcomes under the two treatment arms have a bounded support. Replacing E[Y(1)|n] and E[Y(0)|a] in equation (3.2) with  $y^{l}$  and  $y^{h}$ , I obtain sharp bounds on the *ATE* under Assumptions 1.1 through 1.4, and 2.1.

**Proposition 2.1** Under Assumptions 1.1 through 1.4, and 2.1, the bounds  $LB \le ATE \le UB$  are sharp, where

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + y^l p_{0|1} - y^u p_{1|0}$$
$$UB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + y^u p_{0|1} - y^l p_{1|0}.$$

The bounds in Proposition 2.1, which is presented for reference, coincide with the IV bounds in Manski (1990), Heckman and Vytlacil (2000) and Kitagawa (2009), and with those in Huber and Mellace (2010). When the outcome is binary, these bounds also coincide with those in Balke and Pearl (1997).

### **3.2.1** Bounds under Monotonicity

In this subsection, the monotonicity assumption is introduced to improve the identification power of the bounds in Proposition 2.1.

**Assumption 2.2** (Monotonicity in D of Average Outcomes of Strata). (i) Either E[Y(1)|k]

$$\geq E[Y(0)|k] \text{ for all } k = a, n, c; \text{ or } E[Y(1)|k] \leq E[Y(0)|k] \text{ for all } k = a, n, c. \text{ (ii)}$$
$$E[Y(1) - Y(0)|c] \neq 0.$$

Assumption 2.2 requires that the *LATEs* of the three existing strata are all either non-negative or non-positive. This assumption is similar to that in BSV (2008), with the important distinction that the monotonicity is imposed on the *LATEs* rather than on the individual effects, which makes it more plausible in practice by allowing some individuals to have a treatment effect of the sign different from that of the *ATE*. Since *Z* is ordered so that  $E[D|Z = 1] \ge E[D|Z = 0]$ , the direction of the monotonicity is identified from the sign of the IV estimand (*LATE<sub>c</sub>*) under the current assumptions. The following proposition presents sharp bounds on the *ATE* under the additional Assumption 2.2.

**Proposition 2.2** Under Assumptions 1.1 through 1.4, 2.1 and 2.2, the bounds  $LB \le ATE \le UB$  are sharp, where, if E[Y|Z=1] - E[Y|Z=0] > 0,

$$LB = E[Y|Z=1] - E[Y|Z=0]$$
$$UB = \overline{Y}^{11}p_{1|1} - \overline{Y}^{00}p_{0|0} + y^{\mu}p_{0|1} - y^{l}p_{1|0};$$

and if E[Y|Z = 1] - E[Y|Z = 0] < 0,

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + y^l p_{0|1} - y^u p_{1|0}$$
$$UB = E[Y|Z=1] - E[Y|Z=0].$$

Depending on the sign of  $LATE_c$ , either the lower or the upper bound in Proposition 2.2 improves upon the corresponding bound in Proposition 2.1. When  $LATE_c > 0$ , the lower bounds on  $LATE_a$  and  $LATE_n$  become zero; otherwise, their upper bounds become zero. Consequently, either the lower or upper bound on the ATE equals the ITT effect, depending on the sign of  $LATE_c$ . When the outcome is binary, the bounds in Proposition 2.2 coincide with those in BSV (2008) and Chiburis (2010b), which both equal the bounds in Shaikh and Vytlacil (2011) and Chesher (2010) when there are no exogenous covariates other than the binary instrument. Moreover, if  $LATE_c$  is positive (negative) and Assumptions 1.1 through 1.4, 2.1 and 2.2, hold, then the bounds in Proposition 2.2 equal the bounds obtained by imposing the mean independence assumption of the instrument and increasing (decreasing) MTR assumptions in Manski and Pepper (2000). MTR imposes monotonicity of the outcome in the treatment at the individual level, and it requires one to know the direction of the effect a priori. Depending on the sign of the individual effect, BSV (2008) shows the equivalence of their bounds to those under the IV and MTR assumptions for the case of a binary outcome. Thus, in this setting along with the relaxed version of the monotonicity assumption, these results can be seen as an extension of those in BSV (2008) to the case of a non-binary outcome.<sup>3</sup>

### **3.2.2 Bounds under Mean Dominance**

In practice, some strata are likely to have more favorable characteristics and thus better mean potential outcomes than others. The three alternative assumptions below formal-

<sup>&</sup>lt;sup>3</sup>For a discussion of the trade-off between the MTR assumption and assuming monotonicity of the treatment in the instrument, see BSV (2008).
ize the notion that under the same treatment status, never-takers tend to have the best average potential outcome among the three strata, while always-takers tend to have the worst one.

**Assumption 2.3a**  $E[Y(d)|a] \le E[Y(d)|n]$  for d = 0, 1.

Assumption 2.3b  $E[Y(0)|a] \le E[Y|Z = 0, D = 0]$  and  $E[Y(1)|n] \ge E[Y|Z = 1, D = 1]$ . Assumption 2.3c  $E[Y(0)|a] \le E[Y(0)|c]$  and  $E[Y(1)|n] \ge E[Y(1)|c]$ .

The direction of these assumptions can be inverted depending on the application in question. The always-takers and never-takers are likely to be the most "extreme" groups in many applications, so Assumption 2.3a may be viewed as the weakest of the three. Assumption 2.3b compares the mean Y(0) and Y(1) of the always-takers and never-takers, respectively, to those of a weighted average of the other two strata, while Assumption 2.3c compares them to those of the compliers. Although none of these assumptions is directly testable, it is possible to obtain indirect evidence about their plausibility by comparing relevant average pre-treatment characteristics of the different strata that are highly related to the outcome. These average characteristics of each stratum can be estimated from an overidentified nonparametric GMM problem. For details on the GMM procedure see Appendix C. For Assumption 2.3c, the direction may also be inferred by comparing point identified quantities, E[Y(1)|a] to E[Y(1)|c] and E[Y(0)|n]to E[Y(0)|c], if these inequalities also hold under the alternative treatment status.

The following bounds are presented under Assumptions 1.1 through 1.4, 2.1, and each of the three versions of Assumption 2.3. In each case, the lower bound is higher than that in Proposition 2.1.

**Proposition 2.3** Let  $UB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + y^u p_{0|1} - y^l p_{1|0}$ .

(a) Under Assumptions 1.1 through 1.4, 2.1 and 2.3a the bounds  $LB \le ATE \le UB$  are sharp, where

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + \overline{Y}^{01} p_{0|1} - \overline{Y}^{10} p_{1|0}.$$

(b) Under Assumptions 1.1 through 1.4, 2.1 and 2.3b the bounds  $LB \le ATE \le UB$  are sharp, where

$$LB = \overline{Y}^{11} - \overline{Y}^{00}.$$

(c) Under Assumptions 1.1 through 1.4, 2.1 and 2.3c the bounds  $LB \le ATE \le UB$  are sharp, where

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + \frac{\overline{Y}^{11} p_{1|1} - \overline{Y}^{01} p_{1|0}}{p_{1|1} - p_{1|0}} p_{0|1} - \frac{\overline{Y}^{00} p_{0|0} - \overline{Y}^{10} p_{0|1}}{p_{1|1} - p_{1|0}} p_{1|0}.$$

Assumptions 2.3a through 2.3c have testable implications when combined with Assumption 2.2, if  $LATE_c < 0$ . The following inequalities are expected to hold:  $\overline{Y}^{01} \leq \overline{Y}^{10}$ (2.3a);  $\overline{Y}^{01} \leq \overline{Y}^{00}$  and  $\overline{Y}^{11} \leq \overline{Y}^{10}$  (2.3b); and,  $\overline{Y}^{01} \leq E[Y(0)|c]$  and  $E[Y(1)|c] \leq \overline{Y}^{10}$ (2.3c). If some (or all) of these inequalities are not rejected in applications, then their corresponding assumptions are expected to hold. The following three propositions provide bounds when Assumptions 2.2 and 2.3 are combined.

**Proposition 2.4** Under Assumptions 1.1 through 1.4, 2.1, 2.2 and 2.3a the bounds  $LB \le ATE \le UB$  are sharp, where, if E[Y|Z=1] - E[Y|Z=0] > 0,

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + \max\{\overline{Y}^{10}, \overline{Y}^{01}\} p_{0|1} - \min\{\overline{Y}^{10}, \overline{Y}^{01}\} p_{1|0}$$
$$UB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + y^{u} p_{0|1} - y^{l} p_{1|0};$$

and if E[Y|Z=1] - E[Y|Z=0] < 0,

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + \overline{Y}^{01} p_{0|1} - \overline{Y}^{10} p_{1|0}$$
$$UB = E[Y|Z=1] - E[Y|Z=0].$$

**Proposition 2.5** Under Assumptions 1.1 through 1.4, 2.1, 2.2 and 2.3b the bounds  $LB \leq$ 

 $ATE \leq UB$  are sharp, where, if E[Y|Z=1] - E[Y|Z=0] > 0,

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + \max\{\overline{Y}^{10}, \overline{Y}^{11}\} p_{0|1} - \min\{\overline{Y}^{01}, \overline{Y}^{00}\} p_{1|0}$$
$$UB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + y^{\mu} p_{0|1} - y^{\ell} p_{1|0};$$

and if E[Y|Z = 1] - E[Y|Z = 0] < 0,

$$LB = \overline{Y}^{11} - \overline{Y}^{00}$$
$$UB = E[Y|Z=1] - E[Y|Z=0].$$

**Proposition 2.6** Under Assumptions 1.1 through 1.4, 2.1, 2.2 and 2.3c the bounds  $LB \leq$ 

 $ATE \leq UB$  are sharp, where, if E[Y|Z = 1] - E[Y|Z = 0] > 0,

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + \max\{\overline{Y}^{10}, \frac{\overline{Y}^{11} p_{1|1} - \overline{Y}^{01} p_{1|0}}{p_{1|1} - p_{1|0}}\} p_{0|1}$$
$$-\min\{\overline{Y}^{01}, \frac{\overline{Y}^{00} p_{0|0} - \overline{Y}^{10} p_{0|1}}{p_{1|1} - p_{1|0}}\} p_{1|0}$$
$$UB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + y^{\mu} p_{0|1} - y^{l} p_{1|0};$$

and if E[Y|Z = 1] - E[Y|Z = 0] < 0,

$$LB = \overline{Y}^{11} p_{1|1} - \overline{Y}^{00} p_{0|0} + \frac{\overline{Y}^{11} p_{1|1} - \overline{Y}^{01} p_{1|0}}{p_{1|1} - p_{1|0}} p_{0|1} - \frac{\overline{Y}^{00} p_{0|0} - \overline{Y}^{10} p_{0|1}}{p_{1|1} - p_{1|0}} p_{1|0}$$
$$UB = E[Y|Z=1] - E[Y|Z=0].$$

Note that, if  $LATE_c < 0$ , the bounds in Propositions 2.4 through 2.6 do not require boundedness of the outcome, because Assumption 2.2 improves upon the upper bound in Proposition 2.1, while Assumption 2.3 improves upon the lower bound. In contrast, if  $LATE_c > 0$ , Assumptions 2.2 and 2.3 each improve only upon the lower bound in Proposition 2.1. The bounds in Propositions 2.4 through 2.6 are narrower compared with the bounds in Proposition 2.2 and the corresponding bounds in Proposition 2.3. This is because under the combined assumptions, the monotonicity assumption improves upon further either the lower or upper bound in Proposition 2.3, depending on the sign of  $LATE_c$ , while the mean dominance assumptions further improve upon the lower bound in Proposition 2.2.

Proposition 2.5 overlaps with the bounds recently derived by Chiburis (2010a) under the MTR assumption without specifying a priori direction and the decreasing MTS assumption, as well as the mean independence assumption of the instrument. This is because Assumption 2.3b coincides with the decreasing MTS assumptions imposed on the counterfactual average outcomes for always-takers and never-takers (i.e., E[Y(0)|a] and E[Y(1)|n]). The form of Chiburis' bounds, however, cannot simplify to Proposition 2.6, in that his monotonicity assumptions also involve the counterfactual average outcome for the mixture of never-takers and compliers and that for the mixture of always-takers and compliers (i.e., E[Y(1)|Z = 0, D = 0] and E[Y(0)|Z = 1, D = 1]), which are not involved in the current setting.

It is important to note that the bounds in Proposition 2.6 are also sharp for ATE if Assumption 2.3c is replaced with the assumption,  $E[Y(d)|a] \le E[Y(d)|c] \le E[Y(d)|n]$ for d = 0, 1. However, since E[Y(d)|c] may suffer from the potential issue of a weak IV (i.e.,  $p_{1|1} - p_{1|0}$  is close to zero), and thus it may be more difficult to estimate than E[Y|Z = z, D = d].

#### **3.3 Bounds Addressing Sample Selection and Noncompliance**

#### **3.3.1** Setup, Principal Strata, and Parameter of Interest

To setup the framework in the presence of sample selection and noncompliance, again assume a random sample of size *n* from a large population is available. For each unit *i* in the sample, let  $Z_i = z \in \{0, 1\}$  indicate whether the unit was randomly assigned to the treatment group  $(Z_i = 1)$  or to the control group  $(Z_i = 0)$ . Let  $D_i = d \in \{0, 1\}$ indicate whether individual *i* actually received the treatment  $(D_i = 1)$  or not  $(D_i = 0)$ . Let me further introduce the sample selection indication and the latent outcome. Let  $S_i =$  $s \in \{0, 1\}$  be a post-treatment sample selection variable indicating whether the latent outcome variable  $Y_i^*$  is observed  $(S_i = 1)$  or not  $(S_i = 0)$ . In the setting of evaluating wage effects,  $S_i$  specifies whether individual *i* is employed or not, and  $Y_i^*$  is the offered market wage. The observed outcome variable is  $Y_i = Y_i^*$  if  $S_i = 1$ , and  $Y_i$  is missing if  $S_i = 0$ .

As in IA (1994) and AIR (1996), D(z) denotes the potential compliance behavior as a function of the treatment assignment. In addition, let S(z,d) and  $Y^*(z,d)$  denote the potential values of the selection indicator and the potential latent outcome, respectively, as a function of the treatment assignment (z) and the treatment received (d). In the new setting, we observe  $\{Z_i, D_i(Z_i), S_i(Z_i, D_i(Z_i))\}$  for all units, and  $Y_i^*(Z_i, D_i(Z_i))$  for those with  $S_i = 1$ . The following assumptions are imposed in the new setting to address sample selection and noncompliance simultaneously.

Assumption 3.1 (Randomly Assigned Instrument).  $\{Y^*(z,d), S(z,d), D(z)\}$  is independent of Z for all  $z, d \in \{0,1\}$ .

Assumption 3.2 (Exclusion Restriction of Z).  $Y^*(z,d) = Y^*(z',d) = Y^*(d)$  and S(z,d) = S(z',d) = S(d) for all  $z, d \in \{0,1\}$ .

**Assumption 3.3** (Nonzero Average Effect of Z on D).  $E[D(1) - D(0)] \neq 0$ .

Assumption 3.2 states that any effect of the instrument Z on the potential outcomes  $Y^*$  and on the potential sample selection indicator S must be via an effect of Z on the treatment D. In other words, this assumption prevents the instrument from having a direct effect on  $Y^*$  and S. In the context of the empirical application, Assumption 3.2 requires that randomization affects potential wages and employment only through its effect on JC enrollment. Assumption 3.2 allows me to write the potential variables  $Y^*(z,d)$  and S(z,d) as a function of the treatment d only.

As in IA (1994) and AIR (1996), a valid instrument in this context should satisfy Assumptions 3.1, 3.2, and 3.3 simultaneously. An important difference with respect to the assumptions in those two papers is that here Z is required to be a valid instrument for both  $Y^*$  and S.

To derive bounds for wage effects accounting for sample selection and noncompliance, a principal stratification framework (Frangakis and Rubin, 2002) is employed. This framework, which generalizes the approach in AIR (1996), is useful for studying causal effects when controlling for post-treatment or intermediate variables (i.e., variables that have been affected by the treatment). The basic principal stratification with respect to a given post-treatment variable is a partition of the population into groups such that, within each group, all individuals share the same potential values of the post-treatment variable under each treatment arm. A principal effect is then defined as a comparison of potential outcomes within a given stratum. Since membership to a particular stratum is not affected by treatment assignment, individuals within a group are comparable and thus principal effects are causal effects.

The intermediate variables to control for are the compliance behavior (*D*) and the sample-selection (*S*) indicator. Thus, in this setting, the principal strata are defined by the joint potential values of  $\{D(z = 0), D(z = 1)\} \times \{S(z = 0), S(z = 1)\}$ . Four strata are defined based on the potential compliance behavior: always-takers (*a*), never-takers (*n*), compliers (*c*), and defiers (*d*) in AIR (1996). Following ZRM (2008) and Frumento et al. (2012), the following subpopulations are defined based on potential employment status:

- $EE = \{i : S_i(0) = S_i(1) = 1\}$ , the "always-employed", those who would be employed regardless of treatment assignment;
- $NN = \{i : S_i(0) = S_i(1) = 0\}$ , the "never-employed", those who would be unemployed regardless of treatment assignment for them;
- NE = {i: S<sub>i</sub>(0) = 0, S<sub>i</sub>(1) = 1}, those who would be employed only if assigned to the treatment group;
- $EN = \{i : S_i(0) = 1, S_i(1) = 0\}$ , those who would be employed only if assigned to the control group.

In total, sixteen strata:  $\{a, n, c, d\} \times \{EE, NN, NE, EN\}$  are defined. These strata are the same as those in Frumento et al. (2012), and they result from combining the strata employed in AIR (1996) to account for noncompliance with those used in ZRM (2008) to address sample selection.

An important characteristic of principal strata is that they are latent subpopulations, meaning that, in general, econometricians cannot observe to which stratum each individual belongs. Thus, additional assumptions are usually imposed to point or partially identify effects of interest by reducing the number of strata that exist in the population. Note that Assumption 3.2 implies that the following four strata do not exist: aNE, aEN, nNE, and nEN. The reason is that for the individuals in these four strata there exists an effect of the treatment assignment (Z) on employment (S) that is not through their JC enrollment status (since  $D_i(1) = D_i(0)$ ), which contradicts the exclusion restriction of Z.

The following assumption, which was also employed by AIR (1996) is introduced to further reduce the number of existing strata.

**Assumption 3.4** (Individual-Level Monotonicity of D in Z).  $D_i(1) \ge D_i(0)$  for all i.

Assumption 3.4 rules out the existence of defiers, thus eliminating the strata dEE, dNN, dEN, and dNE in this setting. In the context of the application, it eliminates the existence of individuals who would enroll in JC only if assigned to the control group. A necessary condition for this assumption to hold is that Z has a non-negative average effect on D, which can be falsified by the data. As further discussed in Subsection 3.3.4,

Under Assumptions 3.1 through 3.4, IV estimators point identify the average treatment effect of D on  $Y^*$  (and S) for the compliers in the absence of sample selection. If sample selection is present, however, those assumptions are not enough to point identify the average effect of D on  $Y^*$ .

In this section, the parameter of interest is the average treatment effect of D on wages for the "always-employed compliers" (i.e., the *cEE* stratum):<sup>4</sup>

$$\Delta = E[Y^*(1) - Y^*(0)|cEE] = E[Y(1) - Y(0)|cEE].$$
(3.3)

As can be seen from the definition of the different subpopulations above, this stratum is the only one for which wages are observed for individuals who enrolled and did not enroll into JC after imposing Assumption 3.4. The parameter in (3.3) is also considered in Frumento et al. (2012). It is the average effect for the intersection of the subpopulation IA (1994) and AIR (1996) focus on when accounting for noncompliance with what Lee (2009) and ZRM (2008) focus on when addressing sample selection. In the application, this stratum is the largest one in the population, accounting for about 40 percent. The following subsections construct bounds for (3.3) by considering two more assumptions.

### **3.3.2** Bounds under Monotonicity

This subsection derives the bounds for  $\Delta$  in equation (3.3) by extending the trimming procedure bounds in Zhang and Rubin (2003), ZRM (2008), and Lee (2009) to allow

<sup>&</sup>lt;sup>4</sup>Note that since for compliers Z = D, the stratum *cEE* can also be interpreted as those compliers who would be always employed regardless of treatment *receipt*.

for noncompliance. The following assumption is imposed to further reduce the number of principal strata.

**Assumption 3.5** (Individual-Level Monotonicity of S in D).  $S_i(1) \ge S_i(0)$  for all i.<sup>5</sup>

Assumption 3.5 states that there is a non-negative effect of D on S for every unit. In the application, it assumes that there is a non-negative effect of JC on employment for every individual. While this assumption is similar to the monotonicity assumption employed in ZRM (2008) and Lee (2009), it differs in that the monotonicity of S is imposed in the actual treatment received (D), rather than in the treatment assigned (Z). This type of monotonicity assumption has been employed in the partial identification literature to address problems other than sample selection (e.g., AIR, 1996; Manski and Pepper, 2000; Flores and Flores-Lagunes, 2010a). A testable implication of Assumption 3.5 is that the average effect of D on S for compliers, which is point identified under Assumptions 3.1 through 3.4, is non-negative. Similar to Assumption 3.4, and as further discussed in Subsection 3.3.4, it is possible to relax Assumption 3.5 by letting the direction of the monotonicity be unknown.

Assumption 3.5 rules out strata where the selection indicator *S* is negatively affected by *D*. From the strata remaining after imposing Assumptions 3.1 through 3.4, Assumption 3.5 rules out the existence of the *cEN* stratum. Therefore, under Assumptions 3.1 through 3.5 there are seven strata in the population: *aEE*, *aNN*, *nEE*, *nNN*, *cEE*, *cNN* and *cNE*. The relationship between these seven strata and the observed groups defined by the values of  $\{Z, D(Z), S(Z, D(Z))\}$  is given in Table 3.1.

<sup>&</sup>lt;sup>5</sup>Under Assumptions 3.1 through 3.4, Assumption 3.5 can be relaxed as " $S_i(1) \ge S_i(0)$  for all compliers" in deriving the bounds for  $\Delta$ .

Z = 0				Z = 1				
D				D				
		0	1			0	1	
S	0	cNE, cNN, nNN	aNN	S	0	nNN	cNN, aNN	
	1	cEE,nEE	aEE		1	nEE	cNE, cEE, aEE	

Table 3.1: Observed Groups and Principal Strata

Thus, while some observed groups are composed of only one stratum, some of them are mixtures of two or more strata. Under Assumptions 3.1 through 3.5, the proportion of each stratum in the population can be identified. Let  $\pi_k$  denote the proportion of stratum k in the population, and let  $p_{ds|z} \equiv \Pr(D = d, S = s|Z = z)$  and  $q_{s|z} \equiv \Pr(S = s|Z = z)$ . Then:

$$\pi_{aNN} = p_{10|0}; \pi_{aEE} = p_{11|0}; \pi_{nNN} = p_{00|1}; \pi_{nEE} = p_{01|1}$$
(3.4)  
$$\pi_{cEE} = p_{01|0} - p_{01|1}; \pi_{cNN} = p_{10|1} - p_{10|0}; \pi_{cNE} = q_{1|1} - q_{1|0}.$$

In addition, the mean outcomes for those observed cells with S = 1 can be written as a function of mean potential outcomes for different strata. Letting  $\overline{Y}^{zds} \equiv E[Y|Z = z, D = d, S = s]$ , then:

$$\overline{Y}^{011} = E[Y(1)|aEE] \tag{3.5}$$

$$\overline{Y}^{101} = E[Y(0)|nEE] \tag{3.6}$$

$$\overline{Y}^{001} = E[Y(0)|cEE] \frac{\pi_{cEE}}{p_{01|0}} + E[Y(0)|nEE] \frac{\pi_{nEE}}{p_{01|0}}$$
(3.7)

$$\overline{Y}^{111} = E[Y(1)|cEE]\frac{\pi_{cEE}}{p_{11|1}} + E[Y(1)|cNE]\frac{\pi_{cNE}}{p_{11|1}} + E[Y(1)|aEE]\frac{\pi_{aEE}}{p_{11|1}}$$
(3.8)

The average potential outcomes under treatment and control are point identified for the *nEE* and *aEE* strata, respectively. Moreover, it is possible to point identify E[Y(0)|cEE] by combining equations (3.4), (3.6), and (3.7) to obtain:

$$E[Y(0)|cEE] = \frac{p_{01|0}}{p_{01|0} - p_{01|1}} \overline{Y}^{001} - \frac{p_{01|1}}{p_{01|0} - p_{01|1}} \overline{Y}^{101}.$$
(3.9)

Thus, one of the terms of  $\Delta$  in (3.3) is point identified. But the term E[Y(1)|cEE] is not point identified because two of the conditional means in (3.8) are not point identified. Next, bounds are constructed for E[Y(1)|cEE] and  $\Delta$ .

In a setting without noncompliance, Zhang and Rubin (2003), ZRM (2008), and Lee (2009) construct bounds for the non-point identified expectation of the potential outcome in the definition of their average effect based on a cell containing only two strata. To illustrate the main idea behind their "worst-case" bounds, suppose that there were no individuals in the *aEE* stratum, so that  $\pi_{aEE} = 0$  and the cell  $\{Z = 1, D = 1, S =$ 1 $\}$  contained only two strata, *cEE* and *cNE*. Then, E[Y(1)|cEE] would be bounded from above (below) by the mean of Y for the fraction  $\pi_{cEE}/(\pi_{cEE} + \pi_{cNE})$  of the largest (smallest) values of Y for those individuals in that cell. A key difference between the bounds derived in those studies and this one is that in the current setting the bounds for E[Y(1)|cEE] are derived from a cell containing three strata.

By equations (3.5) and (3.8), although the observed mean  $\overline{Y}^{111}$  is a weighted average of the mean potential outcome of Y(1) for three strata, the mean E[Y(1)|aEE] is point identified. Thus, the bounds are constructed by considering "worst-case" scenarios that exploit the information that  $\overline{Y}^{011} = E[Y(1)|aEE]$ . To motivate the way to construct the bounds, the problem can be thought of as finding "worst-case" scenarios

for E[Y(1)|cEE] subject to the constraint that  $\overline{Y}^{011} = E[Y(1)|aEE]$ . The strategy to derive bounds for E[Y(1)|cEE] is to solve the unconstrained problem first, and then check whether the value of E[Y(1)|aEE] implied by this solution can satisfy the constraint that  $\overline{Y}^{011} = E[Y(1)|aEE]$ . If the constraint can be satisfied, then the unconstrained solution is just the solution to the constrained problem. Otherwise, impose the constraint first and then obtain the solution to the constrained problem.

Additional notations are introduced to describe the bounds for  $\Delta$ . Let  $y_r^{111}$  be the *r*-th quantile of *Y* in the cell {Z = 1, D = 1, S = 1}, and let

$$\overline{Y}(y_{r'}^{111} \le Y \le y_r^{111}) \equiv E[Y|Z=1, D=1, S=1, y_{r'}^{111} \le Y \le y_r^{111}].$$
(3.10)

Hence,  $\overline{Y}(y_{r'}^{111} \leq Y \leq y_{r}^{111})$  gives the mean outcome in the cell  $\{Z = 1, D = 1, S = 1\}$  for those outcomes between the r'-th and r-th quantiles of Y in that cell. Suppose that I want to derive the lower bound for E[Y(1)|cEE]. To begin, I consider the problem without the constraint and ignore the information about aEE. In this case, I can directly apply the existing trimming procedure in ZRM (2008) and Lee (2009) and bound E[Y(1)|cEE]from below by the expected value of Y for the  $\pi_{cEE}/p_{11|1}$  fraction of the smallest values of Y in the cell  $\{Z = 1, D = 1, S = 1\}$ , or,  $\overline{Y}(Y \leq y_{\pi_{cEE}/p_{11|1}}^{111})$ , where  $p_{11|1} = \pi_{cEE} + \pi_{cNE} + \pi_{aEE}$ . Next, check whether this solution is consistent with the constraint that  $\overline{Y}^{011} = E[Y(1)|aEE]$ . To do this, I construct the "worst-case" scenario lower bound for E[Y(1)|aEE], call it  $LY_{1,aEE}$ , implied by the unconstrained solution by assuming that all the observations that belong to the aEE stratum are at the bottom of the remaining observations in the cell  $\{Z = 1, D = 1, S = 1\}$ . This yields  $LY_{1,aEE} = \overline{Y}(y_{\pi_{cEE}/p_{11|1}}^{111} \leq Y \leq y_{1-(\pi_{cNE}/p_{11|1})}^{111})$ . If  $LY_{1,aEE} \leq \overline{Y}^{011}$ , the unconstrained solution is consistent with the constraint and the lower bound for E[Y(1)|cEE] is  $\overline{Y}(Y \leq y_{\pi_{cEE}/p_{11|1}}^{111})$ . If  $LY_{1,aEE} > \overline{Y}^{011}$ , then the unconstrained solution is inconsistent with  $\overline{Y}^{011} = E[Y(1)|aEE]$ . Intuitively, having  $LY_{1,aEE} > \overline{Y}^{011}$  implies that some observations from the aEE stratum must be at the bottom  $\pi_{cEE}/p_{11|1}$  fraction of the smallest values of Y in the cell  $\{Z = 1, D = 1, S =$ 1 $\}$  and thus  $\overline{Y}(Y \leq y_{\pi_{cEE}/p_{11|1}}^{111})$  is not a sharp lower bound for E[Y(1)|cEE]. In this case, the "worst-case" scenario lower bound for E[Y(1)|cEE] is constructed by placing all the observations in the aEE and cEE strata at the bottom of the distribution of Y in the cell  $\{Z = 1, D = 1, S = 1\}$ . Thus, if  $LY_{1,aEE} > \overline{Y}^{011}$ , the lower bound for E[Y(1)|cEE], call it  $LY_{1,cEE}$ , is derived from the equation:

$$\overline{Y}(Y \le y_{1-(\pi_{cNE}/p_{11|1})}^{111}) = \frac{\pi_{cEE}}{\pi_{cEE} + \pi_{aEE}} LY_{1,cEE} + \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} \overline{Y}^{011},$$
(3.11)

where  $\overline{Y}(Y \le y_{1-(\pi_{cNE}/p_{11|1})}^{111})$  is the mean of *Y* for the  $1 - (\pi_{cNE}/p_{11|1})$  fraction of the smallest values of *Y* in the cell  $\{Z = 1, D = 1, S = 1\}$ .

Note that the lower bound  $LY_{1,cEE}$  derived from equation (3.11) does not yield a sharp bound for E[Y(1)|cEE] if  $LY_{1,aEE} \leq \overline{Y}^{011}$ . For example, if  $\overline{Y}^{011} = E[Y(1)|aEE]$  is large, so that it is impossible that all individuals from the aEE stratum are at the bottom  $1 - (\pi_{cNE}/p_{11|1})$  fraction of the smallest values of Y in the cell  $\{Z = 1, D = 1, S = 1\}$ , then the lower bound derived from (3.11) will be much lower than  $\overline{Y}(Y \leq y_{\pi_{cEE}/p_{11|1}}^{111})$ , the lower bound derived without using the information on E[Y(1)|aEE]. Intuitively, in this case the value of  $\overline{Y}^{011} = E[Y(1)|aEE]$  is so large that it provides little information about the "worst-case" lower bound scenario for E[Y(1)|cEE] (but it will provide valuable information for the upper bound of E[Y(1)|cEE]). The upper bound for E[Y(1)|cEE] is derived in a similar way as the lower bound, but now by placing the observations in the corresponding strata in the upper part of the distribution of Y in the cell {Z = 1, D = 1, S = 1}. Once bounds for E[Y(1)|cEE] are obtained, they can be combined with the point identification of E[Y(0)|cEE] in (3.9) to construct bounds for the average effect of the always-selected compliers,  $\Delta$  in (3.3). The following proposition presents bounds for  $\Delta$  under Assumptions 3.1 through 3.5.

**Proposition 3.1** If Assumptions 3.1 through 3.5 hold, then  $L_{cEE} \leq \Delta \leq U_{cEE}$ .  $L_{cEE}$  and  $U_{cEE}$  are lower and upper bounds for  $\Delta$  given by:

$$L_{cEE} = LY_{1,cEE} - \overline{Y}^{001} \frac{P_{01|0}}{p_{01|0} - p_{01|1}} + \overline{Y}^{101} \frac{P_{01|1}}{p_{01|0} - p_{01|1}}$$
$$U_{cEE} = UY_{1,cEE} - \overline{Y}^{001} \frac{P_{01|0}}{p_{01|0} - p_{01|1}} + \overline{Y}^{101} \frac{P_{01|1}}{p_{01|0} - p_{01|1}}$$

where

$$LY_{1,cEE} = \begin{cases} \overline{Y}(Y \le y_{\alpha_{cEE}}^{111}), \text{ if } \overline{Y}(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{cNE}}^{111}) \le \overline{Y}^{011} \\ \overline{Y}(Y \le y_{1-\alpha_{cNE}}^{111}) \frac{q_{1|0} - p_{01|1}}{p_{01|0} - p_{01|1}} - \overline{Y}^{011} \frac{p_{11|0}}{p_{01|0} - p_{01|1}}, \text{ otherwise} \end{cases}$$

$$UY_{1,cEE} = \begin{cases} \overline{Y}(Y \ge y_{1-\alpha_{cNE}}^{111}), \text{ if } \overline{Y}(y_{\alpha_{cNE}}^{111} \le Y \le y_{1-\alpha_{cEE}}^{111}) \ge \overline{Y}^{011} \\ \overline{Y}(Y \ge y_{\alpha_{cNE}}^{111}) \frac{q_{1|0} - p_{01|1}}{p_{01|0} - p_{01|1}} - \overline{Y}^{011} \frac{p_{11|0}}{p_{01|0} - p_{01|1}}, \text{ otherwise} \end{cases}$$

$$\alpha_{cEE} = \frac{\pi_{cEE}}{p_{11|1}} = \frac{p_{01|0} - p_{01|1}}{p_{11|1}}, \text{ and}$$

$$\alpha_{cNE} = \frac{\pi_{cNE}}{p_{11|1}} = \frac{q_{1|1} - q_{1|0}}{p_{11|1}}.$$

**Proof.** See Appendix A.

## **3.3.3** Bounds under Mean Dominance

This subsection considers a mean dominance assumption that narrows the bounds in Proposition 3.1.

Assumption 3.6 (Mean Dominance).  $E[Y(1)|cEE] \ge E[Y(1)|cNE]$ .

The intuition behind Assumption 3.6 is the same as the one behind Assumption 2.3: some strata are expected to have more favorable characteristics and thus better potential outcomes than others. In the application, this assumption states that the mean potential outcome under treatment of the always-employed compliers is greater than or equal to that of those who would be employed only if they enrolled in JC. Assumption 3.6 implies a positive correlation between employment and wages, which is supported by standard economic models of labor supply. Zhang and Rubin (2003), ZRM (2008) and Huber and Mellace (2010) consider stochastic-dominance versions of Assumption 3.6. For example, in the current setting, such an assumption would state that the potential outcome under treatment of the cEE stratum at any rank of the outcome distribution is weakly less than that of the cNE stratum. For the purposes of the dissertation, stochastic dominance is much stronger than needed.

As Assumption 2.3 in the Section 3.2, even though Assumption 3.6 is not directly testable, it is possible to get indirect evidence about its plausibility by comparing the average baseline characteristics of the *cEE* and *cNE* strata that are closely related to the outcome of interest (e.g., values of the outcome prior to randomization). Assumption 3.6 is less likely to hold if these comparisons suggest that the *cNE* stratum has better characteristics at the baseline than does the *cEE* stratum. Under Assumptions 3.1 through 3.5 it is possible to point identify the average characteristics for all seven strata at the baseline. This can be seen by noting that the observed average characteristics at the observed groups  $\{Z, D, S\}$  in Table 3.1 is a weighted average of the average characteristics for the different strata (see, for reference, equations (3.7)

and (3.8)), with the weights being point identified from (3.4). Based on these equations, the average characteristics at the baseline for all seven strata can be estimated by solving an overidentified GMM problem. This tool is implemented in the application and further details are provided in the Appendix C.

The mean dominance assumption above tightens the bounds in Proposition 3.1 by increasing the lower bound on E[Y(1)|cEE]. To get the new lower bound, note that, similar to equation (3.8), I can write

$$\overline{Y}^{111} = E[Y(1)|cEE, cNE] \frac{\pi_{cEE} + \pi_{cNE}}{p_{11|1}} + E[Y(1)|aEE] \frac{\pi_{aEE}}{p_{11|1}}, \quad (3.12)$$

where the stratum proportions and E[Y(1)|aEE] are point identified. With Assumption 3.6,  $E[Y(1)|cEE] \ge E[Y(1)|cEE, cNE]$ , which provides a lower bound for E[Y(1)|cEE]that is greater than or equal to the one obtained in Proposition 3.1. The following proposition presents bounds for  $\Delta$  under Assumptions 3.1 through 3.6.

**Proposition 3.2** If Assumptions 3.1 through 3.6 hold, then  $L_{cEE} \leq \Delta \leq U_{cEE}$ .  $L_{cEE}$  and  $U_{cEE}$  are lower and upper bounds for  $\Delta$ , where  $U_{cEE}$  is equal to the upper bound for  $\Delta$  given in Proposition 3.1 and  $L_{cEE}$  equals:

$$L_{cEE} = LY_{1,cEE} - \overline{Y}^{001} \frac{P_{01|0}}{p_{01|0} - p_{01|1}} + \overline{Y}^{101} \frac{P_{01|1}}{p_{01|0} - p_{01|1}}$$

,

with

$$LY_{1,cEE} = \frac{p_{11|1}\overline{Y}^{111} - p_{11|0}\overline{Y}^{011}}{p_{11|1} - p_{11|0}}.$$

**Proof.** See Appendix A.

# 3.3.4 Remarks on Assumptions

The following remarks discuss how to relax the assumptions used in Section 3.3.

**Remark 1.** As Assumption 1.4 in Section 3.1, it is possible to relax the individuallevel monotonicity assumptions (Assumptions 3.4 and 3.5) by not requiring prior knowledge about their direction. This is closely related to the work by Shaikh and Vytlacil (2011), who derive bounds for average treatment effects on binary outcomes with a valid instrument by imposing monotonicity (or threshold crossing models) assumptions similar to those in Assumptions 3.4 and 3.5 without specifying the direction of the monotonicity (see also BSV, 2008). In this setting, Assumptions 3.4 and 3.5 can be replaced with the following assumptions.

Assumption 3.4' (Individual-Level Monotonicity of D in Z, unknown direction). Either  $D_i(1) \ge D_i(0)$  for all i or  $D_i(1) \le D_i(0)$  for all i.

**Assumption 3.5'** (Individual-Level Monotonicity of S in D, unknown direction). Either  $S_i(1) \ge S_i(0)$  for all i or  $S_i(1) \le S_i(0)$  for all i.

**Assumption 3.7**  $E[S|Z = 1] - E[S|Z = 0] \neq 0$ .

Under Assumptions 3.1, 3.2, 3.3, 3.4', 3.5', and 3.7, it is possible to infer the directions of the monotonicity assumptions above from the observed data, and hence, derive bounds for the average effect of D on Y for either the *cEE* or *dEE* stratum by the procedure described in Subsection 3.3.2.

The direction of the monotonicity of *D* in *Z* can be inferred directly from E[D|Z = 1] - E[D|Z = 0]. In addition, note that under Assumption 3.7 the instrumental variable estimator of the effect of *D* on *S*, (E[S|Z = 1] - E[S|Z = 0])/(E[D|Z = 1] - E[D|Z = 0]), point identifies the effect of *D* on *S* for a subpopulation (either the compliers or the defiers, depending of the direction of Assumption 3.4'). Since all the individuals in the

population share the same direction of the monotonicity, the direction of the monotonicity of S in D can be inferred from the sign of the instrumental variable estimator.<sup>6</sup>

Depending on the direction of the monotonicity in Assumption 3.4', the parameter of interest would be the average effect of D on Y either for the always-employed compliers or the always-employed defiers. In the context of constructing bounds for this effect, the three assumptions above imply the existence of only one stratum out of cNE, cEN, dNE and dEN.<sup>7</sup> Thus, similar to the case considered in Subsection 3.3.2, one of the mean potential outcomes in the parameter of interest would be point identified (e.g., E[Y(0)|cEE]), while the other is partially identified (e.g., E[Y(1)|cEE]). The bounds for the non-point identified term can be constructed following the same procedure described in Subsection 3.3.2. Moreover, an appropriate mean dominance assumption similar to Assumption 3.6 could be used to narrow the bounds, as in Subsection 3.3.3.

**Remark 2.** It is possible to construct bounds on  $\Delta$  without Assumption 3.5, in which case the stratum *cEN* is not ruled out and appears in the observed cells {Z = 0, D = 0, S = 1} and {Z = 1, D = 1, S = 0} in Table 3.1. Although the proportions of the strata *aEE*, *aNN*, *nEE*, and *nNN* are still point identified, neither the proportions of the strata *cEE*, *cNN*, *cNE*, and *cEN* nor the term E[Y(0)|cEE] is now point identified. To construct bounds for  $\Delta$  in this case, the approach in Subsection 3.3.3 can be combined with that followed by Zhang and Rubin (2003), ZRM (2008), Imai (2008), and Huber and Mellace (2010) in a setting with sample selection but without the noncompliance

<sup>&</sup>lt;sup>6</sup>Note that, under the current assumptions, if E[S|Z=1] - E[S|Z=0] = 0, the number of strata reduces to six: *aEE*, *aNN*, *nEE*, *nNN*, plus either *cEE* and *cNN* or *dEE* and *dNN*. In this case, the average treatment effect of *D* on *Y* is point identified for either the *cEE* or the *dEE* stratum.

<sup>&</sup>lt;sup>7</sup>To see this, note that by Assumptions 3.1 and 3.2,  $E[D|Z=1] - E[D|Z=0] = \pi_c - \pi_d$  and  $E[S|Z=1] - E[S|Z=0] = [\Pr(cNE|c) - \Pr(cEN|c)]\pi_c + [\Pr(dNE|d) - \Pr(dEN|d)]\pi_d$ .

issue. The main idea for constructing their bounds is to consider "worst-case" scenarios for their effect of interest that are consistent with the possible values that  $\pi_{EN}$  (and hence  $\pi_{EE}$ ) can take based on the data.<sup>8</sup> In the current setting, the "worst-case" scenarios for  $\Delta$ occur when  $\pi_{cEE}$  is at its minimum value that is consistent with the data, which equals  $p_{01|0} - p_{01|1} - p_{10|1} + p_{10|0}$ .<sup>9</sup> Given this lower bound for  $\pi_{cEE}$ , the same approach as in Subsection 3.3.2 to derive bounds for  $\Delta$  by constructing bounds for E[Y(1)|cEE] and E[Y(0)|cEE]. However, the bounds in this case will be wider than those presented in Proposition 3.1, and may result in uninformative bounds (see e.g., Blanco, Flores, and Flores-Lagunes, 2012).

**Remark 3.** In the absence of Assumptions 3.5 and 3.6, the lower bound for  $\Delta$  in Proposition 3.2 provides information for another parameter of interest,  $ATE_{cEE,cNE}$ , which is defined as the weighted average of  $\Delta$  and the *ATE* for *cNE*.  $L_{cEE}$  in Proposition 3.2 can be viewed as the lower bound for  $ATE_{cEE,cNE}$ , under Assumptions 3.1 through 3.4 and the following assumption.

Assumption 3.5"  $E[Y(0)|cEE, cEN] \ge E[Y(0)|cEE, cNE]$ , where  $E[Y(0)|k_1, k_2]$  is the weighted average of Y(0) between two strata  $k_1$  and  $k_2$ .

This assumption states that the mean value of Y(0) (i.e., the potential wage if not attending JC) for compliers who would be employed if they did not attend JC (*cEE* and

<sup>&</sup>lt;sup>8</sup>Zhang and Rubin (2003) and ZRM write the bounds for their parameter as functions of  $\pi_{EN}$ , and then obtain the upper or lower bound by minimizing or maximizing the bounds over all possible values of  $\pi_{EN}$  that are consistent with the data. Huber and Mellace (2010) show that the numerical optimization is not necessary, and the bounds are obtained at the maximal value of  $\pi_{EN}$ .

<sup>&</sup>lt;sup>9</sup>The range of possible values of  $\pi_{cEE}$  is calculated from the eight cells in Table 3.1, which yields  $\pi_{cEE} \in [\max(0, p_{01|0} - p_{01|1} - p_{10|1} + p_{10|0}), \min(p_{01|0} - p_{01|1}, p_{11|1} - p_{11|0})]$ . As noticed by Lee (2009), the bounds of  $\Delta$  are well defined only if  $\pi_{cEE} > 0$ , which implies  $\pi_{cEE}$  is minimized at  $p_{01|0} - p_{01|1} - p_{10|1} + p_{10|0}$ .

*cEN*) is greater than or equal to that for compliers who would be employed if they did attend JC (*cEE* and *cNE*). This assumption exploits the positive correlation between employment and wages implied by standard models of labor supply, as the *cEN* are employed under the control treatment but the *cNE* are not. Indirect evidence regarding the plausibility of Assumption 3.5" can be obtained by comparing the weighted average baseline characteristics, E[X|cEE,cEN] and E[X|cEE,cNE], derived from the cells  $\{Z = 0, D = 0, S = 1\}$  and  $\{Z = 1, D = 1, S = 1\}$ , respectively. In applications where Assumption 3.5 or 3.6 are difficult to justify, Assumption 3.5" may become attractive. Furthermore, the mixture of the strata *cEE* and *cNE* seems to be an interesting target group, since those are individuals who would comply with the treatment assignment and would be employed if they attended JC.

**Remark 4.** This section focuses on the average treatment effect of the *cEE* stratum. It is possible to combine the methods in the previous subsections with those in Huber and Mellace (2010) to construct bounds for the average effects of other subpopulations. For instance, consider the average effect of the treated and selected individuals (those with D = 1 and S = 1), or of the treated and selected compliers. As can be seen from Table 3.1, these other subpopulations are mixtures of different strata for which, with the exception of *cEE*, wages are unobserved under one of the treatment assignments. For example, wages are never observed under the control treatment for those who would be unemployed if they did not enroll in JC (the *cNE* group), or for those who would be always employed regardless of treatment assignment (the always-takers). Thus, additional assumptions (e.g., a bounded outcome) are needed to partially identify the effects for other strata or subpopulations.

## **3.4 Estimation and Inference**

A first consideration when performing statistical inference in partially identified models is whether one wants to construct confidence regions for the identified set (e.g., [LB, UB],  $[L_{cEE}, U_{cEE}]$ ) or for the true value of the parameter (e.g., ATE,  $\Delta$ ). The section focuses on confidence regions for the partially identified parameter.

Imbens and Manski (2004) and Stoye (2009) provide confidence intervals that asymptotically cover the true value of a parameter  $\theta_0$  with a fixed probability when the bounds are of the form  $\theta_0^l \leq \theta_0 \leq \theta_0^u$  and there are estimators of  $\theta_0^l$  and  $\theta_0^u$  that behave asymptotically like sample means. Their analysis, however, does not apply to all the bounds in the dissertation because the bounds in Propositions 2.4, 2.5, 2.6, and 3.1, 3.2 involve minimum (min) and maximum (max) operators. For example, the upper bound  $UY_{1,cEE}$  for E[Y(1)|cEE] in Proposition 3.1 can be written as  $UY_{1,cEE} = \min\{\overline{Y}(Y \ge 1)\}$  $y_{1-\alpha_{cEE}}^{111}$ ,  $\overline{Y}(Y \ge y_{\alpha_{cNE}}^{111}) \frac{q_{1|0}-p_{01|1}}{p_{01|0}-p_{01|1}} - \overline{Y}^{011} \frac{p_{11|0}}{p_{01|0}-p_{01|1}}$ . When the two quantities in the min operator are close to each other, the distribution of the estimator for the upper bound is not well approximated by a normal distribution. Those two operators create complications for estimation and inference. First, sample analog estimators of the bounds can be severely biased in small samples. Because of the concavity (convexity) of the min (max) function, sample analog estimates of the bounds tend to be much narrower than the true bounds. Second, closed-form characterization of the asymptotic distribution of estimators for parameters involving min or max functions is very difficult to derive and, thus, usually unavailable. Moreover, Hirano and Porter (2012) show that there exist no locally asymptotically unbiased estimators and no regular estimators for parameters that are nonsmooth functionals of the underlying data distribution, such as those involving min or max operators.<sup>10</sup> Hence, the uniform asymptotic normality condition in Imbens and Manski (2004) and Stoye (2009) does not hold when the two operators are present. This has generated a growing literature on inference methods for partially identified models of this type (see Tamer, 2010, and the references therein).

To address those issues, the dissertation employs the methodology proposed by Chernozhukov, Lee, and Rosen (2011) that are applicable to bounds of the form  $[\theta_0^l, \theta_0^u]$ , where  $\theta_0^l = \sup_{v \in \mathscr{V}} \theta^l(v)$ ,  $\theta_0^u = \inf_{v \in \mathscr{V}} \theta^u(v)$ ,  $\theta^l(v)$  and  $\theta^u(v)$  are bounding functions, and  $\mathscr{V}$  is the set over which the infimum and supremum are taken. They employ precision-corrected estimates of the bounding functions before applying the infimum and supremum operators. The precision adjustment consists of adding to each estimated bounding function its pointwise standard error times an appropriate critical value. Hence, estimates with higher standard errors require larger adjustments. Depending on the choice of the critical value, it is possible to obtain confidence regions for either the identified set or the true parameter value, as well as half-median unbiased estimators for the lower and upper bounds. The half-median-unbiasedness property means that the upper (lower) bound estimator exceeds (falls below) the true value of the upper (lower) bound with the probability of at least one-half asymptotically. This property is important because achieving local asymptotic unbiasedness is impossible (Hirano and Porter, 2012). For details on the procedure of applying their method to the bounds in these five propositions see Appendix B. For the bounds without min or max operators, sample

<sup>&</sup>lt;sup>10</sup>As documented by Hirano and Porter (2011), nonexistence of local asymptotic unbiased estimators implies that bias correction procedures cannot completely get rid of local bias and that reducing bias too much will eventually lead to arbitrarily large variance.

analog estimators and the confidence regions for the true parameter value proposed by Imbens and Manski (2004) are used.

#### CHAPTER 4

### **APPLICATION: EVALUATION OF JOB CORPS**

## 4.1 Job Corps and the National Job Corps Study

Since 1964, Job Corps (JC) has been a central part of the federal government efforts to provide job training and employment assistance to disadvantaged youth. Services such as academic education, vocational training, residential living, health care and health education, counseling and job placement assistance are delivered at more than 120 centers nationwide. To be eligible for the program, an individual must be a legal resident of the United States, be between 16 and 24 years old and come from a low-income household. According to the US Department of Labor (2005), a typical JC student lives at a local JC center for eight months and receives about 1100 hours of academic and vocational instruction, which is equivalent to approximately one year in high school.

This dissertation employs data from the National Job Corps Study (NJCS), a randomized experiment funded by the US Department of Labor to evaluate the effectiveness of JC. The study examined the impacts of JC on labor market outcomes, welfare dependence and several other outcomes to help assess whether the program achieved its goals of helping students become more responsible and productive citizens. Eligible individuals who applied to JC for the first time between November 1994 and December 1995 (80,833 individuals) were randomly assigned to a program, control, or program non-research group. Individuals in the control group (5,977) were embargoed from the program for a period of three years, while those in the program (treatment) group (9,409) were allowed to enroll in JC. The research sample was interviewed at the time of random assignment and at 12, 30, and 48 months after random assignment. Due to both design and programmatic reasons, some subpopulations were randomized in the NJCS with different (but known) probabilities (Schochet, Burghardt, and Glazerman, 2001). Hence, design weights are employed throughout the analysis.

Taking advantage of randomization, most of previous literature evaluating the JC program studies ITT effects or LATEs for compliers (e.g., Burghardt et al., 2001; Schochet, Burghardt, and Glazerman, 2001; Schochet, Burghardt, and McConnell, 2008; Lee, 2009; Blanco, Flores, and Flores-Lagunes, 2012). The noncompliance behavior, however, tends to dilute the impacts of JC. Across the samples in the dissertation, 73%of individuals of the treatment group actually enrolled in JC, while 4% of individuals of the control group also enrolled. Burghardt et al. (2001), Schochet, Burghardt, and Glazerman (2001) and Schochet, Burghardt, and McConnell (2008) also adjust to noncompliance by examining the *LATE*, which is representative for the subpopulation (i.e., compliers) accounting for 69% in the population. Different from the previous literature, using the random assignment as an IV, the present research addresses this noncompliance issue and analyzes the population ATEs of actual JC enrollment on participants' labor market outcomes and welfare. It also examines wage effects of JC enrollment by focusing on a specified subpopulation, i.e., the always-employed compliers. Thus, the dissertation also contributes to the previous literature on the evaluation of the JC program by providing credible partial identification results for treatment effects other than LATE and the ITT effect.

### **4.2** The Population *ATE* of JC Enrollment

# 4.2.1 Specific Sample

Given the objectives and services provided by JC (e.g., academic and vocational training, job search assistance), making inference about the population average treatment effects of JC enrollment is of great public interest. This section uses data on individuals who responded to the 48-month interview to examine the *ATEs* on weekly earnings and employment at Week 208 (i.e., four years) and public assistance benefits received during the fourth year after randomization<sup>11</sup>. On one hand, JC tends to have positive effects on participants' labor market outcomes given its objectives and services. On the other hand, participants may experience a reduction in welfare receipt while they enroll in JC, because the program provides shelter (except to nonresidential students), food, and a stipend. After they leave JC, participants may receive less public income support because of higher earnings. Schochet, Burghardt, and Glazerman (2001) report that the reductions in benefit receipt persisted throughout four years after randomization.

The treatment variable indicates whether or not the individual ever enrolled in JC during the 208 weeks after random assignment. The random assignment indicator serves as an instrument for JC enrollment. Two samples are obtained by dropping individuals with missing relevant variables from the survey.<sup>12</sup> The sample for weekly earnings and employment involves 10,520 individuals (4,187 and 6,333 in the control and treatment

<sup>&</sup>lt;sup>11</sup>Benefits include Aid to Families with Dependent Children (AFDC) or Temporary Assistance for Needy Families (TANF), food stamps, Supplemental Security Income (SSI) or Social Security Retirement, Disability, or Survivor (SSA), and General Assistance.

<sup>&</sup>lt;sup>12</sup>Two samples are derived because individuals with missing labor market outcomes and with missing public benefits are different.

groups, respectively), and for public benefits 10,976 individuals (4,387 and 6,589 in the control and treatment groups, respectively).

Table 4.1 reports the average baseline characteristics of both samples by treatment assignment status along with the percentage of missing values for each of those variables. The pre-treatment variables include demographic characteristics, education and background variables, employment, earnings and public benefits dependency at baseline, as well as labor market outcomes in the year prior to randomization. As one would expect, the average pre-treatment characteristics of the treatment and control groups are similar in both the samples due to randomization, with the difference in means being statistically different from zero at the five percent level for only one variable (personal income: 3,000-6,000). Thus, both samples maintain the balance of baseline variables between the control and treatment groups.

#### 4.2.2 Assessment of Assumptions

Assumptions 1.1 through 1.4 are commonly used in the literature to address noncompliance in experimental settings. This subsection concentrates on the discussion of assessment of Assumptions 2.2 and 2.3.

Table 4.2 shows some relevant point identified averages for both samples. The noncompliance behavior is similar between the two samples. As already mentioned, 73% of individuals of the treatment group actually enrolled in JC, while 4% of individuals of the control group also enrolled during the 208 weeks after randomization. The *ITT* effects on weekly earnings, employment and public benefits are 22.19, .038 and -84.29, respectively. These effects are all statistically significant, with their signs as expected.

	Sample for Labor Market Outcomes			Sample for Public Assistance Benefits				
	Missing Prop.	ing Prop. Z=1 Z=0 Diff. (Std.Err.)		Missing Prop.	Z=1	Z=0	Diff. (Std.Err.)	
Female	0	.417	.407	.009 (.010)	0	.415	.406	.009 (.010)
Age at Baseline	0	18.42	18.38	.035 (.042)	0	18.41	18.38	.031 (.041)
White, Non-hispanic	0	.273	.266	.007 (.009)	0	.274	.269	.005 (.009)
Black, Non-Hispanic	0	.483	.478	.005 (.010)	0	.477	.474	.003 (.010)
Hispanic	0	.171	.179	008 (.008)	0	.172	.180	008 (.007)
Other Race/Ethnicity	0	.073	.078	005 (.005)	0	.076	.076	.000 (.005)
Never Married	.017	.916	.915	.001 (.006)	.020	.914	.915	001 (.005)
Married	.017	.020	.022	002 (.003)	.020	.020	.022	001 (.003)
Living Together	.017	.040	.041	001 (.004)	.020	.040	.041	001 (.004)
Separated	.017	.024	.022	.002 (.003)	.020	.025	.022	.003 (.003)
Has Child	.007	.181	.184	003 (.008)	.008	.181	.183	002 (.008)
Number of Children	.011	.253	.248	.005 (.012)	.012	.251	.247	.004 (.012)
Personal Education	.018	10.08	10.09	008 (.031)	.021	10.08	10.10	019 (.030)
Mother's Education	.194	11.50	11.51	011 (.058)	.197	11.49	11.53	042 (.057)
Father's Education	.391	11.43	11.54	110 (.073)	.394	11.45	11.57	127* (.072)
Ever Arrested	.017	.258	.263	005 (.009)	.019	.259	.266	007 (.009)
Household Inc.: <3000	.368	.252	.258	006 (.011)	.371	.250	.255	005 (.011)
3000-6000	.368	.201	.204	004 (.010)	.371	.198	.208	010 (.010)
6000-9000	.368	.116	.111	.006 (.008)	.371	.117	.109	.008 (.008)
9000-18000	.368	.245	.243	.001 (.011)	.371	.246	.241	.005 (.011)
>18000	.368	.187	.183	.003 (.010)	.371	.189	.187	.002 (.010)
Personal Inc.: <3000	.083	.786	.790	004 (.008)	.086	.783	.788	006 (.008)
3000-6000	.083	.129	.129	.000 (.007)	.086	.130	.131	000 (.007)
6000-9000	.083	.055	.046	.009** (.005)	.086	.056	.046	.010** (.004)
>9000	.083	.031	.036	005 (.004)	.086	.031	.035	004 (.004)
Have Job	.031	.216	.209	.007 (.008)	.034	.219	.211	.009 (.008)
Weekly Hours Worked	0	21.69	21.13	.563 (.417)	0	21.71	21.14	.576 (.407)
Weekly Earnings	0	110.35	104.29	6.059 (4.482)	0	110.66	104.53	6.136 (4.328)
Had Job, Prev. Yr.	.016	.651	.643	.008 (.010)	.019	.653	.646	.007 (.009)
Months Employed, Prev. Yr.	0	3.575	3.516	.058 (.085)	0	3.582	3.518	.064 (.083)
Earnings, Prev.Yr.	.081	2991.8	2873.1	118.65 (109.10)	.084	3020.7	2893.8	126.84 (107.01)
Received Public Benefits	.115	.590	.595	005 (.010)	.118	.582	.590	008 (.010)
Months Received Benefits	.127	6.554	6.542	.012 (.125)	.129	6.469	6.493	024 (.122)
Numbers of Observations	10520	6333	4187		10976	6589	4387	

Table 4.1: Summary Statistics of Baseline Variables (Population ATE of JC)

Note: Z denotes whether the individual was randomly assigned to participate (Z = 1) or not (Z = 0) in the program. Benefits include AFDC/TANF, food stamps, SSI/SSA, and General Assistance. Numbers in parentheses are standard errors. \*\* and \* denote that difference is statistically different from 0 at 5% and 10% level, respectively. Computations use the weights that account for sample and interview design and interview non-response.

The  $LATE_c$  estimates for compliers on earnings, employment and public benefits are 32.29, .055 and -122.28, respectively, 45 percent higher than their corresponding *ITT* estimates. By Assumption 2.2, the sign of  $LATE_c$  identifies the sign of the LATEs for the other two strata. Thus, the estimates of  $LATE_c$  indicate positive population average treatment effects on weekly earnings and employment and a negative population effect on public benefits.

Table 4.2 also shows the proportion of each stratum in the samples. In both of them, the proportion of compliers is the largest, .69, followed by never-takers, .27, and always-takers, .04. And by Assumption 2.2, there are no defiers in the samples. The end part of Table 4.2 reports the point identified averages cited in Assumptions 2.2 and 2.3.<sup>13</sup> These estimates are all statistically significant and follow a certain pattern in both samples: under the treated status, the average outcome for always-takers is the smallest, followed by the average for the mixture of always-takers and compliers, and the average for compliers, while under the untreated status, the average outcome for compliers, and the average for never-takers. Always-takers seem to be the least favorable group despite their strong initiative of participation while never-takers seem to do fine even without participation.

As mentioned previously, differences across these point identified averages may provide a preliminary hint for Assumption 2.3. To begin, the direction of Assump-

<sup>&</sup>lt;sup>13</sup>As in Lee (2009), a *transformed measure* is used to estimate the sample averages of weekly earnings and public benefits to minimize the effect of outliers. Specifically, the entire observed outcome distribution (for either weekly earnings or public benefits) is split into 20 percentile groups  $(5^{th}, 10^{th}, \ldots, 95^{th}, 100^{th})$ , and then the mean outcome within each of the 20 groups is assigned to each individuals.

	Labor-M	Labor-Market-Outcome Sample			enefit Sample				
Variables:	Enrollment	Earnings	Employment	Enrollment	Public benefits				
Averages for $Z = 1$	.730** (.006)	228.78** (3.004)	.608** (.006)	.732** (.005)	747.21** (23.40)				
Averages for $Z = 0$	.043** (.003)	206.60** (3.552)	.570** (.008)	.043** (.003)	831.50** (30.28)				
ITT Effects	.687** (.006)	22.19** (4.652)	.038** (.010)	.689** (.006)	-84.29** (38.27)				
$LATE_c$		32.29** (7.007)	.055** (.015)		-122.28** (56.78)				
Proportions of Strata under Assumptions 1.1 through 1.4									
$\pi_n$	.270** (.006)			.268** (.006)					
$\pi_c$	.687** (.007)			.689** (.006)					
$\pi_a$	.043** (.003)			.043** (.003)					
Other Point Identified	l Average Out	comes under	r Assumptions 1	.1 through 1.4	Ļ				
E[Y(1) a]		132.10** (14.94)	.393** (.037)		545.45** (110.12)				
E[Y(0) n]		223.79** (5.967)	.600** (.012)		880.67** (47.98)				
E[Y(1) c]		236.82** (4.022)	.624** (.008)		707.81** (28.26)				
E[Y(0) c]		204.53** (5.655)	.569** (.012)		830.09** (49.69)				
E[Y Z=1, D=1]		230.63** (3.614)	.611** (.007)		698.35** (25.87)				
E[Y Z=0, D=0]		209.96** (3.709)	.578** (.008)		844.25** (33.18)				

Table 4.2: Point Identified Average Outcomes after Random Assignment

Note: Z denotes whether the individual was randomly assigned to participate (Z = 1) or not (Z = 0) in the program. D denotes whether the individual was ever enrolled in the program (D = 1) or not (D = 0) during the 4 years (208 weeks) after randomization. Benefits include AFDC/TANF, food stamps, SSI/SSA, and General Assistance. Numbers in parentheses are standard errors. \*\* denotes estimate is statistically different from 0 at 5% level. Computations use the weights that account for sample and interview design and interview non-response. The standard errors of LATEs, proportions of strata and other identified average outcomes are calculated by 5000-repetition bootstrap.

tion 2.3c may be inferred by comparing the identified averages of always-takers and never-takers to those of compliers under the same treatment status. The hypotheses that  $E[Y(1)|at] \leq E[Y(1)|c]$  and  $E[Y(0)|c] \leq E[Y(0)|nt]$  are not rejected for all of the three outcomes. Thus, if the same relationship with compliers also hold under the alternative treatment status, Assumption 2.3c is expected to hold. Furthermore, since the *ITT* effect on public benefits is negative, testable implications are available when Assumption 2.3c is combined with Assumption 2.2, as discussed in Subsection 3.2.3. These testable implications are not rejected in the application.

More importantly, indirect evidence of Assumption 2.3 is obtained by comparing the average baseline characteristics across strata. These average characteristics of each stratum are estimated from a nonparametric GMM problem. For each baseline variable, 5 moment functions (4 derived from the conditional expectations defined by  $\{Z, D\}$ plus 1 from the entire sample) are used to identify three (stratum) means. The procedure for estimating this overidentified nonparametric GMM problem is provided in Appendix C. Tables 4.3 and 4.4 show these estimates and their differences across strata for the samples. The average characteristics across strata are similar between the two samples. Among the three strata, never-takers are more likely to be female, older, married, have children, a higher level of education, personal income above \$9,000 (less likely to have personal income below \$3,000), higher weekly earnings at baseline, and to have better labor market outcomes the year before randomization. By contrast, always-takers tend to be male, younger, have a lower level of education at baseline, and have lower earnings in the previous year. The higher earnings of never-takers may be explained by their higher level of education and around one more year of working experience compared with those of always-takers. Related to the identified outcomes of the two strata in Table 4.2, this may explain why never-takers lacked an initiative to participate in the JC. Doing household work and taking care of children may also prevent them from spending extra time training. The statistically significant difference between always-takers and never-takers indirectly supports Assumption 2.3a, while the differences obtained by comparing to compliers (i.e., columns 4 and 5) tend to support Assumption 2.3c (except those in proportion of individuals who have household income above \$18,000 or personal income below \$3,000). When the differences across the three strata are all statistically significant, Assumption 2.3b are more likely to hold. Note that the differences across the strata in the public benefit dependency at the baseline are not statistically significant.<sup>14</sup> Thus, it is concluded from these results that the data do not provide indirect evidence against Assumption 2.3, and that the point estimates of the differences suggest that this assumption is plausible.

# 4.2.3 Empirical Results

Table 4.5 shows the bounds on the population ATEs on the labor market outcomes and the public dependency under Proposition 2.1 through Proposition 2.6. Under each pair of the estimated bounds, a 95% level confidence interval for the true parameter is reported. Since the bounds for weekly earnings and employment in Propositions 2.4 through 2.6 involve max or min operators, this subsection reports the half-median unbiased estimators and the corresponding confidence intervals proposed by Chernozhukov,

<sup>&</sup>lt;sup>14</sup>Unfortunately, information about the amount of public benefits in dollars is unavailable at the baseline.

Variable	nt	С	at	nt-c	c-at	nt – at
Female	.467**	.397**	.324**	.070**	.073**	.143**
Ago at Pacalina	(.011) 18 74**	(.007) 18 32**	(.035) 17 64**	(.015) //28**	(.037) 674**	(.036) 1 102**
Age at Dasenne	(.052)	(.029)	(.133)	(.063)	(.137)	(.143)
White, Non-hispanic	.284**	.263**	.296**	.021*	033	012
Black Non-Hispanic	(.011) 472**	(.000) 484**	(.034) 488**	- 012	- 004	- 016
Didek, Hon Hispanie	(.012)	(.007)	(.037)	(.015)	(.039)	(.039)
Married	$.035^{**}$	$.016^{**}$	.005	.019**	$.011^{**}$	.030**
Has Child	.237**	.162**	.148**	.075**	.015	.089**
	(.010)	(.005)	(.028)	(.012)	(.030)	(.029)
Personal Education	10.27 * *	$10.05^{**}$	$9.637^{**}$	$.224^{**}$	$.408^{**}$	.632**
Household Inc.:<3000	.267**	.255**	.187**	.012	.068**	.080**
	(.008)	(.005)	(.021)	(.010)	(.022)	(.022)
>18000	.181**	$.181^{**}$	$.233^{**}$	.000	052*	052*
Personal Inc.: <3000	.750**	.799**	.843**	049**	044	093**
	(.010)	(.005)	(.026)	(.012)	(.027)	(.027)
>9000	.042** (.005)	.030** (.002)	.015* (.008)	.012*	.015*	.027** (.009)
Have Job	.224**	.208**	.216**	.015	008	.008
,, ,_ , ,	(.010)	(.006)	(.031)	(.012)	(.033)	(.032)
Weekly Hrs.Worked	22.07** (.488)	21.29** (.272)	20.44** (1.652)	.775 (.585)	.853 (1.734)	1.629 (1.700)
Weekly Earnings	113.79**	102.76**	92.63**	11.03**	10.13	21.15**
	(2.989)	(2.041)	(7.986)	(3.989)	(8.328)	(8.562)
During the Year Prior to	Random Ass	signment				
Had Job	.667**	.640**	.651**	.027**	010	.016
	(.010)	(.006)	(.035)	(.013)	(.036)	(.035)
Mths.Worked	3.684**	3.527**	3.120**	.157	.407	.563*
Earnings	3246.8**	2831.5**	2302.9**	415.30**	528.64**	943.94**
-	(101.80)	(63.58)	(251.57)	(127.99)	(263.42)	(273.94)
Received Benefits	.607**	.588** (.006)	.596** (.037)	.020 (.013)	009	.011
Mths. Received	6.744**	6.503**	6.518**	.240	014	.226
	(.122)	(.073)	(.414)	(.155)	(.457)	(.424)

Table 4.3: Average Baseline Characteristics in the Sample for Labor Market Outcomes

Note: Benefits include AFDC/TANF, food stamps, SSI/SSA, and General Assistance. Numbers in parentheses are standard errors. \*\* and \* denote that estimate is statistically different from 0 at 5% and 10% level, respectively. Computations use the weights that account for sample and interview design and interview non-response. Missing values for each of the baseline variables were imputed with the mean of the variable.

Variable	nt	С	at	nt-c	c-at	nt - at	
Female	.464**	.396**	.330**	.069**	.066*	.134**	
	(.011)	(.006)	(.035)	(.014)	(.037)	(.036)	
Age at Baseline	18.75**	18.31**	17.68**	.435**	.635**	1.070**	
	(.049)	(.027)	(.126)	(.061)	(.135)	(.133)	
White, Non-hispanic	.289**	.265**	.289**	.024*	024	000	
	(.011)	(.006)	(.035)	(.014)	(.037)	(.036)	
Black, Non-Hispanic	.461**	.480**	.503**	019	023	042	
	(.012)	(.007)	(.037)	(.015)	(.039)	(.039)	
Married	.036**	.016**	.006	.020**	.010**	.030**	
	(.004)	(.002)	(.005)	(.005)	(.005)	(.006)	
Has Child	.234**	.163**	.164**	.072**	001	.071**	
	(.009)	(.005)	(.029)	(.012)	(.031)	(.030)	
Personal Education	10.27** (.034)	10.05** (.020)	9.663** (.091)	.225** (.043)	.382** (.096)	.607** (.094)	
Household Inc.:<3000	.262**	.253**	.198**	.009	.055**	.064**	
	(.008)	(.004)	(.020)	(.010)	(.022)	(.021)	
>18000	.184**	.184**	.233**	.000	050*	049*	
	(.007)	(.004)	(.028)	(.009)	(.029)	(.028)	
Personal Inc.: <3000	.746**	.797**	.840**	051**	043*	094**	
	(.010)	(.005)	(.024)	(.012)	(.026)	(.025)	
>9000	.042**	.030**	.015**	.012**	.015*	.027**	
	(.005)	(.002)	(.007)	(.006)	(.008)	(.009)	
Have Job at Baseline	.227**	.211**	.213**	.016	002	.014	
	(.010)	(.005)	(.028)	(.012)	(.030)	(.029)	
Weekly Hrs. Worked	21.80**	21.41**	20.63**	.392	.774	1.165	
	(.460)	(.291)	(1.426)	(.594)	(1.548)	(1.494)	
Weekly Earnings	112.60**	103.55**	94.21**	9.025**	9.342	18.37**	
	(2.890)	(2.180)	(7.394)	(4.094)	(7.954)	(7.804)	
During the Year Prior to Random Assignment							
Had Job	.667**	.642**	.668**	.025*	026	001	
	(.011)	(.006)	(.031)	(.013)	(.033)	(.032)	
Mths. Employed	3.644**	3.553**	3.060**	.091	.492	.584*	
	(.103)	(.057)	(.282)	(.130)	(.302)	(.299)	
Earnings	3241.9**	2863.6**	2390.4**	378.31**	473.14*	851.45**	
	(99.19)	(65.20)	(233.19)	(130.21)	(250.73)	(249.72)	
Received Benefits	.601**	.581**	.583**	.020	001	.019	
	(.010)	(.006)	(.033)	(.013)	(.035)	(.034)	
Mths. Received	6.684**	6.433**	6.395**	.251	.038	.289	
	(.122)	(.076)	(.378)	(.158)	(.408)	(.385)	

Table 4.4: Average Baseline Characteristics in the Sample for Public Assistance Benefits

Note: Benefits include AFDC/TANF, food stamps, SSI/SSA, and General Assistance. Numbers in parentheses are standard errors. \*\* and \* denote that estimate is statistically different from 0 at 5% and 10% level, respectively. Computations use the weights that account for sample and interview design and interview non-response. Missing values for each of the baseline variables were imputed with the mean of the variable.

	Earnings	Employment	Public Benefits		
	LB UB	LB UB	LB UB		
Bounds under Each Assum	otion				
Proposition 2.1	-69.86 201.02	150 .163	-632.86 1812.4		
Bounded Outcome (A2.1)	(-78.34, 210.61)	(167, .179)	(-702.21, 1901.6)		
Proposition 2.2	22.19 201.02	.038 .163	-632.86 -84.29		
Monotonicity (A2.2)	(14.18, 210.61)	(.021, .179)	(-702.21, -22.13)		
Proposition 2.3a	-6.507 201.02	027 .163	-188.43 1812.4		
Mean Dominance (A2.3a)	(-16.65, 210.61)	(050, .179)	(-265.90, 1901.6)		
Proposition 2.3b	20.67 201.02	.033 .163	-145.90 1812.4		
Mean Dominance (A2.3b)	(11.97, 210.61)	(.015, .179)	(-212.69, 1901.6)		
Proposition 2.3c	22.57 201.02	.037 .163	-142.76 1812.4		
Mean Dominance (A2.3c)	(13.72, 210.61)	(.019, .179)	(-210.62, 1901.6)		
Bounds under Combined As	ssumptions				
Proposition 2.4	20.43 201.02	.034 .163	-188.43 -84.29		
(A2.1, A2.2, A2.3a)	(13.01, 210.58)	(.018, .180)	(-265.95, -22.09)		
Proposition 2.5	22.97 201.01	.039 .163	-145.90 -84.29		
(A2.1, A2.2, A2.3b)	(14.53, 210.56)	(.020, .180)	(-213.01, -21.83)		
Proposition 2.6	24.61 201.04	.042 .163	-142.76 -84.29		
(A2.1, A2.2, A2.3c)	(16.01, 210.59)	(.023, .180)	(-210.62, -22.13)		

 Table 4.5: Bounds on the Population Average Treatment Effects

Note: Benefits include AFDC/TANF, food stamps, SSI/SSA, and General Assistance. Numbers in parentheses are 95% level confidence intervals for true parameters of interest. The confidence intervals of the bounds under each assumption are calculated by the method of Imbens and Manski (2004). For earnings and employment, the confidence intervals of the bounds under combined assumptions are calculated by the method of Chernozhukov, Lee and Rosen (2011), while the bounds under combined assumptions are estimated by their half-median unbiased estimators. For public benefits, the confidence intervals are calculated by the method of Imbens and Manski (2004). Computations use the weights that account for sample and interview design and interview non-response. Standard errors are calculated by 5000-repetition bootstrap.

Lee, and Rosen (2011). The bounds without max or min operators are estimated with sample analogs and their confidence intervals are obtained by the method of Imbens and Manski (2004).<sup>15</sup>

<sup>15</sup>Specifically, the Imbens and Manski (2004) confidence interval at 95% level is calculated from  $(\widehat{\Delta}_{LB} - \overline{C}_n * \widehat{\sigma}_{LB} / \sqrt{N}, \widehat{\Delta}_{UB} + \overline{C}_n * \widehat{\sigma}_{UB} / \sqrt{N})$ , where  $\widehat{\sigma}_{LB} = \sqrt{\widehat{V(\widehat{\Delta}_{LB})}}$ ,  $\widehat{\sigma}_{UB} = \sqrt{\widehat{V(\widehat{\Delta}_{UB})}}$ , and  $\overline{C}_n$  satisfies  $\Phi(\overline{C}_n + \sqrt{N}(\widehat{\Delta}_{UB} - \widehat{\Delta}_{LB}) / \max(\widehat{\sigma}_{LB}, \widehat{\sigma}_{UB})) - \Phi(-\overline{C}_n) = .95$ .
Let me begin with the ATE on weekly earnings in the first two columns. Proposition 2.1 provides the bounds in the AIR setting under the bounded-outcome assumption (A2.1). The estimated bounds are rather wide and fail to identify the sign of the ATE. Note that these bounds are also the IV bounds proposed by Manski (1990), Heckman and Vytlacil (2000), Kitagawa (2009) and Huber and Mellace (2010). The ATE in Proposition 2.2 under the monotonicity assumption (A2.2) is bounded between [22.19, 201.02], obtained by identifying positive *LATEs* for always-takers and never-takers. Note that they are also the ones under the IV and MTR assumptions proposed by Manski and Pepper (2000). The mean dominance assumptions (A2.3) improve upon the lower bound in Proposition 2.1, and negative effects are ruled out in Propositions 2.3b and 2.3c in the absence of inferring the sign of the  $LATE_a$  and  $LATE_n$  from that of the  $LATE_c$ . When Assumptions 1.1 through 1.4, and 2.1 through 2.3 are all imposed together, all of the bounds and their corresponding confidence intervals lie in the positive region. In particular, the bounds on the ATE on the weekly earnings in Proposition 2.6 are the narrowest, [24.61, 201.04], with the lower bound 10 percent higher than the *ITT* effect (22.19), while the  $LATE_c$  for compliers (32.29) falls within the bounds. It turns out that inferring the unobserved terms (i.e., E[Y(0)|a] and E[Y(1)|n]) from the point identified outcomes of compliers (A2.3) provides a sharper lower bound on the ATE on the weekly earnings than that obtained by identifying the sign of  $LATE_a$  and  $LATE_n$  under the monotonicity assumption (A2.2).

The next two columns show the bounds on the ATE on employment, whose value is between 0 and 1. A similar pattern to the bounds for the weekly earnings is also found in the bounds for employment. Note that the identified set in Proposition 2.1, [-.015, .163], is unable to identify the sign of the ATE, and that it also coincides with those in Balke and Pearl (1997) for the case of a binary outcome. The bounds in Proposition 2.2 indicate a positive ATE, which varies between [.038, .163], and also equal the ones proposed by BSV (2008), Chesher (2010), Chiburis (2010b) and Shaikh and Vytlacil (2011), all of which analyze a binary outcome. Again, Proposition 2.6 provides the narrowest bounds on the ATE on employment under Assumptions 2.1, 2.2 and 2.3c, [.042, .163], with the lower bound 10 percent higher than the ITT effect (.038), while the  $LATE_c$  for compliers (.055) falls within the bounds in Proposition 2.6.

The final two columns report the bounds on the ATE on public benefits. Different from the labor market outcomes, since the *ITT* effect on public benefits is negative, imposing only the monotonicity assumption improves upon its upper bound in Proposition 2.1, while imposing only the mean dominance assumptions improves upon its lower bound. The bounds in Proposition 2.1 are extremely wide and uninformative. The monotonicity assumption (A2.2) has strong identification power compared with the mean dominance assumptions (A2.3) in the case of the public benefits, in that the former identifies the negative sign of the ATE on the public benefits, though the latter greatly sharpens the lower bound in Proposition 2.1 by at least 70 percent. However, once the two types of assumptions are taken into account together, the bounded outcome assumption is no longer necessary and the width of the bounds shrinks dramatically. Under the combined assumptions, the estimated bounds and their corresponding confidence intervals lie in the negative region. Proposition 2.6 provides the narrowest bounds on the ATE on public benefits, [-142.76, -84.29], with the upper bound equal to the ITT effect, while the  $LATE_c$  on compliers (-122.28) falls within the bounds.

I draw the following conclusions from the empirical results of the population average treatment effects of JC enrollment. Focusing on the labor market outcomes and welfare dependence, the monotonicity assumption (A2.2) and the mean dominance assumption (A2.3) together provide the narrowest bounds on the  $ATE_s$ . In particular, when the former is combined with Assumption 2.3c, these bounds are [24.61, 201.04] for weekly earnings, [.042, .163] for employment, and [-142.76, -84.29] for public benefits. These bounds are significantly narrower than the IV bounds proposed by Manski (1990), Heckman and Vytlacil (2000), Kitagawa (2009) and Huber and Mellace (2010). The width of these bounds is also smaller than that under the IV and MTR assumptions of Manski and Pepper(2000), especially for public benefits. The bounds on employment are also narrower than the ones proposed by Balke and Pearl (1997), BSV (2008), Chesher (2010), Chiburis (2010b) and Shaikh and Vytlacil (2011) for the case of a binary outcome. The lower bounds for weekly earnings and employment are 10 percent higher than their respective intention-to-treat (ITT) effects (22.19 and .038), while the upper bound for public benefits is equal to its ITT effect. The LATEs for compliers on the three outcomes also fall within these narrowest bounds. In sum, these empirical results suggest that enrolling into the Job Corps program increases weekly earnings by at least \$24.61 and employment by at least 4.3 percentage points four years after randomization, and decreases yearly dependence on public welfare benefits by at least \$84.29.

## 4.3 The Wage Effect of JC Enrollment

#### 4.3.1 Specific Sample

Assessing the effect of training programs on wages is of great importance to policy makers. Further analysis of the wage effects of the Job Corps program answers an important question of whether the program increases participants' human capital accumulation and thus leads to an increase in their wages. To compare with the wage effects of JC in the absence of addressing the noncompliance issue, this section employs the same sample from the NJCS used by Lee (2009). This sample involves only individuals with non-missing values for weekly earnings and weekly hours worked for every week after random assignment (9,145 individuals). I construct the data set by adding enrollment information at Week 208 (i.e., 48 months) after random assignment. Again, this binary variable indicates whether or not the individual was ever enrolled in JC during the 208 weeks after random assignment. 55 observations are dropped from Lee's sample due to the missing enrollment variable, resulting in the final sample of 9,090 individuals (3,599 and 5,491 individuals in the control and treatment groups, respectively). Wages at Week 208 are obtained by dividing weekly earnings by weekly hours worked at that week. An individual is regarded as unemployed when the wage is missing, and as employed otherwise.

The first four columns of Table 4.6 report the average baseline characteristics of the entire sample by treatment assignment status, along with the percentage of missing values for each of those variables. The pre-treatment variables include demographic characteristics, education and background variables, employment and earnings information at the baseline, and labor market outcomes in the year prior to randomization. The average pre-treatment characteristics of the treatment and control groups are similar as expected, with the difference in means being statistically different from zero at the five percent level only for weekly hours worked at baseline. Thus, this sample maintains the balance of baseline variables between the treatment and control groups.

The first three columns of Table 4.7 show the averages of some relevant post-treatment variables based on treatment assignment status, along with their differences, at Week 208 after randomization. The first row shows information on the JC enrollment variable. By Week 208, 73.8 percent of those assigned to the treatment group and 4.4 percent of the control group had ever enrolled in JC.<sup>16</sup> The difference in these two numbers, which equals the proportion of compliers in the population, is 69.4 percent. The rest of the rows in Table 4.7 present the *ITT* effect and *LATE* for JC compliers on various labor market outcomes at Week 208. All the *ITT* effects of JC on weekly hours worked, weekly earnings, and employment are positive and statistically significant. The *LATE* estimates for those three variables are also positive and statistically significant, and they are larger than the *ITT* estimates by about 44, 44 and 50 percent, respectively. The estimated average effects of JC on earnings and employment for compliers is 39.9 dollars and 6 percentage points, respectively. These results are consistent with the findings in the NJCS (Burghardt et al., 2001).

For reference, Table 4.7 also shows the estimated ITT and LATE effects of JC on ln(wage) for employed individuals. The *LATE* estimate implies an average effect of JC on ln(wage) of about 5.4 percent for compliers. However, these estimates are biased

<sup>&</sup>lt;sup>16</sup>From these controls, 3.2 percent enrolled after the end of the embargo period, while 1.2 percent of them enrolled in the program despite the three-year embargo imposed on them.

		Entire	Sample			Non-F	lispanics	
	Missing Prop.	Z=1	Z=0	Diff.(Std.Err.)	Missing Prop.	Z=1	Z=0	Diff.(Std.Err.)
Female	0	.454	.458	004 (.011)	0	.454	.454	010 (.012)
Age at Baseline	0	18.44	18.35	.087* (.046)	0	18.44	18.34	.096* (.050)
White, Non-hispanic	0	.265	.263	.002 (.009)	0	.319	.318	.001 (.011)
Black, Non-Hispanic	0	.494	.491	.003 (.011)	0	.595	.593	.002 (.012)
Hispanic	0	.169	.172	003 (.008)	_	_	—	_
Other Race/Ethnicity	0	.072	.074	002 (.006)	0	.087	.089	003 (.007)
Never married	.017	.917	.916	.002 (.006)	.018	.926	.924	.002 (.006)
married	.017	.020	.023	003 (.003)	.018	.015	.018	003 (.003)
Living together	.017	.039	.040	002 (.004)	.018	.035	.037	002 (.004)
Separated	.017	.024	.021	.003 (.003)	.018	.023	.020	.003 (.003)
Has Child	.007	.189	.193	004 (.008)	.006	.187	.190	004 (.009)
Number of children	.010	.270	.268	.002 (.014)	.180	.269	.271	003 (.015)
Personal Education	.018	10.12	10.11	.013 (.033)	.018	10.14	10.12	.022 (.036)
Mother's Education	.188	11.49	11.46	.030 (.061)	.182	11.81	11.83	021 (.055)
Father's Education	.383	11.40	11.54	145* (.077)	.379	11.72	11.86	147** (.072)
Ever Arrested	.017	.248	.249	001 (.009)	.017	.255	.257	002 (.010)
Household Inc.: <3000	.358	.253	.251	.001 (.012)	.357	.248	.244	.004 (.013)
3000-6000	.358	.205	.208	003 (.011)	.357	.202	.213	012 (.012)
6000-9000	.358	.117	.114	.003 (.009)	.357	.119	.105	.015 (.009)
9000-18000	.358	.246	.245	.001 (.011)	.357	.244	.248	003 (.013)
>18000	.358	.180	.182	002 (.010)	.357	.187	.191	003 (.012)
Personal Inc.: <3000	.079	.788	.789	001 (.009)	.077	.787	.788	001 (.010)
3000-6000	.079	.128	.131	003 (.008)	.077	.129	.136	007 (.008)
6000-9000	.079	.053	.046	.007 (.005)	.077	.052	.043	.009* (.005)
>9000	.079	.031	.034	003 (.004)	.077	.031	.033	001 (.004)
At Baseline:								
Have job	.021	.198	.192	.007 (.009)	.021	.204	.187	.017* (.009)
Weekly hours worked	0	21.83	20.91	.922** (.447)	0	21.97	20.75	1.216** (.491)
Weekly earnings	0	111.08	102.89	8.183 (5.134)	0	107.79	102.28	5.516** (2.804)
Had job, Prev. Yr.	.017	.635	.627	.008 (.010)	.017	.642	.627	.015 (.011)
Months employed, Prev. Yr.	0	3.603	3.530	.074 (.091)	0	3.654	3.512	.143 (.100)
Earnings, Prev.Yr.	.062	2911.0	2810.5	100.56 (117.58)	.064	2900.3	2794.7	105.57 (106.34)
Numbers of observations	9090	5491	3599		7529	4551	2978	

### Table 4.6: Summary Statistics of Baseline Variables (Wage Effect of JC)

Note: Z denotes whether the individual was randomly assigned to participate (Z = 1) or not (Z = 0) in the program. Numbers in parentheses are standard errors. \*\* and \* denote that difference is statistically different from 0 at 5% and 10% level, respectively. Computations use design weights.

	Entire Sample			N	Non-Hispanics			
	Z = 1	Z = 0	Diff. (Std.Err.)	Z = 1	Z = 0	Diff. (Std.Err.)		
Enrollment Variable								
Ever enrolled in JC	.738	.044	.694** (.007)	.737	.047	.689** (.008)		
Intention-to-Treat $(ITT)$ Effects								
Hours per week	27.80	25.83	1.967** (.559)	28.05	25.53	2.523** (.617)		
Earnings per week	228.19	200.50	27.69** (5.121)	230.22	194.66	35.57** (5.555)		
Employed	.607	.566	.041** (.011)	.609	.559	.050** (.012)		
ln(wage)	2.029	1.991	.038** (.011)	2.028	1.977	.050** (.013)		
Local ATE for Compliers, LATE (IV estimates)								
Hours per week			2.834** (.782)			3.661** (.867)		
Earnings per week			39.90** (6.457)			51.62** (6.960)		
Employed			.060** (.015)			.072** (.017)		
ln(wage)			.054** (.016)			.073** (.017)		

Table 4.7: Summary Statistics of Post-treatment Variables by Random Assignment

Note: Z denotes whether the individual was randomly assigned to participate (Z = 1) or not (Z = 0) in the program. Numbers in parentheses are standard errors. \*\* denotes that difference is statistically different from 0 at 5% level. Computations use design weights. The standard error of the effect for employed compliers is calculated by an ML estimator, where the endogenous dummy variable is the treatment receipt indicator. The treatment assignment indicator is used as the exclusion restriction and all baseline characteristics (where mean values were imputed for missing values) are included in both the selection and outcome equations.

because of sample selection.

# 4.3.2 Assessment of Assumptions

Since Assumptions 3.1 through 3.4 have been previously used in the NJCS to estimate the effect of JC on labor market outcomes that are not affected by sample selection (Burghardt et al., 2001; Schochet, 2001; Schochet, Burghardt, and Glazerman, 2001), here the discussion is concentrated on the plausibility of Assumptions 3.5 and 3.6.

Assumption 3.5 states that there is a non-negative effect of JC enrollment on employment for every individual at Week 208. This assumption seems plausible in the application given the objectives and services provided by JC (e.g., academic and vocational training, job search assistance). A testable implication of this assumption is that the *LATE* for JC compliers on employment is non-negative. As discussed above, this effect is positive and highly statistically significant.

There are two potential threats to the validity of Assumption 3.5. First, individuals who enroll in JC may be less likely to be employed while undergoing training than those who do not enroll, which is usually referred to as the "lock-in" effect (van Ours, 2004). Second, trained individuals may raise their reservation wages because of the JC training, which may lead them to reject some job offers that they would otherwise accept if they had not received training. Both potential threats, however, are likely to become less relevant in the long run, as trained individuals are no longer "locked-in" away from employment, and individuals who chose to remain unemployed in the short run because of raising their reservation wages find jobs in the long run. Consistent with this view, Schochet, Burghardt, and Glazerman (2001), and Lee (2009) find negative effects of JC on employment in the short run, and positive effects in the long run. Thus, the analysis focuses on wages at Week 208 after random assignment, which is the latest wage measure available in the NJCS.

Based on a likelihood-based analysis, Frumento et al. (2012) provide evidence that there may be a positive proportion of compliers in the population for whom JC has a negative effect on employment at Week 208, even though this proportion falls over time after randomization. Therefore, to further increase the plausibility of Assumption 3.5 in the application, a sample that excludes Hispanics is also considered. Hispanics were the only demographic group in the NJCS for which negative (although statistically insignificant) effects of JC on both employment and earnings were found (Schochet, Burghardt, and Glazerman, 2001; Flores-Lagunes, Gonzalez, and Neumann, 2010). Flores-Lagunes, Gonzalez and Neumann (2010) investigate this issue and find that the different outcomes of white, blacks and Hispanics are strongly related to the different local labor market conditions they face. In particular, they find evidence that during that period Hispanics faced worst local unemployment rates than blacks and whites. Therefore, Assumption 3.5 may not be appropriate for Hispanics. The last set of columns in Tables 4.6 and 4.7 present summary statistics of pre- and post-treatment variables for the Non-Hispanics sample (7,529 individuals). As expected, the estimated ITT and LATEeffects of JC on labor outcomes for Non-Hispanics are stronger than those for the entire sample, with a statistically significant average effect on employment for compliers of 7.2 percentage points.

Assumption 3.6 states that the mean potential outcome under treatment of the alwaysemployed compliers (*cEE* stratum) is greater than or equal to that of individuals who would be employed only if they enrolled in JC (*cNE* stratum). As discussed before, it is possible to indirectly shed some light on the plausibility of this assumption by comparing average baseline characteristics of the *cEE* and *cNE* strata that are likely to be highly correlated to wages at Week 208. Appendix Table C.1 presents the average characteristics of two strata for the entire and Non-Hispanics samples, obtained by estimating the overidentified GMM procedure described in Appendix C. Focusing on the Non-Hispanics sample, relative to individuals in the cNE stratum, individuals in the cEE stratum are more likely to be male and white, to have never been arrested at the baseline, and to have better labor market outcomes the year before randomization. These differences, however, are not statistically different from zero. Thus, it is concluded that the data do not provide indirect evidence against Assumption 3.6, and that the point estimates of the differences suggest that the assumption is plausible.

Related to their plausibility, I also check the sensitivity of the bounds in Propositions 3.1 and 3.2 to these assumptions. Subsection 3.3.4 provides simulation exercises to examine how the bounds behave when Assumptions 3.5 and 3.6 fail.

### 4.3.3 Empirical Results

This analysis begins with bounding the average effect of being allowed to enroll in JC on wages (*ITT* effect) for the individuals who would always be employed regardless of treatment *assignment*, and then bound the average effect of JC enrollment on wages for those compliers who would always be employed regardless of the treatment *receipt* ( $\Delta$  in (3.3)). The first parameter is the one considered in Lee (2009) and ZRM (2008), which ignores the noncompliance issue. In their setting, the principal strata are *EE*, *NN*, *NE*, and *EN*, where the last stratum is ruled out by assuming the monotonicity of *S* in *Z*.

Table 4.8 presents bounds on the average ITT effect of JC on ln(wage) for alwaysemployed individuals (*EE* stratum). The first column of Table 4.8 presents results for the entire sample. The proportion of the always-employed individuals (*EE*) in the population equals 56.6 percent, and the proportion of those in the *NE* stratum (which equals the *ITT* effect of JC on employment) equals 4.1 percent. Under the monotonicity of *S* in *Z* assumption, the estimated lower and upper bounds for the *ITT* effect of JC on ln(wage) for the *EE* stratum are -.022 and .100, respectively. These results are very similar to those obtained by Lee (2009).<sup>17</sup> As noted by Lee (2009), although the bounds include zero, they rule out plausible negative effects. Also as noted by Lee (2009), these particular lower bounds are based on the extreme and unintuitive assumption that wages are perfectly negatively correlated with the probability of being employed.<sup>18</sup> This is contradicted by standard models of labor supply, in which individuals with higher wages are also more likely to be employed.

The last set of rows in Table 4.8 present the bounds on the *ITT* effect of JC on ln(wage) for the *EE* stratum under the mean dominance assumption that the average potential wage under z = 1 of the *EE* stratum is greater than that of the *NE* stratum. This assumption can be seen as a way to rule out the implausible extreme case mentioned above by implying a positive correlation between wages and employment. In this case, the estimated lower bound for the *ITT* effect for the *EE* stratum is .038. Thus, under this additional assumption, the bounds are able to rule out the negative *ITT* effect of JC on wages, which illustrates the identifying power of this assumption. Table 4.8 also presents 95 percent confidence intervals, which are calculated based on the results from

<sup>&</sup>lt;sup>17</sup>The results are not numerically the same as those in Lee (2009) who uses a transformed wage variable to calculate bounds to minimize the effect of outliers by splitting the entire observed wage distribution into 20 percentile groups ( $5^{th}$ ,  $10^{th}$ , ...,  $95^{th}$ ,  $100^{th}$ ), and then assigning the mean wage within each of the 20 groups to each individual. Here the original wage variable is used, and 55 observations are dropped from Lee's sample because of missing enrollment information. For reference, the corresponding lower and upper bounds in Lee (2009) are -.019 and .093, respectively.

<sup>&</sup>lt;sup>18</sup>To obtain the lower bound all the *EE* individuals are placed at the bottom of the wage distribution in the cell  $\{Z = 1, S = 1\}$ , which is a mixture of the *EE* and *NE* strata.

	Entire Sample	Non-Hispanics				
Proportions of strata:						
Always-employed( $EE$ )	.566** (.009)	.559** (.009)				
Never-employed( $NN$ )	.393** (.007)	.391** (.007)				
Employed only if assigned to $program(NE)$	.041** (.011)	.050** (.012)				
E[Y(Z=0) EE]	1.991** (.009)	1.977** (.010)				
Proportion of <i>EE</i> in cell $\{Z = 1, S = 1\}$	.932** (.017)	.918** (.019)				
Bounds with Monotonicity						
Lower bound for the $ITT$ effect for $EE$ stratum	022 (.016)	018 (.017)				
Upper bound for the $ITT$ effect for $EE$ stratum	.100** (.014)	.119** (.015)				
Imbens and Manski 95% confidence interval	[048, .123]	[047, .144]				
Bounds with Monotonicity and Mean Dominance						
Lower bound for the effect for $EE$ stratum	.038** (.012)	.050** (.013)				
Upper bound for the $ITT$ effect for $EE$ stratum	.100** (.014)	.119** (.015)				
Imbens and Manski 95% confidence interval	[.019, .123]	[.029, .144]				

Table 4.8: Bounds for the *ITT* Effects on ln(wage) at Week 208

Note: Numbers in parentheses are standard errors. \*\* denotes that estimate is statistically different from 0 at 5% level. Computations use design weights. The standard error is calculated by a 5,000-repetition bootstrap. Imbens and Manski 95% confidence interval is calculated as  $(\widehat{\Delta}_{LB}-1.645*\widetilde{\sigma}_{LB},\widehat{\Delta}_{UB}+1.645\widetilde{\sigma}_{UB})$ , where  $\overline{C}_n=1.645$  and  $\widetilde{\sigma}_{LB}$  and  $\widetilde{\sigma}_{UB}$  are calculated by bootstrap.

Imbens and Manski (2004) and asymptotically cover the true value of the parameter with .95 probability.<sup>19</sup> As above, while the 95 percent confidence intervals do not rule out a zero effect under the monotonicity assumption, they rule out negative effects once the mean dominance assumption is employed.

The second column of Table 4.8 presents results for Non-Hispanics, for whom the monotonicity assumption of S in Z is more plausible. In general, the lower and upper bounds under the two sets of assumptions are larger for Non-Hispanics than for the entire

<sup>&</sup>lt;sup>19</sup>Imbens and Manski (2004) confidence intervals are valid for the *ITT* effect of JC on wages because the bounding functions do not involve min or max operators.

population. Under the monotonicity and mean dominance assumptions, the estimated lower and upper bounds on the *ITT* effect of JC on  $\ln(\text{wage})$  for the *EE* stratum are .050 and .119, respectively, and the 95 percent confidence interval is [.029, .144].

Table 4.9 shows the estimation results for the parameter of interest, the average effect of JC enrollment on ln(wage) for always-employed compliers. This table shows the estimated proportions, relevant quantities used in estimating the bounds, and the unbiased half-median estimators for the bounds and the confidence intervals for true parameters proposed by Chernozhukov, Lee, and Rosen (2011) (hereafter, CLR). As above, the first column presents the results for the entire sample, and the second shows the results for Non-Hispanics. For both samples, the largest stratum is the *cEE* stratum, with an estimated proportion of almost 40 percent, while the stratum of always-employed always-takers (*aEE*) is the smallest stratum, with an estimated proportion of about 1.6 and 1.8 percent for the entire and Non-Hispanics samples, respectively. The estimated proportion of always-takers ( $\pi_{aEE} + \pi_{aNN}$ ) is 4.4 percent for the entire sample, while the proportion of never-takers is 26.2 percent. These proportions are slightly higher for the Non-Hispanics sample. All the estimated stratum proportions in Table 4.9 are statistically different from zero.

For the entire population, the estimated lower and upper bounds on the average effect of JC on ln(wage) for the *cEE* stratum under Assumptions 3.1 through 3.5 are -.022and .130, respectively, while the corresponding numbers for Non-Hispanics are -.014and .161. Given the weak effects of JC on labor market outcomes for Hispanics, it is not surprising that the bounds for Non-Hispanics cover a larger positive region than those for the entire population. For both samples, the estimated lower and upper bounds are larger

Entire Sample	Non-Hispanics
9090	7529
.016** (.002)	.018** (.002)
.158** (.005)	.160** (.005)
.391** (.010)	.381** (.011)
.041** (.011)	.050** (.012)
.028** (.003)	.030** (.003)
.104** (.004)	.104** (.005)
.261** (.007)	.258** (.008)
.872** (.023)	.849** (.025)
2 033** ( 059)	2 016** ( 061)
2.033** (.016)	2.010 (.001)
1.972** (.015)	1.952** (.016)
2 420** ( 0.00)	0.000
2.429** (.066)	2.3/6** (.05/)
1.676** (.034)	1.703** (.035)
tion 3.1)	
[1.951, 2.102]	[1.938, 2.113]
(1.921, 2.128)	(1.907, 2.140)
[022, .130]	[014, .161]
(061, .168)	(057, .201)
n Dominance (Propo	sition $3.2$
[2.027, 2.102]	[2.026. 2.113]
(2.011, 2.129)	(2.008, 2.141)
[.055, .130]	[.074, .161]
(.023, .170)	(.039202)
	Entire Sample 9090 .016** (.002) .158** (.005) .391** (.010) .041** (.011) .028** (.003) .104** (.004) .261** (.007) .872** (.023) 2.033** (.016) 1.972** (.015) 2.429** (.066) 1.676** (.034) fion 3.1) [1.951, 2.102] (1.921, 2.128) [022, .130] (061, .168) m Dominance (Propo. [2.027, 2.102] (2.011, 2.129) [.055, .130] (.023, .170)

Table 4.9: Bounds for the Effects of JC on ln(wage) for the cEE Stratum

Note: \*\* denotes estimate is statistically different from 0 at 5% level and numbers in parentheses are standard errors. Computations use design weights. The standard error is calculated by a 5,000-repetition bootstrap. Numbers in square brackets are half-median unbiased estimators for the bounds, while numbers below them are CLR 95% level confidence intervals.

than the corresponding bounds for the ITT effect presented in Table 4.8, especially the upper bound (e.g., for Non-Hispanics, .119 versus .161, or a 35.3 percent increase). From Tables 4.8 and 4.9, the positive region covered by the bounds on the effect of JC enrollment on wages for the *cEE* stratum is larger than the positive region covered by the bounds on the *ITT* effect of JC on wages for the *EE* stratum. This suggests that the effects of JC on wages obtained by Lee (2009) were conservative, as the effect was weakened by noncompliance to the assigned treatment.

As in Lee (2009), the bounds in Proposition 3.1 are unable to rule out zero effects of JC on wages at Week 208 employing only the monotonicity assumption on the effect of JC on employment. However, as before, the lower bounds are constructed under the implausible "worst-case" scenario of a perfect negative correlation between employment and wages, which is contradicted by standard economic models. The mean dominance assumption rules out this implausible extreme case and helps to increase the lower bound. The last set of rows in Table 4.9 show results when the monotonicity and mean dominance assumptions are both used. Under Assumptions 3.1 through 3.6, the estimated lower bound on the average effect of JC on ln(wage) for the *cEE* stratum is .055 for the entire population, and it is .074 for Non-Hispanics. Therefore, under all six assumptions, the results imply positive average effects of JC on wages for the *cEE* stratum in both the entire and Non-Hispanics samples. These results also reinforce the notion that the *ITT* effects of JC on wages are likely to be lower than the effect of JC enrollment on wages. Also note that, as already mentioned in Remark 3, the lower bound for  $\Delta$  in Proposition 3.2 can be interpreted as the lower bound for  $ATE_{cEE,cNE}$  under Assumptions 3.1 through 3.4 and 3.5".

The following conclusions are derived from the empirical analysis of the effects of JC on wages. First, the results in the subsection strongly suggest a positive average effect of participating in JC on wages four years after random assignment for the always employed compliers. For Non-Hispanics, for whom the monotonicity assumption on the effect of JC on employment is more likely to hold, the estimated bounds under Assumptions 3.1 through 3.6 imply that, on average, JC enrollment increases the average wage (as oppose to ln(wage)) of the always-employed compliers who participate in JC by at least 7.7 and at most 17.5 percent. Therefore, this evidence suggests that JC has an effect on participants' earnings not only by increasing their probability of being employed but also by increasing their wages, which is most likely a consequence of their human capital accumulation during enrollment in JC.

Second, the analysis implies that the results from the study of the *ITT* effects of JC on wages (e.g., Lee, 2009) are conservative because the noncompliance issue is likely to dilute the effect of JC enrollment on wages. In particular, for the two samples, and regardless of whether or not Assumption 3.6 is employed, the positive region covered by the bounds on the effect of JC enrollment on wages for the *cEE* stratum is larger than the positive region covered by the bounds on the effect of ythe bounds on the *ITT* effect of JC on wages for the *EE* stratum. This is consistent with the results presented in Subsection 4.3.1, regarding the effect of JC on other labor market outcomes not suffering from sample selection, which show that the *LATE* estimates of the effects are larger than the *ITT* estimates. This conclusion is also consistent with the literature on point estimation of the wage effects of JC.<sup>20</sup>

<sup>&</sup>lt;sup>20</sup>Frumento et al. (2012) find that their point estimate of the effect of JC enrollment on wages at week 208 for the always-employed compliers is larger than the point estimate of the *ITT* effect in Zhang, Rubin,

Finally, to analyze the sensitivity of the results for missing values of relevant variables, and to compare the results to those in Frumento et al. (2012), I estimate the bounds on  $\Delta$  that account for this issue. I employ another weight constructed by the NJCS using non-public data that accounts for both sample design and non-response.<sup>21</sup> Thus, the key assumption is that the probability that the information is missing for a given individual is random conditional on the set of variables used to construct the weight. Frumento et al. (2012) employ the same assumption but use variables available in the public version of the NJCS data. To construct the data used in this exercise, all the individuals who responded to the 48-month interview are included and those with missing values for weekly working hours, weekly earnings, or enrollment information are dropped.<sup>22</sup>

The estimation results are presented in Table 4.10. The estimated lower and upper bounds for the effect of JC on ln(wages) for the *cEE* stratum are below those estimated in Table 4.9, which ignore the non-response issue. Focusing on Non-Hispanics, the positive region covered by the bounds under both sets of assumptions (monotonicity and mean dominance) is slightly less than the corresponding region covered by the bounds in Table 4.9. The estimated upper bound in Table 4.10 equals .153, which is close to the one presented in Table 4.9 (.161). The estimate of the lower bound in Table 4.10 is also lower than, but still relatively close to, that presented in Table 4.9. In general, although adjusting for non-response slightly weakens the previous findings, the results

and Mealli (2009), who estimate the effect of JC on wages for the always-employed individuals without adjusting for non-compliance.

<sup>&</sup>lt;sup>21</sup>More specifically, the weights address sample design, 48-month interview design, and 48-month interview non-response.

<sup>&</sup>lt;sup>22</sup>As discussed in Section 4.3.1, the sample used in the previous tables includes only individuals with non-missing values for weekly earnings and weekly hours worked for every week after random assignment. This is done to make the results comparable to those in Lee (2009).

still strongly suggest a positive average effect of JC enrollment on wages four years after random assignment for the always-employed compliers.

As previously mentioned, Frumento et al. (2012) point identify the average effect of JC enrollment on wages, adjusting for sample selection, non-compliance, and missing outcomes, by imposing a different set of assumptions from the two sets discussed here. Employing a different sample from the one here, they estimate this effect to be about 3.8 percent for the always-employed compliers in the population. This point estimate is consistent with the estimated bounds for this effect presented in Table 4.10 for the entire population under Assumptions 3.1 through 3.5 and, although 3.8 percent is below the estimated lower bound under Assumptions 3.1 through 3.6 (4.4 percent), it falls inside the 95 percent confidence interval constructed under the six assumptions. Thus, the point estimate of the effect of JC on wages for the *cEE* stratum in Frumento et al. (2012) is consistent with the bounds adjusting for non-response.

## 4.3.4 Simulation Exercises

Simulation exercises aim to examine how the bounds in Propositions 3.1 and 3.2 behave when Assumptions 3.5 and 3.6 fail. By mimicking the characteristics of the sample involving all the individuals with continuously non-missing values of key variables, the simulation results suggest that the bounds are relatively robust even if the two assumptions fail.

In each simulation, 9090 observations are generated, with each randomly assigned to the treatment with the probability of 5491/9090. Membership of principal strata is drawn from a uniform distribution. By mimicking  $p_{ds|z}$  in the sample, the true values

	Entire Sample	Non-Hispanics
Number of observations	10520	8701
$\pi_{aEE}$	.017** (.002)	.018** (.002)
$\pi_{nEE}$	.162** (.005)	.162** (.005)
$\pi_{cEE}$	.391** (.009)	.381** (.010)
$\pi_{cNE}$	.038** (.010)	.049** (.011)
$\pi_{aNN}$	.026** (.003)	.028** (.003)
$\pi_{nNN}$	.108** (.004)	.108** (.004)
$\pi_{cNN}$	.258** (.006)	.254** (.007)
$lpha_{cEE}$	.877** (.022)	.851** (.024)
E[Y(1) aEE]	2 010** ( 050)	2 001** ( 052)
E[Y(0) nEE]	2.033** (.015)	2.032** (.016)
E[Y(0) cEE]	1.985** (.013)	1.961** (.015)
$\overline{Y}(v_{\alpha}^{111} < Y < v_{1}^{111})$	2.449** (.065)	2.383** (.056)
$\overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111})$	1.666** (.036)	1.702** (.034)
Bounds with Monotonicity (Propos	ition 1)	
$[LY_{1 \ CFF}, UY_{1 \ CFF}]$	[1.957, 2.102]	[1.940, 2.114]
CLR 95% level confidence interval	(1.928, 2.127)	(1.911, 2.140)
$[L_{cEE}, U_{cEE}]$	[028, .117]	[021, .153]
CLR 95% level confidence interval	(065, .153)	(060, .191)
Bounds with Monotonicity and Me	an Dominance (Prope	osition 2)
$[LY_{1,cEE}, UY_{1,cEE}]$	[2.029, 2.102]	[2.028, 2.114]
CLR 95% level confidence interval	(2.013, 2.128)	(2.011, 2.142)
$[L_{cEE}, U_{cEE}]$	[.044, .117]	[.067, .153]
CLR 95% level confidence interval	(.015, .155)	[.035, .193]

Table 4.10: Bounds for the Effects of JC on ln(wage) for cEE Adjusting for Non-Response

Note: \*\* denotes estimate is statistically different from 0 at 5% level and numbers in parentheses are standard errors. Computations use weights accounting for sample design, interview design and interview non-response. The standard error is calculated by a 5,000-repetition bootstrap. Numbers in square brackets are half-median unbiased estimators for the bounds, while numbers below them are CLR 95% level confidence intervals.

of proportions of strata are set as follows:  $\pi_{aNN} = .030$ ,  $\pi_{aEE} = .015$ ,  $\pi_{nNN} = .105$ ,  $\pi_{nEE} = .155$ ,  $\pi_{cEE} = .395 - \pi_{cEN}$ ,  $\pi_{cNN} = .260 - \pi_{cEN}$ , and  $\pi_{cNE} = .040 + \pi_{cEN}$ . Note that without Assumption 3.5,  $\pi_{cEN} \neq 0$ . The observed treatment and employment statuses, D and S, are jointly determined by the membership of principal strata and the random assignment indicator. Observed wages of different strata follow lognormal distributions with means: E[Y(1)|cEE] = 2.04, E[Y(0)|cEE] = 1.97, E[Y(1)|aEE] = 2.035, E[Y(0)|nEE] = 2.035 and E[Y(0)|cEN] = 2. Thus, the true effect  $\Delta$  is equal to .07. The variances of these average outcomes are equal to .2. These values are chosen such that they are close to the observed  $\overline{Y}^{zds}$  in the sample.

Simulation 1 examines the bounds when only Assumption 3.5 fails. Assumption 3.6 holds as E[Y(1)|cEE] - E[Y(1)|cNE] = .16. Figure 4.1 shows the bounds when  $\pi_{cEN}$  changes over [0,.26]. Note that the range of  $\pi_{cEN}$  is determined by the proportions of other strata. In order to give a sense about the variability of the estimates of the bounds, this subsection also shows the 5th percentile of the estimates of the lower bound from 1000 repeated simulations, and the 95th percentile of the estimates of the upper bound.<sup>23</sup> The bounds slope downward as  $\pi_{cEN}$  increases. This is because in the trimming cell {Z = 1, D = 1, S = 1},  $\pi_{cEE}$  decreases and  $\pi_{cNE}$  increases along with the increasing  $\pi_{cEN}$ , and since E[Y(1)|cEE] is larger than E[Y(1)|cNE], the proportion of the observations with larger values of  $\log(wage)$  increases while the proportion with smaller values decreases. As a result, with the trimming proportions and range of  $\log(wage)$  in the trimming cell fixed, the averages obtained from both the top and the bottom distribution become smaller. The true effect is above the lower bound in

<sup>&</sup>lt;sup>23</sup>For example, 95 percent of the estimates of the upper bound fall below p95\_UB, and 95 percent of the estimates of LB1 lie above p5 LB1.



Figure 4.1: Bounds When Monotonicity Fails in Simulation 1

Proposition 3.2 over the entire range of  $\pi_{cEN}$ , and it falls outside the upper bound when  $\pi_{cEN} > .16$ . Note that the lower bound in Proposition 3.2 identifies a positive treatment effect as long as  $\pi_{cEN} \leq .12$ . Figure 4.2 shows percentage of times out of 1000 repetition the true effect falls inside the bounds. The percentage of times for Proposition 3.2 starts at a lower value than that for Proposition 3.1 because the true effect is very close to the lower bound in Proposition 3.2 at the origin. The percentage of times for both propositions coincide with each other when  $\pi_{cEN} > .06$ . Note that both assumptions hold at the origin. The simulated bounds are [-.027, .136] in Proposition 3.1 and [.054, .136] in Proposition 3.2, respectively, similar to the empirical results for the entire sample in Table 4.9.

Simulation 2 analyzes how the mean dominance assumption affects the bounds when  $\pi_{cEN} = 0$ . The horizontal line shows the difference between E[Y(1)|cEE] and E[Y(1)cNE], and the assumption holds in the right region of the axis origin. The down-



Figure 4.2: Percentage of Times True Effect Falls within the Bounds in Simulation 1

ward slope of the bounds in Figure 4.3 follows a different story in Simulation 2: larger values of the outcome of cNE members are gradually replaced with smaller values; with the constant proportions of cEE and cNE in the trimming cell, the bounds decrease along the x-axis. The true effect  $\Delta$  falls within the bounds in Proposition 3.1 over the entire range [-.5, .5] of the difference. Instead,  $\Delta$  crosses the lower bound in Proposition 3.2 at the origin. This is because when  $\pi_{cNE} = 0$  the direction of the mean dominance assumption determines whether the quantity in Proposition 3.2 is a lower bound or an upper bound for the parameter of interest  $\Delta$ . The lower bound in Proposition 3.2 identifies a positive treatment effect over the entire range of the difference. Note that the lower bound in Proposition 3.1 also identifies a positive effect when the direction of the mean dominance assumption is reversed at  $E[Y(1)|cNE] - E[Y(1)cEE] \ge .15$ . As shown in Figure 4.4,  $\Delta$  falls within the bounds in Proposition 3.2 increases as the mean



Figure 4.3: Bounds When Mean Dominance Fails in Simulation 2

Figure 4.4: Percentage of Times True Effect Falls within the Bounds in Simulation 2



dominance assumption holds to a greater extent. Note that at the origin the percentage of times is around 50%.

Simulations 3 to 5 illustrate how the bounds behave when both assumptions fail, with  $\pi_{cEN}$  set as .05, .10 and .15, respectively. Comparing among Figures 4.3, 4.5,

4.7, and 4.9,  $\Delta$  falls within the bounds in Proposition 3.1 over the shrinking range of E[Y(1)|cEE] - E[Y(1)|cNE] as  $\pi_{cEN}$  increases. In contrast to Figure 4.3 where  $\pi_{cEN} =$ 0, the intersection of  $\Delta$  with the lower bound in Proposition 3.2 moves slightly leftward in Figures 4.5, 4.7 and 4.9. Mathematically, when the difference between E[Y(0)|cEN]and E[Y(0)|cEE] stays positive, the intersection moves leftward away from the origin as  $\pi_{cEN}$  increases. The tilt of the bounds over the entire range of the difference increases because the effect of *cEN* members through their wages on the identified quantity of E[Y(0)|cEE] is strengthened as  $\pi_{cEN}$  increases at each value of the difference. The increasing tilt plays different roles in how these bounds identify the sign of  $\Delta$ . When  $\pi_{cEN} = .05$  and  $\pi_{cEN} = .10$ , the upper bound stays positive. Accordingly, the extent to which the lower bound in Proposition 3.2 identifies a positive treatment effect is attenuated by the increasing proportion of  $\pi_{cEN}$ , as shown by the narrower range of the difference over which the lower bound stays beyond the zero line. Instead, the range over which the lower bound in Proposition 3.1 identifies a positive effect enlarges slightly, in spite of the negative region of the axis. When  $\pi_{cEN} = .15$ , the lower bounds in Propositions 3.1 and 3.2 identify a positive effect when the difference is not larger than -.05 and .15, respectively. Moreover, the upper bound identifies a negative effect when the difference is not smaller than .35. In contrast with Figure 4.4, Figures 4.6, 4.8, and 4.10 display a bell shape of percentage of times the propositions are verified. The bell covers a shrinking area as  $\pi_{cEN}$  increases. The percentages of times for both propositions coincide with each other when  $\Delta$  falls outside the bounds in Proposition 3.1. Otherwise, the one for Proposition 3.1 is above that for Proposition 3.2, with its center at the origin. Instead, the center for Proposition 3.2 moves slightly leftward as



Figure 4.5: Bounds When Both Assumptions Fail with  $\pi_{cEN} = .05$  in Simulation 3

Figure 4.6: Percentage of Times True Effect Falls within the Bounds in Simulation 3



 $\pi_{cEN}$  increases. Note that the percentage of times for Proposition 3.2 increases at the origin as  $\pi_{cEN}$  increases.

The following conclusions can be drawn from the simulation exercises above. First, when Assumption 3.6 holds, the bounds in both propositions are relatively robust to the



Figure 4.7: Bounds When Both Assumptions Fail with  $\pi_{cEN} = .10$  in Simulation 4

Figure 4.8: Percentage of Times True Effect Falls within the Bounds in Simulation 4





Figure 4.9: Bounds When Both Assumptions Fail with  $\pi_{cEN} = .15$  in Simulation 5

Figure 4.10: Percentage of Times True Effect Falls within the Bounds in Simulation 5



existence of a small fraction of *cEN* members. As shown in Simulation 1, the allowable limit of  $\pi_{cEN}$  is .16. Second, when both assumptions fail, the bounds in Proposition 3.1 are still valid over a symmetric range of E[Y(1)|cEE] - E[Y(1)|cNE] around the origin, though the range shrinks gradually as  $\pi_{cEN}$  increases. The range for the bounds in Proposition 3.2 also shrinks, but the lower bound becomes less sensitive to the violation of Assumption 3.6 at the origin as  $\pi_{cEN}$  increases. The lower bound in Proposition 3.2 remains to identify a positive treatment effect for *cEE* as long as small departures occur to the assumptions. In summary, the simulation results show that the bounds in the presence of sample selection and noncompliance are relatively robust to the key identification Assumptions 3.5 and 3.6. This reinforces the robustness of the bounds obtained for the wage effects of JC enrollment in the previous subsection. For the always-employed compliers, the wage effect is bounded between 5.7 percent and 13.9 percent four years after random assignment. This evidence suggests that Job Corps has positive effects not only on the employability of its participants but also on their wages, implying that Job Corps is likely to have positive effects on their human capital.

#### CHAPTER 5

#### CONCLUSION

Research on partial identification approaches has recently received increasing attention in program evaluation literature. The main advantage of partial identification over traditional point identification is its dependence on milder assumptions, thus delivering more credible results in empirical applications. This dissertation extends Instrumental Variable (IV) methods in the program evaluation literature by partially identifying treatment effects of interest.

Imbens and Angrist (1994) and Angrist, Imbens and Rubin (1996) developed an influential approach within the IV framework. They show that, when allowing for heterogeneous effects, IV estimators point identify the local average treatment effect (LATE) for compliers. This dissertation advances the current IV literature in two important ways. First, inspired by a common criticism of their approach that the focus is on the effect for a subpopulation, the dissertation derives sharp nonparametric bounds for the population average treatment effect (ATE) within the LATE framework. It improves the bounds on the ATE proposed in the current IV literature by combining two sets of assumptions. The first is monotonicity in the treatment of the average outcomes of strata without specifying a priori direction. This assumption infers the sign of average treatment effects for other subpopulations from that for compliers. In contrast to the existing literature, monotonicity imposed on the average outcomes within the same subpopulation allows some individuals to experience a treatment effect that has the opposite sign to the LATE. The second set of assumptions is mean dominance that compares average potential outcomes across different subpopulations. This assumption infers the

unobserved terms from the identified average outcomes across subpopulations. Different from the current literature, some of the bounds do not require a bounded-outcome assumption once combining the monotonicity and mean dominance assumptions. Moreover, indirect evidence regarding mean dominance can be obtained by estimating pretreatment average characteristics of each stratum from an overidentified nonparametric GMM problem.

Second, this dissertation extends the LATE framework to bound treatment effects in the presence of sample selection and noncompliance. The sample selection issue arises when outcomes of interest are only observed for a selected group. The noncompliance problem appears because individuals can choose whether or not to actually take the treatment in most of randomized experiments in economics and other social science fields, especially the experiments with encouragement designs. As a result, it is common that some individuals in the treatment group do not take the treatment while some individuals in the control group do. The dissertation extends the partial identification results in Zhang, Rubin, and Mealli (2008) and Lee (2009), who construct bounds in the presence of sample selection, to also account for noncompliance. Within the framework of principal stratification, it derives nonparametric bounds on the average treatment effect for the always-selected compliers, who would comply with their assigned treatment and whose outcomes are always-selected regardless of treatment assignment. This is the only group of individuals whose outcomes are observable under both treatment *receipt* arms. Additional assumptions are necessary to derive bounds for other subpopulations. More generally, these bounds can be employed in settings where two identification problems are present (e.g., endogeneity and missing outcomes) and there is a valid instrument to address one of them. Thus, the dissertation provides an important extension to the current partial identification literature in program evaluation, as it is common in empirical applications to face more than one identification problem.

The dissertation employs the derived bounds to evaluate the effectiveness of the Job Corps (JC) program, which is the most comprehensive and largest federally-funded job training program for disadvantaged youth in the United States. It uses experimental data from the National Job Corps Study (NJCS). Though individuals were randomly assigned to a treatment group or a control group, noncompliance arises. In the NJCS, 26% of individuals who were assigned to participate in JC did not enroll, while 4% of individuals who should have been embargoed from JC participation did enroll. Accordingly, previous literature on the evaluation of the JC program usually analyzes intention-to-treat (*ITT*) effects or the *LATE* for compliers. Using random assignment as an IV to address noncompliance, the dissertation contributes to this empirical literature by providing credible and informative bounds for treatment effects other than *LATE* and *ITT* effects.

Focusing on the population average treatment effect on labor market outcomes and welfare dependence, the monotonicity assumption and the mean dominance assumption together provide the narrowest bounds on the *ATEs* of JC enrollment: [24.61, 201.04] for weekly earnings and [.042, .163] for employment four years after randomization, and [-142.76, -84.29] for the yearly dependence on public welfare benefits. These bounds are significantly narrower than the bounds proposed in the current IV literature. The lower bounds for weekly earnings and employment are 10 percent higher than their respective *ITT* effects (22.19 and .038), while the upper bound for public benefits is equal

to its *ITT* effect. The *LATE*s for compliers on the three outcomes also fall within these narrowest bounds. Thus, it is safe to conclude that JC enrollment increases weekly earnings by at least \$24.61 and employment by at least 4.3 percentage points, and decreases the fourth yearly dependence on public welfare benefits by at least \$84.29.

The dissertation also evaluates the wage effect of JC, where both the sample selection issue (wages are only observed for employed individuals) and noncompliance are present. The two key assumptions are a monotonicity assumption on the effect of JC on employment, and a mean dominance assumption stating that the average potential wage under treatment of the always-employed compliers is greater than that of compliers who would be employed only if they participated in JC. As discussed in the dissertation, both assumptions are plausible, especially for Non-Hispanics. The simulation exercises also reinforce the identification power of the bounds under these two assumptions. Focusing on the always-employed compliers, who would comply with their assigned treatment and who would be always employed regardless of assignment status, the wage effect of JC enrollment is between 5.7 percent and 13.9 percent four years after random assignment, and between 7.7 and 17.5 percent for Non-Hispanics. This evidence suggests that Job Corps has positive effects not only on the employability of its participants but also on their wages, implying that Job Corps is likely to have positive effects in their human capital. Therefore, it is very important to consider the potential benefits of Job Corps and other training programs on wages when evaluating their effect.

The assumptions and methodologies employed in this dissertation can be used in partial identification of treatment effects in other settings. For example, one important extension is to bound the population ATE when the instrument does not satisfy the ex-

clusion restriction. Argument over an IV usually concentrates on the plausibility of the exclusion restriction. Thus, how to make inference about the ATE when this assumption is violated is of great importance. Another interesting extension is to identify direct and indirect effects (e.g., Rubin, 2004; Sjölander, 2009; VanderWeele, 2011) in the presence of one identification issue (e.g., sample selection, noncompliance). When the causal channel between the treatment and the outcome is intervened by an intermediate variable, the direct effect refers to the causal effect of the treatment on the outcome net of the part that works through the intermediate variable while the indirect effect refers the part that works through the intermediate variable. Current literature on direct and indirect effects usually involves intensive computation and thus deliver numerical solutions. The methodology in this dissertation can be used to derive analytical bounds in the presence of one identification issue, for example, endogeneity. While beyond the scope of this dissertation, these extensions are at the top of the research agenda. Further topics to develop the present methodology include narrowing bounds by exploiting variations among covariates and deriving bounds for treatment effects in alternative designs, for example, an IV with three values, which is common in the fields of public health and epidemiology.

#### **BIBLIOGRAPHY**

- [1] Angrist, J. (1990), "Lifetime Earnings and the Vietnam Era Draft Lottery: Evidence from Social Security Administrative Records," American Economic Review 80, 313-335.
- [2] Angrist, J., Chen, S. and Frandsen, B. (2009), "Did Vietnam Veterans Get Sicker in the 1990s? The Complicated Effects of Military Service on Self-Reported Health," NBER Working Paper No. 14781.
- [3] Angrist, J. and Fernandez-Val, I. (2010), "Extrapolate-ing: External Validity and Overidentification in the *LATE* Framework," Working Paper 16566, NBER.
- [4] Angrist, J., Imbens, G. and Rubin, D. (1996), "Identification of Causal Effects Using Instrumental Variables," Journal of the American Statistical Association, 91, 444-472.
- [5] Balke, A. and Pearl, J. (1997), "Bounds on Treatment Effects from Studies with Imperfect Compliance," Journal of the American Statistical Association 92(439), 1171-1176.
- [6] Bhattacharya, J., Shaikh, A. and Vytlacil, E. (2008), "Treatment Effect Bounds under Monotonicity Assumptions: An Application to Swan-Ganz Catheterization," American Economic Review: Papers & Proceedings, 98 (2), 351-356.
- [7] Blanco, G., Flores, C. and Flores-Lagunes, A. (2012), "Bounds on Average and Quantile Treatment Effects of Job Corps Training on Wages," forthcoming, Journal of Human Resources.
- [8] Burghardt, J., Schochet, P.Z., McConnell, S., Johnson, T., Gritz, R.M., Glazerman, S., Homrighausen, J. and Jackson, R. (2001), Does Job Corps Work? Summary of the National Job Corps Study, 8140-530, Mathematica Policy Research, Inc., Princeton, NJ.
- [9] Chernozhukov, V., Lee, S. and Rosen, A. (2011), "Intersection Bounds: Estimation and Inference," forthcoming, Econometrica.
- [10] Chesher, A. (2010), "Instrumental Variable Models for Discrete Outcomes," Econometrica 78 (2), 575-601.
- [11] Chiburis, R. (2010a), "Bounds on Treatment Effects Using Many Types of Monotonicity," Working paper.

- [12] Chiburis, R. (2010b), "Semiparametric Bounds on Treatment Effects," Journal of Econometrics 159, 267-275.
- [13] Deaton, A. (2009), "Instruments of Develpment: Randomization in the Tropics, and the Search for the Eclusive Keys to Economic Dvelpment," NBER Working paper, No. 14690.
- [14] Deaton, A. (2010), "Instruments, Randomization, and Learning about Development," Journal of Economic Literature 48, 424-455.
- [15] Flores, C. and Flores-Lagunes, A.(2010a), "Partial Identification of Local Average Treatment Effects with an Invalid Instrument," Working Paper.
- [16] Flores, C. and Flores-Lagunes, A. (2010b), "Nonparametric Partial Identification of Causal Net and Mechanism Average Treatment Effects," Working Paper.
- [17] Flores, C., Flores-Lagunes, A., Gonzalez, A. and Neumann, T. (2012), "Estimating the Effects of Length of Exposure to Instruction in a Training Program: The Case of Job Corps," The Review of Economics and Statistics, 94 (1), 153-171.
- [18] Flores-Lagunes, A., Gonzalez, A. and Neumann, T. (2010), "Learning but not Earning? The Impact of Job Corps Training on Hispanic Youth," Economic Inquiry, 48 (3), 651-667.
- [19] Frangakis, C.E. and Rubin D. (2002), "Principal Stratification in Causal Inference," Biometrics, 58, 21-29.
- [20] Frumento, F., Mealli, F., Pacini, B. and Rubin, D. (2012), "Evaluating the Effect of Training on Wages in the Presence of Noncompliance and Missing Outcome Data," forthcoming, Journal of the American Statistical Association.
- [21] Hahn, J. (2010), "Bounds on ATE with Discrete Outcomes," Economics Letters 109, 24-27.
- [22] Heckman, J. (1979), "Sample Selection Bias as a Specification Error," Econometrica, 47, 153-162.
- [23] Heckman, J. (1996), "On Air: Identification of Causal Effects Using Instrumental Variables," Journal of The American Statistical Association, 91, 459-462.

- [24] Heckman, J. (2010), "Building Bridges between Structural and Program Evaluation Approaches to Evaluating Policy," Journal of Economic Literature, American Economic Association 48 (2), 356-398.
- [25] Heckman, J. and Vytlacil, E. (2000), "Instrumental Variables, Selection Models, and Tight Bounds on the Average Treatment Effect," Technical Working Paper 259, NBER.
- [26] Heckman, J. and Urzua, S. (2010), "Comparing IV with Structural Models: What Simple IV Can and Cannot Identify," Journal of Econometrics, 156 (1), 27–37.
- [27] Hirano, K. and Porter, J. (2012), "Impossibility Results for Nondifferentiable Functionals," Econometrica, 80 (4), 1769-1790.
- [28] Horowitz, J.L. and Manski, C.F. (1995), "Identification and Robustness with Contaminated and Corrupted Data," Econometrica, 63 (2), 281-302.
- [29] Huber, M. and Mellace G. (2010), "Sharp Bounds on Average Treatment Effects under Sample Selection," Working Paper.
- [30] Imai, K. (2008), "Sharp Bounds on the Causal Effects in Randomized Experiments with 'Truncation-by-Death'," Statistics & Probability Letters, 78, 141-149.
- [31] Imbens, G. and Angrist, J. (1994), "Identification and Estimation of Local Average Treatment Effects," Econometrica, 62 (2), 467-475.
- [32] Imbens, G. and Manski, C.F. (2004), "Confidence Intervals for Partially Identified Parameters," Econometrica, 72 (6), 1845-1857.
- [33] Imbens, G. (2009), "Better LATE than Nothing: Some Comments on Deaton (2009) and Heckman and Urzua (2009)," NBER Working paper, No. 14896.
- [34] Imbens, G. and Wooldridge, J. (2009), "Recent Developments in the Econometrics of Program Evaluation," Journal of Economic Literature, 47, 5-86.
- [35] Kitagawa, T. (2009), "Identification Region of the Potential Outcome Distributions under Instrument Independence," CEMMAP working paper.
- [36] Lechner, M. and B. Melly (2010), "Partial Identification of Wage Effects of Training Programs," Working Paper.
- [37] Lee, D. (2009), "Training, Wages, and Sample Selection: Estimating Sharp Bounds on Treatment Effects," Review of Economic Studies, 76, 1071-102.
- [38] Machado, C., Shaikh, A. and Vytlacil, E. (2009), "Instrumental Variables and the Sign of the Average Treatment Effect," Working paper.
- [39] Manski, C. (1990) "Nonparametric Bounds on Treatment Effects," American Economic Review: Papers and Proceedings 80, 319-323.
- [40] Manski, C. and Pepper, J. (2000), "Monotone Instrumental Variables: with an Application to the Returns to Schooling," Econometrica 68 (4), 997-1010.
- [41] Newey, W. and McFadden, D. (1994), "Large Sample Estimation and Hypothesis Testing," Handbook of Econometrics, vol. IV, Chapter 36, 2111-2245.
- [42] Oreopoulos, P. (2006), "Estimate Average and Local Average Treatment Effects of Education When Compulsory Schooling Laws Really Matter," American Economic Review 96, 152-175.
- [43] Robins, J. and Greenland, S. (1996), "Comment on Angrist, Imbens and Rubin: Estimation of the Global Average Treatment Effects Using Instrumental Variables," Journal of the American Statistical Association, 91, 456-458.
- [44] Rosenzweig, M. and Wolpin, K. (1980), "Testing the Quantity-Quality Fertility Model: the Use of Twins as a Natural Experiment," Econometrica, 48, 227-240.
- [45] Rubin, D. (1976), "Inference and Missing Data," Biometrika, 63 (3), 581-592.
- [46] Rubin, D. (1978), "Bayesian Inference for Causal Effects," The Annals of Statistics, 6, 34-58.
- [47] Rubin, D. (1980), Comment on "Randomization Analysis of Experimental Data: The Fisher Randomization Test," by D. Basu, Journal of American Statistical Association, 75, 591-593.
- [48] Rubin, D. (1990), Comment: "Neyman (1923) and Causal Inference in Experiments and Observational Studies," Statistical Science, 5, 472-480.
- [49] Rubin, D. (2004), "Direct and Indirect Causal Effects via Potential Outcomes (With discussion)," Scandinavian Journal of Statistics 31, 161-198.

- [50] Rubin, D. (2005), "Causal Inference Using Potential Outcomes: Design, Modeling, Decisions," Journal of the American Statistical Association, 100, 322-331.
- [51] Schochet, P. (2001), National Job Corps Study: Methodological Appedixes on the Impact Analysis, 8140-530, Mathematica Policy Research, Inc., Princeton, NJ.
- [52] Schochet, P., Burghardt, J. and Glazerman, S. (2001), National Job Corps Study: The Impacts of Job Corps on Participants' Employment and Related Outcomes, 8140-530, Mathematica Policy Research, Inc., Princeton, NJ.
- [53] Schochet, P., Burghardt, J. and McConnell, S. (2008), "Does Job Corps Work? Impact Findings from the National Job Corps Study," American Economic Review, 98 (5), 1864-1886.
- [54] Shaikh, A. and Vytlacil, E. (2005), "Threshold Crossing Models and Bounds on Treatment Effects: A Nonparametric Analysis," Technical Working Paper 307, National Bureau of Economic Research.
- [55] Shaikh, A. and Vytlacil, E. (2011), "Partial Identification in Triangular Systems of Equations with Binary Dependent Variables," Econometrica, 79 (3), 949-955.
- [56] Sjölander, A. (2009), "Bounds on Natural Direct Effects in the Presence of Confounded Intermediate Variables," Statistics in Medicine 28, 558-571.
- [57] Tamer, E., 2010, "Partial Identification in Econometrics," Annual Review of Economics 2, 167-195.
- [58] U.S. Department of Labor (2005), "Job Corps Fact Sheet," http://www.doleta.gov/Programs /factsht /jobcorps.cfm (December 24, 2006).
- [59] van Ours, J. (2004), "The Locking-in Effect of Subsidized Jobs," Journal of Comparative Economics, 32, 37-55.
- [60] VanderWeele, T.J. (2011), "Causal Mediation Analysis with Survival Data," Epidemiology 22, 582-585.
- [61] Vytlacil, E. (2002), "Independence, Monotonicity, and Latent Index Models: an Equivalence Result," Econometrica 70 (1), 331-341.
- [62] Zhang, J.L. and Rubin, D. (2003), "Estimation of Causal Effects via Principal Stratification When Some Outcomes are Truncated by 'Death'," Journal of Educational and Behavioral Statistics, 28 (4), 353-368.

- [63] Zhang, J.L., Rubin, D. and Mealli, F. (2008), "Evaluation the Effects of Job Training Programs on Wages Through Principal Stratification," in D. Millimet et al. (eds) Advances in Econometrics, XXI, Elsevier.
- [64] Zhang, J.L., Rubin, D. and Mealli, F. (2009), "Likelihood-Based Analysis of Causal Effects of Job-Training Programs Using Principal Stratification," Journal of the American Statistical Association, 104 (485), 166-176.

#### APPENDIX A

#### **DERIVATION OF THE BOUNDS**

## A.1 Derivation of the Bounds in Section 3.2

This section only presents the proof of Proposition 2.2, as the proofs for the rest of the propositions are similar.

**Proof.** Under Assumptions 1.1 through 1.4, AIR (1996) show that  $LATE_c = (E[Y|Z = 1] - E[Y|Z = 0])/(p_{1|1} - p_{1|0})$ . By Assumption 2.2(ii), and since Z is ordered such that  $p_{1|1} > p_{1|0}$ , the direction of the monotonicity in Assumption 2.2(i) is identified from the sign of  $LATE_c$ . Here I consider only the case when  $LATE_c > 0$ , as the sharp bounds when  $LATE_c < 0$  are constructed in the same way. By equation (2.1),  $ATE = \pi_{at}(E[Y(1)|at] - E[Y(0)|at]) + \pi_{nt}(E[Y(1)|nt] - E[Y(0)|nt]) + \pi_c LATE_c$ . Under Assumptions 1.1 though 1.4, the sampling process identifies each of the quantities to the right of this equation except for E[Y(1)|nt] and E[Y(0)|at], and thus equation (2.2) follows. Since there are no restrictions on these two means other than those imposed by Assumptions 2.1 and 2.2(i), these two assumptions directly imply the bounds  $y^{\mu} \ge E[Y(1)|nt] \ge E[Y(0)|nt] = \overline{Y}^{10}$  and  $\overline{Y}^{01} = E[Y(1)|at] \ge E[Y(0)|at] \ge y^{l}$ . The lower (upper) bound on ATE in Proposition 2.2 is obtained from equation (2) by setting E[Y(1)|nt] at its lower (upper) bound and E[Y(0)|at] at its upper (lower) bound.

For sharpness, first of all, ATE attains its smallest value when  $E[Y(0)|at] = \overline{Y}^{01}$ and  $E[Y(1)|nt] = \overline{Y}^{10}$ . Otherwise, always-takers or never-takers violate Assumption 2.2(i). Similarly, ATE attains its largest value when  $E[Y(0)|at] = y^{l}$  and  $E[Y(1)|nt] = y^{u}$ . Otherwise, always-takers or never-takers violate Assumption 2.1. Next, I will show that  $\forall \alpha \in [LB, UB]$ , there exist distributions consistent with observed data, and  $ATE = \alpha$  evaluated under such distributions.  $\forall \alpha \in [LB, UB]$ , it can be written as  $\alpha = \overline{Y}^{11}p_{1|1} - \overline{Y}^{00}p_{0|0} + q_1p_{0|1} - q_0p_{1|0}$ , where  $q_1 \in [\overline{Y}^{10}, y^u]$  and  $q_0 \in [y^l, \overline{Y}^{01}]$ . Let  $F_{Y_1|Z,D}(y_1|1,d)$  denote the distribution of the potential outcome Y(1) conditional on Z = 1 and D = d. Similarly,  $F_{Y_0|Z,D}(y_0|0,d)$  denotes the distribution of the potential outcome Y(0) conditional on Z = 0 and D = d. Then, define

$$F_{Y_1|Z,D}(y_1|1,d) = \begin{cases} F_{Y|Z,D}(y|1,1), & \text{if } D = 1\\ 1[y_1 \ge q_1], & \text{if } D = 0 \end{cases}$$

and

$$F_{Y_0|Z,D}(y_0|0,d) = \begin{cases} F_{Y|Z,D}(y|0,0), & \text{if } D = 0\\ 1[y_0 \ge q_0], & \text{if } D = 1 \end{cases}$$

$$\begin{split} ATE &= E[Y(1) - Y(0)] \\ &= E[Y(1)|Z = 1] - E[Y(0)|Z = 0] \\ &= p_{1|1}E[Y(1)|Z = 1, D = 1] + p_{0|1}E[Y(1)|Z = 1, D = 0] - p_{1|0}E[Y(0)|Z = 0, D = 0] \\ &= p_{1|1}E[Y|Z = 0, D = 0] \\ &= p_{1|1}E[Y|Z = 1, D = 1] + p_{0|1}E[Y(1)|Z = 1, D = 0] - p_{1|0}E[Y(0)|Z = 0, D = 1] - p_{0|0}E[Y|Z = 0, D = 0] \\ &= p_{1|1}\overline{Y}^{11} + p_{0|1}q_1 - p_{1|0}q_0 - p_{0|0}\overline{Y}^{00} \\ &= \alpha. \end{split}$$

The second line follows Assumption 1.1, the third line follows Law of Iterated Expectation, and the fourth and fifth lines follow the defined distributions.

#### A.2 Derivation of the Bounds in Section 3.3

## A.2.1 **Proof of Proposition 3.1**

First, I show that under Assumptions 3.1 through 3.5  $L_{cEE}$  and  $U_{cEE}$  are the smallest and largest possible values, respectively, for the average treatment effect for the stratum cEE. Next, I prove that for  $\forall \Delta \in [L_{cEE}, U_{cEE}]$ , there exist distributions for cEE, aEE, and cNE consistent with the observed data of Y in  $\{Z = 1, D = 1, S = 1\}$  and the constraint that  $E[Y(1)|aEE] = \overline{Y}^{011}$ . In other words, the interval  $[L_{cEE}, U_{cEE}]$  contains any other bounds that are consistent with Assumptions 3.1 through 3.5. The first-step proof is similar to that in Horowitz and Manski (1995), except that a binding constraint should be satisfied under the lower and upper bounds. Since both  $L_{cEE}$  and  $U_{cEE}$  depend on the range of  $\overline{Y}^{011}$ , I need to discuss multiple cases in the proof.

**Proof.** First by Assumptions 3.1 through 3.5, the proportions of each stratum are uniquely determined by the observed data. Thus, the proof is completed given the proportions of the strata. Second since E[Y(0)|cEE] is point identified by Assumptions 3.1 through 3.5, the proof can be completed with respect to E[Y(1)|cEE] instead of the average treatment effect for *cEE*. Let  $\theta = E[Y(1)|cEE]$ , and then  $\Delta = \theta - E[Y(0)|cEE]$ . Third, since both  $LY_{1,cEE}$  and  $UY_{1,cEE}$  depend on the range of  $\overline{Y}^{011}$ , I have to discuss multiple cases.

Let Qy be the observed distribution of Y in the cell [Z = 1, D = 1, S = 1].  $\overline{y}_{aEE} = \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}), \widetilde{y}_{aEE} = \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111})$ . Denote the expression of  $LY_{1,cEE}$  as  $\widetilde{LY}_{1,cEE}$  when  $\overline{y}_{aEE} > \overline{Y}^{011}$  and the expression of  $UY_{1,cEE}$  as  $\widetilde{UY}_{1,cEE}$  when

 $\tilde{y}_{aEE} < \overline{Y}^{011}$ , The probability density functions for each stratum are  $fy_{cEE}$ ,  $fy_{aEE}$ , and  $fy_{cNE}$ , and their corresponding distributions are  $Fy_{cEE}$ ,  $Fy_{aEE}$ , and  $Fy_{cNE}$ .

For the first-step proof, I discuss the two cases to show that  $LY_{1,cEE}$  is the smallest possible value for E[Y(1)|cEE]. The other two cases for the  $UY_{1,cEE}$  can be shown in the same way.

1. 
$$LY_{1,cEE} = \overline{Y}(Y \le y_{\alpha_{cEE}}^{111})$$
, when  $\overline{Y}^{011} \ge \overline{y}_{aEE}$   
Let  $Gy = \begin{cases} \frac{Qy}{\alpha_{cEE}}, & \text{if } y \le y_{\alpha_{cEE}}^{111} \\ 1, & \text{if } y > y_{\alpha_{cEE}}^{111} \end{cases}$ .

To show that  $\overline{Y}(Y \leq y_{\alpha_{cEE}}^{111})$  is the smallest value, I have  $Gy \geq Fy_{cEE}$ , for all  $Fy_{cEE}$   $\in \{\alpha_{cEE}Fy_{cEE} + \alpha_{aEE}Fy_{aEE} + \alpha_{cNE}Fy_{cNE} = Qy, E[Y(1)|aEE] = \overline{Y}^{011}\}$  and all  $y \in \mathscr{R}$ . If  $y \leq y_{\alpha_{cEE}}^{111}$ ,  $Gy < Fy_{cEE} \Rightarrow Qy < \alpha_{cEE}Fy_{cEE} \Rightarrow Qy < \alpha_{cEE}Fy_{cEE} + \alpha_{aEE}Fy_{aEE} + \alpha_{aEE}Fy_{aEE} + \alpha_{aEE}Fy_{aEE}$ 

 $\alpha_{cNE}Fy_{cNE}$ . This contradicts the feasible set of  $Fy_{cEE}$ .

If 
$$y > y_{\alpha_{cEE}}^{111}$$
,  $Gy - Fy_{cEE} = 1 - Fy_{cEE} \ge 0$ .

Next, I show that the distributions exist, when  $E[Y(1)|cEE] = \overline{Y}(Y \le y_{\alpha_{cEE}}^{111})$  and  $E[Y(1)|aEE] = \overline{Y}^{011}$ .

When  $\overline{Y}^{011} \ge \overline{y}_{aEE}$  and  $\overline{Y}^{011} \ge \overline{Y}(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ , let ty be the observed density of  $Y(Y \ge y_{1-\alpha_{aEE}}^{111})$  and hy the observed density of  $Y(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ . Then  $\exists \tau \in [0,1], s.t. \ \tau \overline{Y}(Y \ge y_{1-\alpha_{aEE}}^{111}) + (1-\tau)\overline{Y}(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111}) = \overline{Y}^{011}$ . Therefore,  $gy = fy_{cEE}, \ \tau ty + (1-\tau)hy = fy_{aEE}$  and  $\frac{(1-\tau)\pi_{aEE}}{\pi_{cNE}}ty + (1-\frac{(1-\tau)\pi_{aEE}}{\pi_{cNE}})hy = fy_{cNE}$ .

When  $\overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}) \leq \overline{Y}^{011} \leq \overline{y}_{aEE}$ , let ty be the observed density of  $Y(Y \geq y_{1-\alpha_{cNE}}^{111})$  and hy the observed density of  $Y(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111})$ . Then  $\exists \tau \in [0,1]$ , s.t.  $\tau \overline{Y}(Y \geq y_{1-\alpha_{cNE}}^{111}) + (1-\tau)\overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}) = \overline{Y}^{011}$ . Similarly,  $gy = fy_{cEE}$ ,  $\tau ty + (1-\tau)hy = fy_{aEE}$  and  $(1 - \frac{\tau \pi_{aEE}}{\pi_{cNE}})ty + \frac{\tau \pi_{aEE}}{\pi_{cNE}}hy = fy_{cNE}$ .

2.  $LY_{1,cEE} = \widetilde{LY}_{1,cEE}$ , when  $\overline{Y}^{011} \leq \overline{y}_{aEE}$ .

In this case, I first prove that  $\overline{Y}(Y \leq y_{p1-1,cNE}^{111})$  is the smallest feasible value for the quantity  $\frac{\pi_{cEE}}{\pi_{cEE} + \pi_{aEE}} E[Y(1)|cEE] + \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} E[Y(1)|aEE]$ , and then show  $\widetilde{LY}_{1,cEE}$  is the smallest feasible value for E[Y(1)|cEE].

Let 
$$Gy = \begin{cases} \frac{Qy}{1 - \alpha_{cNE}}, & \text{if } y \le y_{1 - \alpha_{cNE}}^{111} \\ 1, & \text{if } y > y_{1 - \alpha_{cNE}}^{111} \end{cases}$$

To show that  $\overline{Y}(Y \leq y_{p1-1,cNE}^{111})$  is the smallest value for  $\frac{\pi_{cEE}}{\pi_{cEE} + \pi_{aEE}} E[Y(1)|cEE] + \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} E[Y(1)|aEE]$ , I have  $Gy \geq \frac{\pi_{cEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{cEE} + \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{aEE}$ , for all  $Fy_{cEE} \in \{\alpha_{cEE}Fy_{cEE} + \alpha_{aEE}Fy_{aEE} + \alpha_{cNE}Fy_{cNE} = Qy, E[Y(1)|aEE] = \overline{Y}^{011}\}$  and all  $y \in \mathscr{R}$ .

If 
$$y \leq y_{1-\alpha_{cNE}}^{111}$$
,  $Gy < \frac{\pi_{cEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{cEE} + \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{aEE} \Rightarrow Qy < \alpha_{cEE} Fy_{cEE} + \alpha_{aEE} Fy_{aEE} \Rightarrow Qy < \alpha_{cEE} Fy_{cEE} + \alpha_{aEE} Fy_{aEE} + \alpha_{cNE} Fy_{cNE}$ . This contradicts the fea-

sible set of  $Fy_{cEE}$ .

$$If y > y_{1-\alpha_{cNE}}^{111}, Gy - (\frac{\pi_{cEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{cEE} + \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{aEE}) = 1 - \frac{\pi_{cEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{cEE} - \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} Fy_{aEE} \ge 0.$$

Since  $E[Y(1)|aEE] = \overline{Y}^{011}$ ,  $\widetilde{LY}_{1,cEE}$  is the smallest value for E[Y(1)|cEE].

Next, I show that there exist distributions of *cEE*, *aEE* and *cNE* in the trimming cell, such that  $E[Y(1)|cEE] = \widetilde{LY}_{1,cEE}$  and  $E[Y(1)|aEE] = \overline{Y}^{011}$ .

When  $\overline{Y}^{011} \leq \overline{y}_{aEE}$  and  $\overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111})$ , let sy be the observed density of  $Y(Y \geq y_{1-\alpha_{cNE}}^{111})$ , ty the observed density of  $Y(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111})$  and hy the observed density of  $Y(Y \leq y_{\alpha_{aEE}}^{111})$ . Then  $\exists \tau \in [0,1]$ , s.t.  $\tau \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}) + (1-\tau)\overline{Y}(Y \leq y_{\alpha_{aEE}}^{111}) = \overline{Y}^{011}$ . Therefore,  $sy = fy_{cNE}$ ,  $\tau ty + (1-\tau)hy = fy_{aEE}$  and  $(1 - \frac{\tau \pi_{aEE}}{\pi_{cEE}})ty + \frac{\tau \pi_{aEE}}{\pi_{cEE}}hy = fy_{cEE}$ .

When  $\overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}) \leq \overline{Y}^{011} \leq \overline{y}_{aEE}$ , let *sy* be the observed density of  $Y(Y \geq y_{1-\alpha_{cNE}}^{111})$ , *ty* the observed density of  $Y(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111})$  and *hy* the observed density of  $Y(Y \leq y_{\alpha_{cEE}}^{111})$ . Then  $\exists \tau \in [0, 1]$ , *s.t.*  $\tau \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}) + (1-\tau)\overline{Y}(Y \leq y_{\alpha_{cEE}}^{111})) = \overline{Y}^{011}$ . Therefore,  $sy = fy_{cNE}$ ,  $\tau ty + (1-\tau)hy = fy_{aEE}$  and  $\frac{(1-\tau)\pi_{aEE}}{\pi_{cEE}}ty + (1-\tau)\frac{(1-\tau)\pi_{aEE}}{\pi_{cEE}})hy = fy_{cEE}$ .

For the second-step proof, I have four cases to discuss, taking into account the lower and upper bound simultaneously. Since they form different segmentations of Qy, in each case I use some cutoff values to discuss  $\forall \theta \in [LY_{1,cEE}, cutoff]$  and  $\forall \theta \in [cutoff, UY_{1,cEE}]$  separately. In either interval for  $\theta$ , I have to discuss the range of  $\overline{Y}^{011}$  as I have done in the first-step proof. The four cases are listed as follows.

1. 
$$LY_{1,cEE} = \overline{Y}(Y \le y_{\alpha_{cEE}}^{111}), UY_{1,cEE} = \overline{Y}(Y \ge y_{1-\alpha_{cEE}}^{111})$$

It happens when  $\overline{y}_{aEE} \leq \overline{Y}^{011} \leq \widetilde{y}_{aEE}$ ; in other words,  $\pi_{cNE} \geq \pi_{cEE}$ . Since  $\overline{Y}(Y \leq y_{\alpha_{cEE}}^{111}) \leq \overline{Y}^{111} \leq \overline{Y}(Y \geq y_{\alpha_{cEE}}^{111})$  and  $\overline{Y}(Y \leq y_{1-\alpha_{cEE}}^{111}) \leq \overline{Y}(Y \geq y_{1-\alpha_{cEE}}^{111})$ , thus,  $cutoff = \overline{Y}^{111}$ .

For  $\forall \theta \in [\overline{Y}(Y \leq y_{\alpha_{cEE}}^{111}), cut of f]$ , it is necessary to discuss two cases when  $\overline{Y}^{011} \geq \overline{y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}) \text{ and } \overline{y}_{aEE} \leq \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}).$ For  $\forall \theta \in [cut of f, \overline{Y}(Y \geq y_{1-\alpha_{cEE}}^{111})]$ , it is necessary to discuss two cases when  $\overline{Y}^{011} \leq \overline{y}_{aEE} \& \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111})]$ , it is necessary to  $\overline{Y}^{011} \leq \overline{Y}^{011} \leq \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111}))$  and  $\overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111}) \leq \overline{Y}^{011} \leq \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111}))$ ,  $UY_{1,cEE} = \widetilde{UY}_{1,cEE}$ 

In this case, I have to discuss the relationship between  $\overline{y}_{aEE}$  and  $\widetilde{y}_{aEE}$  first. I solve this problem with two different cutoffs:

For  $\forall \theta \in [\overline{Y}(Y \leq y_{\alpha_{cEE}}^{111}), cut of f]$ , I have two cases to discuss:  $\overline{Y}^{011} \geq \overline{y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$  and  $\overline{y}_{aEE} \leq \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$ .

For  $\forall \theta \in [cut of f, \widetilde{UY}_{1,cEE}]$ , I have two cases to discuss:  $\overline{Y}^{011} \geq \widetilde{y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y^{111}_{\alpha_{cNE}} \leq Y \leq y^{111}_{1-\alpha_{aEE}}))$  and  $\widetilde{y}_{aEE} \leq \overline{Y}^{011} \leq \overline{Y}(y^{111}_{\alpha_{cNE}} \leq Y \leq y^{111}_{1-\alpha_{aEE}}))$ . 3.  $LY_{1,cEE} = \widetilde{LY}_{1,cEE}, UY_{1,cEE} = \overline{Y}(Y \geq y^{111}_{1-\alpha_{cEE}})$ 

Like case 2, I have to discuss the relationship between  $\overline{y}_{aEE}$  and  $\widetilde{y}_{aEE}$  first. I solve this problem with two different cutoffs:

For  $\forall \theta \in [\widetilde{LY}_{1,cEE}, cutoff]$ , I have two cases to discuss:  $\overline{Y}^{011} \leq \overline{y}_{aEE} \& \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}) \text{ and } \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}) \leq \overline{Y}^{011} \leq \overline{y}_{aEE}.$ 

For  $\forall \theta \in [cutoff, \overline{Y}(Y \ge y_{1-\alpha_{cEE}}^{111})]$ , I have two cases to discuss:  $\overline{Y}^{011} \le \widetilde{y}_{aEE} \& \overline{Y}^{011} \le \overline{Y}(y_{\alpha_{aEE}}^{111} \le Y \le y_{1-\alpha_{cEE}}^{111}))$  and  $\overline{Y}(y_{\alpha_{aEE}}^{111} \le Y \le y_{1-\alpha_{cEE}}^{111}) \le \overline{Y}^{011} \le \widetilde{y}_{aEE}$ . 4.  $LY_{1,cEE} = \widetilde{LY}_{1,cEE}, UY_{1,cEE} = \widetilde{UY}_{1,cEE}$ 

This happens when  $\tilde{y}_{aEE} \leq \overline{Y}^{011} \leq \overline{y}_{aEE}$ ; in other words,  $\pi_{cNE} \leq \pi_{cEE}$ . In this case, it is difficult to get a uniform cutoff. I discuss multiple cases conditional on the proportions of the strata.

When  $\pi_{cNE} \leq \pi_{cEE}$  and  $\pi_{aEE} \leq \pi_{cEE}$ , I discuss the intervals  $\forall \theta \in [\widetilde{LY}_{1,cEE}, \overline{y}_{aEE}]$ and  $\forall \theta \in [\widetilde{y}_{aEE}, \widetilde{UY}_{1,cEE}]$  to complete the proof in the entire range, i.e.,  $\forall \theta \in [\widetilde{LY}_{1,cEE}, \widetilde{UY}_{1,cEE}]$ .  $\widetilde{UY}_{1,cEE}]$ . For  $\forall \theta \in [\widetilde{LY}_{1,cEE}, \overline{y}_{aEE}]$ , the two cases are  $\overline{Y}^{011} \leq \overline{y}_{aEE} \& \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}))$  and  $\overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}) \leq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}))$  and  $\widetilde{Y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}))$  and  $\widetilde{y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}))$ .

When  $\pi_{cNE} \leq \pi_{cEE} \leq \pi_{aEE}$ , I discuss the intervals  $\forall \theta \in [\widetilde{LY}_{1,cEE}, \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111})]$  and  $\forall \theta \in [\overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}), \widetilde{UY}_{1,cEE}]$  to complete the proof in the entire range. For  $\forall \theta \in [\widetilde{LY}_{1,cEE}, \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111})]$ , I have  $\overline{Y}^{011} \leq \overline{y}_{aEE} \& \overline{Y}^{011} \leq y_{aEE}$ .

$$\overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cNE}}^{111}). \text{ For } \forall \theta \in [\overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}), \widetilde{UY}_{1,cEE}], \text{ I have } \overline{Y}^{011} \geq \widetilde{y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}).$$

From all the above, I can find that though cutoffs are different in the four cases, the discussion of  $\overline{Y}^{011}$  is repeated. Cases 2 and 3 compose a complete discussion of  $\overline{Y}^{011}$ . In the following, I only write the proof of Case 2. Case 3 can be shown in a similar way.

*First*, let me discuss the relationship between  $\overline{y}_{aEE}$  and  $\widetilde{y}_{aEE}$  in Case 2 and derive two different cutoffs.

When  $\overline{y}_{aEE} \leq \widetilde{y}_{aEE} \leq \overline{Y}^{011}$  (i.e.,  $\pi_{cNE} \geq \pi_{cEE}$ ), I have  $LY_{1,cEE} \leq \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}) \leq \overline{Y}(Y \geq y_{\alpha_{cEE}}^{111})$  and  $\widetilde{UY}_{1,cEE} = \overline{Y}(Y \geq y_{\alpha_{cNE}}^{111}) \frac{\pi_{cEE} + \pi_{aEE}}{\pi_{cEE}} - \overline{Y}^{011} \frac{\pi_{aEE}}{\pi_{cEE}} \geq \overline{Y}(Y \geq y_{\alpha_{cNE}}^{111})) \frac{\pi_{cEE} + \pi_{aEE}}{\pi_{cEE}} - \overline{Y}(Y \geq y_{1-\alpha_{aEE}}^{111}) \frac{\pi_{aEE}}{\pi_{cEE}} = \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}) \geq \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$ . The first inequality is derived from  $\overline{Y}^{011} \leq \overline{Y}(Y \geq y_{1-\alpha_{aEE}}^{111})$ , and the last inequality is derived from  $\pi_{cNE} \geq \pi_{cEE}$ . Thus,  $\widetilde{UY}_{1,cEE} \geq \overline{Y}(y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111}) \geq \overline{Y}(Y_{\alpha_{cEE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$ .

When  $\widetilde{y}_{aEE} \leq \overline{y}_{aEE} \leq \overline{Y}^{011}(\pi_{cNE} \leq \pi_{cEE}), LY_{1,cEE} \leq \overline{Y}(y^{111}_{\alpha_{cNE}} \leq Y \leq y^{111}_{1-\alpha_{aEE}}) \leq \overline{Y}(Y \geq y^{111}_{\alpha_{cNE}}) \leq \overline{Y}(Y \geq y^{111}_{\alpha_{cEE}}).$  The last inequality is derived from  $\pi_{cNE} \leq \pi_{cEE}$ . Thus,  $LY_{1,cEE} \leq \overline{Y}(y^{111}_{\alpha_{cNE}} \leq Y \leq y^{111}_{1-\alpha_{aEE}}) \leq \overline{Y}(Y \geq y^{111}_{\alpha_{cEE}}).$  And  $\widetilde{UY}_{1,cEE} \geq \overline{Y}(y^{111}_{\alpha_{cNE}} \leq Y \leq y^{111}_{1-\alpha_{aEE}}) \geq \overline{Y}(Y \leq y^{111}_{\alpha_{cNE}}).$  Its derivation is the same as that in the last paragraph. Therefore,  $cutoff = \overline{Y}(y^{111}_{\alpha_{cNE}} \leq Y \leq y^{111}_{1-\alpha_{aEE}}).$ 

Second,  $\forall \theta \in [cutoff, \widetilde{UY}_{1,cEE}]$ , where cutoff is the corresponding value in either case  $\overline{y}_{aEE} \leq \widetilde{y}_{aEE}$  or  $\widetilde{y}_{aEE} \leq \overline{y}_{aEE}$ ,  $\exists \lambda \in (0, 1], s.t. \lambda \widetilde{UY}_{1,cEE} + (1 - \lambda) \overline{Y} (Y \leq y_{\alpha_{cNE}}^{111}) = \theta$ , since  $\overline{Y}(Y \leq y_{\alpha_{cNE}}^{111}) \leq \theta \leq \widetilde{UY}_{1,cEE}$ .  $(\lambda = \frac{\theta - \overline{Y}(Y \leq y_{\alpha_{cNE}}^{111})}{\widetilde{UY}_{1,cEE} - \overline{Y}(Y \leq y_{\alpha_{cNE}}^{111})}$ .)

To construct  $fy_{aEE}$ , it is necessary to discuss the value of  $\overline{Y}^{011}$ . One case is  $\overline{Y}^{011} \ge \overline{Y}(y_{\alpha_{cNE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ . This happens when either  $\pi_{cEE} \le \pi_{aEE}$ , or  $\pi_{cEE} \ge \pi_{aEE}$ 

but aEE take up the very top quantiles of the observed distribution. The other case is  $\overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$ . This is true only when  $\pi_{cEE} \geq \pi_{aEE}$ . (1)  $\overline{Y}^{011} \geq \widetilde{y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$ 

Let ty be the observed density of  $Y(y \ge y_{1-\alpha_{aEE}}^{111})$ , hy the density of  $Y(y_{\alpha_{cNE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$  and gy the density of  $Y(y \le y_{\alpha_{cNE}}^{111})$ . Since  $\overline{Y}(Y \le y_{1-\alpha_{aEE}}^{111}) \le \overline{Y}^{011} \le \overline{Y}(Y \ge y_{1-\alpha_{aEE}}^{111}))$ ,  $\exists \tau \in (0,1]$ , s.t.  $\tau \overline{Y}(Y \ge y_{1-\alpha_{aEE}}^{111}) + (1-\tau)\overline{Y}(Y \le y_{1-\alpha_{aEE}}^{111}) = \overline{Y}^{011}$ .  $\tau ty + (1-\tau) \frac{\pi_{cNE}}{\pi_{cNE} + \pi_{cEE}} hy + (1-\tau) \frac{\pi_{cNE}}{\pi_{cNE} + \pi_{cEE}} gy = fy_{aEE}$ . Since  $\widetilde{UY}_{1,cEE}$  is obtained by temporarily assuming aEE is above  $y_{\alpha_{cNE}}^{111}$ ,  $\exists \phi \in [0,1]$ , s.t.  $\phi \overline{Y}(Y \ge y_{1-\alpha_{aEE}}^{111}) + (1-\phi)\overline{Y}(y_{\alpha_{cNE}}^{111}) \le \overline{Y}^{011}$ . Thus,  $\lambda \{\frac{\pi_{aEE}}{\pi_{cEE}} ty + hy - [\phi ty + (1-\phi)hy]\frac{\pi_{aEE}}{\pi_{cEE}}\} + (1-\lambda)gy = fy_{cEE}$ . Since  $\alpha_{aEE}ty + \alpha_{cEE}hy + \alpha_{cNE}gy = \alpha_{aEE}fy_{aEE} + \alpha_{cEE}fy_{cEE} + \alpha_{cNE}fy_{cNE}$ , the corresponding density for cNE is  $\frac{\pi_{aEE}}{\pi_{cNE}}[1-\tau-\lambda(1-\phi)]ty + \frac{\pi_{cEE}}{\pi_{cNE}}[1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}} - \lambda(1-(1-\phi)\frac{\pi_{aEE}}{\pi_{cEE}})]hy + [1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}} - (1-\lambda)\frac{\pi_{cEE}}{\pi_{cNE}}]gy = fy_{cNE}$ . (2)  $\widetilde{y}_{aEE} \le \overline{Y}^{011} \le \overline{Y}(y_{\alpha_{cNE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ 

Let ty be the observed density of  $Y(y \ge y_{1-\alpha_{cEE}}^{111})$ , hy the density of  $Y(y_{\alpha_{cNE}}^{111} \le Y \le y_{1-\alpha_{cEE}}^{111})$  and gy the density of  $Y(y \le y_{\alpha_{cNE}}^{111})$ . Since  $\overline{Y}(Y \le y_{\alpha_{cNE}}^{111}) \le \overline{Y}^{011} \le \overline{Y}(Y \ge y_{\alpha_{cNE}}^{111})$ ,  $\exists \tau \in (0,1)$ ,  $s.t. \ \tau \overline{Y}(Y \ge y_{\alpha_{cNE}}^{111}) + (1-\tau)\overline{Y}(Y \le y_{\alpha_{cNE}}^{111}) = \overline{Y}^{011}$ .  $\tau \frac{\pi_{cEE}}{\pi_{aEE} + \pi_{cEE}} ty + \tau \frac{\pi_{aEE}}{\pi_{aEE} + \pi_{cEE}} hy + (1-\tau)gy = fy_{aEE}$ . As in (1), since  $\widetilde{UY}_{1,cEE}$  is obtained by temporarily assuming aEE is above  $y_{\alpha_{cNE}}^{111}$ ,  $\exists \phi \in [0,1]$ ,  $s.t. \ \phi \overline{Y}(Y \ge y_{1-\alpha_{cEE}}^{111}) + (1-\phi)\widetilde{y}_{aEE} = \overline{Y}^{011}$ . Thus,  $\lambda \{ty + \frac{\pi_{aEE}}{\pi_{cEE}} hy - [\phi ty + (1-\phi)hy] \frac{\pi_{aEE}}{\pi_{cEE}} \} + (1-\lambda)gy = fy_{cEE}$ . Similarly as in (1), I finally get  $\frac{\pi_{cEE}}{\pi_{cNE}} [1 - \tau \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} - \lambda(1 - \phi \frac{\pi_{aEE}}{\pi_{cEE}})]ty + \frac{\pi_{aEE}}{\pi_{cNE}} [1 - \tau \frac{\pi_{aEE}}{\pi_{cEE} + \pi_{aEE}} - \lambda\phi]hy + [1 - (1 - \tau) \frac{\pi_{aEE}}{\pi_{cNE}} - (1 - \lambda) \frac{\pi_{cEE}}{\pi_{cEE}}]gy = fy_{cNE}$ . Third,  $\forall \theta \in [\overline{Y}(Y \le y_{\alpha_{cEE}}^{111}), cut of f], \exists \lambda \in [0, 1), s.t. \ \lambda \overline{Y}(Y \ge y_{\alpha_{cEE}}^{111}) + (1 - \lambda)\overline{Y}(Y \le y_{\alpha_{cEE}}^{111})$ . To discuss the value of  $\overline{Y}^{011}$ , one case is  $\overline{Y}^{011} \ge \overline{Y}(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ . It happens when either  $\pi_{cNE} \le \pi_{aEE}$ , or  $\pi_{cNE} \ge \pi_{aEE}$  but aEE take up very top quantiles of the observed distribution. The other case is  $\overline{Y}^{011} \le \overline{Y}(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$  when  $\pi_{cNE} \ge \pi_{aEE}$ .

(1) 
$$\overline{Y}^{011} \ge \overline{y}_{aEE} \& \overline{Y}^{011} \ge \overline{Y}(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$$

Let ty be the observed density of  $Y(y \ge y_{1-\alpha_{aEE}}^{111})$ , hy the density of  $Y(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ , and gy the density of  $Y(y \le y_{\alpha_{cEE}}^{111})$ . Since  $\overline{Y}(Y \le y_{1-\alpha_{aEE}}^{111}) \le \overline{Y}^{011} \le \overline{Y}(Y \ge y_{1-\alpha_{aEE}}^{111}))$ ,  $\exists \tau \in (0,1]$ ,  $s.t. \ \tau \overline{Y}(Y \ge y_{1-\alpha_{aEE}}^{111}) + (1-\tau)\overline{Y}(Y \le y_{1-\alpha_{aEE}}^{111}) = \overline{Y}^{011}$ .  $\tau ty + (1-\tau)\frac{\pi_{cNE}}{\pi_{cNE}+\pi_{cEE}}hy + (1-\tau)\frac{\pi_{cEE}}{\pi_{cNE}+\pi_{cEE}}gy = fy_{aEE}$ . For cEE, I have  $\lambda \frac{\pi_{aEE}}{\pi_{cNE}+\pi_{aEE}}ty + \lambda \frac{\pi_{cEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\lambda)gy = fy_{cEE}$ . Finally,  $\frac{\pi_{aEE}}{\pi_{cNE}}(1-\tau-\lambda \frac{\pi_{cEE}}{\pi_{aEE}+\pi_{cNE}})ty + [1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}]hy + \frac{\pi_{cEE}}{\pi_{cNE}}[\lambda - (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}})]gy = fy_{cNE}$ . (2)  $\overline{y}_{aEE} \le \overline{Y}^{011} \le \overline{Y}(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ 

Let ty be the observed density of  $Y(y \ge y_{1-\alpha_{cNE}}^{111})$ , hy the density of  $Y(y_{\alpha_{cEE}}^{111} \le Y \le y_{1-\alpha_{cNE}}^{111})$ , and gy the density of  $Y(y \le y_{\alpha_{cEE}}^{111})$ . Since  $\overline{Y}(Y \le y_{1-\alpha_{cNE}}^{111}) \le \overline{Y}^{011} \le \overline{Y}(Y \ge y_{1-\alpha_{cNE}}^{111}))$ ,  $\exists \tau \in (0,1]$ ,  $s.t. \ \tau \overline{Y}(Y \ge y_{1-\alpha_{cNE}}^{111}) + (1-\tau)\overline{Y}(Y \le y_{1-\alpha_{cNE}}^{111}) = \overline{Y}^{011}$ .  $\tau ty + (1-\tau)\frac{\pi_{aEE}}{\pi_{aEE}+\pi_{cEE}}hy + (1-\tau)\frac{\pi_{cEE}}{\pi_{aEE}+\pi_{cEE}}gy = fy_{aEE}$ . For cEE, I have  $\lambda \frac{\pi_{cNE}}{\pi_{cNE}+\pi_{aEE}}ty + \lambda \frac{\pi_{aEE}}{\pi_{cNE}+\pi_{aEE}}hy + (1-\lambda)gy = fy_{cEE}$ . Finally,  $(1-\tau \frac{\pi_{aEE}}{\pi_{cNE}} - \lambda \frac{\pi_{cEE}}{\pi_{aEE}+\pi_{cNE}})ty + \frac{\pi_{aEE}}{\pi_{cNE}}[1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{aEE}}-\lambda \frac{\pi_{cEE}}{\pi_{aEE}+\pi_{cNE}}]hy + \frac{\pi_{cEE}}{\pi_{cNE}}[\lambda - (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{aEE}}]gy = fy_{cNE}$ .

### A.2.2 **Proof of Proposition 3.2**

The proof of Proposition 3.2 is similar to that of Proposition 3.1, except for two differences: multiple cases reduce to two due to  $L_{cEE}$ , and the constructed distributions for cEE and cNE should also satisfy the mean dominance assumption. **Proof.** As in the proof of Proposition 3.1, I first show that  $LY_{1,cEE}$  is the smallest feasible value for E[Y(1)|cEE], and then show  $\forall \theta \in [LY_{1,cEE}, UY_{1,cEE}]$ , there exist distributions  $Fy_{cEE}$ ,  $Fy_{aEE}$  and  $Fy_{cNE}$  satisfying Assumptions 3.1 through 3.6.

For 
$$\forall \theta \in [LY_{1,cEE}, UY_{1,cEE}]$$
, by equation (3.6), I have  

$$E[Y(1)|cEE] - E[Y(1)|cNE]$$

$$= \theta - (\overline{Y}^{111} \frac{\pi_{cEE} + \pi_{cNE} + \pi_{aEE}}{\pi_{cNE}} - \theta \frac{\pi_{cEE}}{\pi_{cNE}} - \overline{Y}^{011} \frac{\pi_{aEE}}{\pi_{cNE}})$$

$$\geq (1 + \frac{\pi_{cEE}}{\pi_{cNE}})LY_{1,cEE} + \overline{Y}^{011} \frac{\pi_{aEE}}{\pi_{cNE}} - \overline{Y}^{111} \frac{\pi_{cEE} + \pi_{cNE} + \pi_{aEE}}{\pi_{cNE}} = 0.$$

If there was another lower bound smaller than  $LY_{1,cEE}$ , E[Y(1)|cEE] - E[Y(1)|cNE]would be negative when E[Y(1)|cEE] reached that lower bound. This contradicts Assumption 3.6. Thus,  $LY_{1,cEE}$  is the smallest value for E[Y(1)|cEE] under Assumptions 3.1 through 3.6.

As in Proposition 3.1, I have to show the distributions exist when  $E[Y(1)|cEE] = LY_{1,cEE}$  and  $E[Y(1)|aEE] = \overline{Y}^{011}$ . According to the range of  $\overline{Y}^{011}$ , I have four cases to discuss:

$$(1) \overline{Y}^{011} \leq \widetilde{y}_{aEE} \& \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111})$$

$$(2) \overline{Y}(y_{\alpha_{aEE}}^{111} \leq Y \leq y_{1-\alpha_{cEE}}^{111}) \leq \overline{Y}^{011} \leq \widetilde{y}_{aEE}$$

$$(3) \overline{Y}^{011} \geq \widetilde{y}_{aEE} \& \overline{Y}^{011} \geq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$$

$$(4) \widetilde{y}_{aEE} \leq \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$$

These four cases correspond with those in the second-step proof that for  $\forall \theta \in [LY_{1,cEE}, UY_{1,cEE}]$ , there exist distributions  $Fy_{cEE}$ ,  $Fy_{aEE}$ , and  $Fy_{cNE}$  satisfying Assumptions 3.1 through 3.6. Since for  $\forall \theta \in [LY_{1,cEE}, UY_{1,cEE}]$ ,  $\exists \lambda \in [0, 1]$ , *s.t.*  $\lambda UY_{1,cEE} + (1 - \lambda)LY_{1,cEE} = \theta$ , the proof that the distributions exist when  $E[Y(1)|cEE] = LY_{1,cEE}$  and  $E[Y(1)|aEE] = \overline{Y}^{011}$  is a special case of the second-step proof, i.e.  $\lambda = 0$ . In the fol-

lowing, I take Case (4) as an example to illustrate the second-step proof of Proposition 3.2.

(4) 
$$\widetilde{y}_{aEE} \leq \overline{Y}^{011} \leq \overline{Y}(y_{\alpha_{cNE}}^{111} \leq Y \leq y_{1-\alpha_{aEE}}^{111})$$

Let ty be the observed density of  $Y(y \ge y_{1-\alpha_{cEE}}^{111})$ , hy the density of  $Y(y_{\alpha_{cNE}}^{111} \le Y \le y_{1-\alpha_{cEE}}^{111})$ , and gy the density of  $Y(y \le y_{\alpha_{cNE}}^{111})$ .  $\forall \theta \in [LY_{1,cEE}, \widetilde{UY}_{1,cEE}], \exists \lambda \in [0, 1], s.t. \lambda \widetilde{UY}_{1,cEE} + (1-\lambda)LY_{1,cEE} = \theta$ . As in the proof of Proposition 3.1, since  $\widetilde{y}_{aEE} \le \overline{Y}^{011} \le \overline{Y}(y_{\alpha_{cNE}}^{111} \le Y \le y_{1-\alpha_{aEE}}^{111})$ ,  $\exists \tau \in (0, 1], s.t. \tau \overline{Y}(Y \ge y_{1-\alpha_{cEE}}^{111}) + (1-\tau)\overline{Y}(Y \le y_{1-\alpha_{cEE}}^{111}) = \overline{Y}^{011}$ . And  $\tau ty + (1-\tau)\frac{\pi_{aEE}}{\pi_{cNE}+\pi_{aEE}}hy + (1-\tau)\frac{\pi_{cNE}}{\pi_{cNE}+\pi_{aEE}}gy = fy_{aEE}$ . Since  $\widetilde{UY}_{1,cEE}$  is obtained by temporarily assuming aEE is above  $y_{\alpha_{cNE}}^{111}, \exists \phi \in [0, 1], s.t. \phi \overline{Y}(Y \ge y_{1-\alpha_{cEE}}^{111}) + (1-\phi)\widetilde{y}_{aEE} = \overline{Y}^{011}$ . Thus,  $\lambda \{ty + \frac{\pi_{aEE}}{\pi_{cEE}}hy - [\phi ty + (1-\phi)hy]\frac{\pi_{aEE}}{\pi_{cEE}}\} + (1-\lambda)(\frac{\pi_{cEE}}{\pi_{cEE}+\pi_{cNE}}ty + \frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + \frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\phi)hy]\frac{\pi_{aEE}}{\pi_{cEE}} + (1-\lambda)(\frac{\pi_{cEE}}{\pi_{cEE}+\pi_{cNE}}ty + \frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\phi)hy]\frac{\pi_{aEE}}{\pi_{cEE}}hy = fy_{cEE}$ . After some algebra,  $[\lambda(1-\phi\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}]\}hy + (1-\lambda)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}[1-(1-\tau)\frac{\pi_{aEE}}{\pi_{aEE}+\pi_{cNE}}hy + (1-\lambda)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{aEE}+\pi_{cNE}}hy + (1-\lambda)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy + (1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}hy = fy_{cEE}$ . Then, the corresponding density for cNE is  $\{-\frac{\pi_{aEE}}{\pi_{cNE}}\tau + \frac{\pi_{cEE}}{\pi_{cNE}}[1-\lambda(1-\phi\frac{\pi_{aEE}}{\pi_{cEE}}) - (1-\lambda)(\frac{\pi_{aEE}}{\pi_{aEE}+\pi_{cNE}}}hy + (1-(1-\lambda)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}]\}hy + [1-(1-\lambda)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}}hy + (1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}}hy + (1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}}hy + (1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}}hy + (1-(1-\tau)\frac{\pi_{aEE}}{\pi_{cEE}+\pi_{cNE}}}hy + (1-(1-\tau)\frac{\pi_{aE$ 

The inequality in the second paragraph of this proof shows that Assumption 3.6 holds, as long as  $E[Y(1)|cEE] \ge LY_{1,cEE}$ . Since  $\theta \in [LY_{1,cEE}, UY_{1,cEE}]$  holds by construction, the constructed densities satisfy Assumption 3.6.

#### APPENDIX B

#### **PROCEDURE FOR ESTIMATION AND INFERENCE**

In this section, I use the upper bound for E[Y(1)|cEE] in Proposition 3.1 as an illustration to show how to employ the methodology proposed by Chernozhukov, Lee, and Rosen (2011) (hereafter CLR) to calculate the half-median-unbiased estimators and the confidence interval for the true parameter of interest when the bounds involve max or min operator.

The upper bound for  $\theta_0 = E[Y(1)|cEE]$  is given by  $\theta_0^u = \min_{v \in \mathscr{V} = \{1,2\}} \theta^u(v)$ , with  $\theta^u(1) = \overline{Y}(Y \ge y_{1-\alpha_{cEE}}^{111})$  and  $\theta^u(2) = \overline{Y}(Y \ge y_{\alpha_{cNE}}^{111}) \frac{q_{1|0}-p_{01|1}}{p_{01|0}-p_{01|1}} - \overline{Y}^{011} \frac{p_{11|0}}{p_{01|0}-p_{01|1}}$ . Then the precision-corrected estimate of  $\theta_0^u$  is given by  $\widehat{\theta}^u(p) = \min_{v \in \{1,2\}} [\widehat{\theta}^u(v) + k(p)s(v)]$ , where  $\widehat{\theta}^u(v)$  is a consistent estimate of  $\theta^u(v)$ , s(v) is its standard error and k(p) is a critical value.

The selection of k(p) relies on standardized process  $Z_n(v) = \{\theta^u(v) - \hat{\theta}^u(v)\}/\sigma(v)$ , where  $\sigma(v)/s(v) \rightarrow 1$  uniformly in v. CLR (2011) approximate this process by a standardized Gaussian process  $Z_n^*(v)$ . Specifically, for any compact set V, CLR (2011) approximate by simulation the *p*-th quantile of  $\sup_{v \in V} Z_n^*(v)$ , denoted by  $k_{n,V}(p)$ , and use it in place of k(p). Since setting  $V = \mathcal{V}$  leads to asymptotically valid but conservative inference, CLR (2011) propose a preliminary set estimator  $\hat{V}_n$  of  $V_0 = \arg \min_{v \in \mathcal{V}} \theta^u(v)$ for the upper bound (where argmin is replaced by argmax for the lower bound), which they call an adaptive inequality selector. Intuitively,  $\hat{V}_n$  selects the bounding functions that are close enough to binding to affect the asymptotic distribution of the estimators of the upper and lower bounds. Now let me describe the precise procedure to obtain half-median-unbiased estimators for the upper bounds of  $\Delta$  and E[Y(1)|cEE] in Proposition 3.1.<sup>24</sup> These upper bounds can be written as  $\theta_{n0}^u = \min_{v \in \mathscr{V} = \{1,2\}} \theta_n^u(v)$ , where the bounding functions  $\theta_n^u(v)$ are given in Proposition 3.1, as illustrated in the example above.<sup>25</sup> Let  $\gamma_n = [\theta_n^u(1) \\ \theta_n^u(2)]'$  be the vector containing the two bounding functions and let  $\hat{\gamma}_n = [\hat{\theta}_n^u(1) \\ \hat{\theta}_n^u(2)]'$  denote its sample analog estimator, which can be shown to be consistent and asymptotically normally distributed using standard results (e.g., Newey and McFadden, 1994; Lee, 2009). The specific steps are as follows.

- 1. Let  $\Omega_n$  denote the asymptotic variance of  $\sqrt{n}(\hat{\gamma}_n \gamma_n)$ . A consistent estimate of  $\Omega_n$ ,  $\widehat{\Omega}_n$ , is obtained by a 5000-repetition bootstrap. Let  $\widehat{g}_n(v)'$  denote the  $v^{th}$  row of  $\widehat{\Omega}_n^{1/2}$ ,  $s_n(v) = \|\widehat{g}_n(v)\| / \sqrt{n}$  and, following CLR (2011), set  $c_n = 1 (.1/\log n)$ .
- 2. Simulate R = 1,000,000 draws from  $\mathcal{N}(0,I_2)$ , denoted  $Z_1,...,Z_R$ , where  $I_2$  is a  $2 \times 2$  identity matrix, and let  $Z_r^*(v) = \widehat{g}_n(v)'Z_r / \|\widehat{g}_n(v)\|$  for r = 1,...,R.
- 3. Let Q<sub>p</sub>(X) denote the *p*-th quantile of a random variable X. Compute k<sub>n,V</sub>(c<sub>n</sub>) = Q<sub>c<sub>n</sub></sub>(max<sub>v∈V</sub> Z<sub>r</sub><sup>\*</sup>(v), r = 1,...,R); that is, for each replication r calculate the maximum of Z<sub>r</sub><sup>\*</sup>(1) and Z<sub>r</sub><sup>\*</sup>(2), and take the *c*-th quantile of those R values. Then, use the critical value to compute the set estimator \$\hat{V}\_n = {v ∈ V : \hat{\theta}\_n^u(v) ≤ min\_{v∈V} {[\hat{\theta}\_n^u(v) + k\_{n,V}(c\_n)s\_n(v)] + 2k\_{n,V}(c\_n)s\_n(v)}.
- Compute k<sub>n,V<sub>n</sub></sub>(p) = Q<sub>p</sub>(max<sub>v∈V<sub>n</sub></sub>Z<sup>\*</sup><sub>r</sub>(v), r = 1,...,R), so that the critical value is based on V̂<sub>n</sub> instead of 𝒴.

<sup>&</sup>lt;sup>24</sup>For further details on the procedure see CLR (2011), especially in Appendix A and Section 4.1 <sup>25</sup>The subscript "*n*" indicates local parameters.

5. To get the half-median-unbiased estimator of  $\theta_{n0}^{u}$ ,  $\hat{\theta}_{n}^{u}(1/2)$ , Set p = 1/2 and compute  $\hat{\theta}_{n}^{u}(1/2) = \min_{v \in \mathscr{V}} [\hat{\theta}_{n}^{u}(v) + k_{n,\widehat{V}_{n}}(1/2)s_{n}(v)].$ 

To obtain half-median-unbiased estimators for the lower bounds in Proposition 3.1, which have the form  $\theta_{n0}^{l} = \max_{v \in \mathscr{V} = \{1,2\}} \theta_{n}^{l}(v)$ ,  $\widehat{V}_{n}$  in step 3 above is replaced by  $\widehat{V}_{n} = \{v \in \mathscr{V} : \widehat{\theta}_{n}^{l}(v) \ge \max_{\widetilde{v} \in \mathscr{V}} [\widehat{\theta}_{n}^{l}(\widetilde{v}) - k_{n,\mathscr{V}}(c_{n})s_{n}(\widetilde{v})] - 2k_{n,\mathscr{V}}(c_{n})s_{n}(v)\}$ , and in step 5 I set  $\widehat{\theta}_{n}^{l}(1/2) = \max_{v \in \mathscr{V}} [\widehat{\theta}_{n}^{l}(v) - k_{n,\widehat{V}_{n}}(1/2)s_{n}(v)].^{26}$ 

To describe the construction of confidence intervals for the true parameter  $\theta_0$ , let  $\hat{\theta}_n^u(p) = \min_{v \in \mathscr{V}} [\hat{\theta}_n^u(v) + k_{n,\hat{V}_n}(p)s_n(v)]$  and  $\hat{\theta}_n^l(p) = \max_{v \in \mathscr{V}} [\hat{\theta}_n^l(v) - k_{n,\hat{V}_n}(p)s_n(v)]$ , where the critical values are obtained as described above. Following CLR (2011), let  $\hat{\Gamma}_n = \hat{\theta}_n^u(1/2) - \hat{\theta}_n^l(1/2)$ ,  $\hat{\Gamma}_n^+ = \max(0,\hat{\Gamma}_n)$ ,  $\rho_n = \max\{\hat{\theta}_n^u(3/4) - \hat{\theta}_n^u(1/4), \hat{\theta}_n^l(1/4) - \hat{\theta}_n^l(3/4)\}$ ,  $\tau_n = 1/(\rho_n \log n)$  and  $\hat{p}_n = 1 - \Phi(\tau_n \hat{\Gamma}_n^+)\alpha$ , where  $\Phi(\cdot)$  is the standard normal CDF. Note that  $\hat{p}_n \in [1 - \alpha, 1 - \alpha/2]$ , with  $\hat{p}_n$  approaching  $1 - \alpha/2$  when the length of the identified set tends to zero. Then, an asymptotically valid  $1 - \alpha$  confidence interval for  $\theta_0$  is given by  $[\hat{\theta}_n^l(\hat{p}_n), \hat{\theta}_n^u(\hat{p}_n)]$ , i.e.,  $\inf_{\theta_0 \in [\theta_{n0}^l, \theta_{n0}^u]} P(\theta_0 \in [\hat{\theta}_n^l(\hat{p}_n), \hat{\theta}_n^u(\hat{p}_n)]) \ge 1 - \alpha + o(1)$ .

<sup>&</sup>lt;sup>26</sup>Note that, because of the symmetry of the normal distribution, no changes are needed when computing the quantiles in steps 3 and 4.

#### APPENDIX C

### **GMM MOMENT FUNCTIONS**

## C.1 GMM Moment Functions for Assumption 2.3

The moment functions for average baseline characteristics of all the strata is based on the conditional expectation defined by  $\{Z,D\}$ . Let  $\bar{x}_k$  denote the expectation of a scalar baseline variable for a certain stratum k. The moment function for this variable is defined as:

$$g(\{\bar{x}_k\}) = \begin{vmatrix} (x - \bar{x}_{at})(1 - Z)D \\ (x - \bar{x}_{nt})Z(1 - D) \\ (x - \bar{x}_c \frac{\pi_c}{p_{1|1}} - \bar{x}_a \frac{\pi_{at}}{p_{1|1}})ZD \\ (x - \bar{x}_c \frac{\pi_c}{p_{0|0}} - \bar{x}_n \frac{\pi_{nt}}{p_{0|0}})(1 - Z)(1 - D) \\ x - \sum_k \pi_k \bar{x}_k \end{vmatrix}$$
(C.1)

where  $\{\bar{x}_k\} = \{\bar{x}_{at}, \bar{x}_{nt}, \bar{x}_c\}$ . By Law of Iterated Expectation,  $E[g(\{\bar{x}_k\})] = 0$  when evaluated at the true value of  $\{\bar{x}_k\}$ .

## C.2 GMM Moment Functions for Assumption 3.6

Similarly, the moment functions for average baseline characteristics of all the strata in this setting is based on the conditional expectation in each cell defined by  $\{Z, D, S\}$ . Let  $\bar{x}_k$  denote the expectation of a scalar baseline variable for a certain stratum k. The moment function for this variable is defined as:

$$g(\{\bar{x}_{k}\}) = \begin{bmatrix} (x - \bar{x}_{aNN})(1 - Z)D(1 - S) \\ (x - \bar{x}_{aEE})(1 - Z)DS \\ (x - \bar{x}_{nNN})Z(1 - D)(1 - S) \\ (x - \bar{x}_{nEE})Z(1 - D)S \\ (x - \bar{x}_{cEE}\frac{\pi_{cEE}}{p_{01|0}} - \bar{x}_{nEE}\frac{\pi_{nEE}}{p_{01|0}})(1 - Z)(1 - D)S \\ (x - \bar{x}_{cNN}\frac{\pi_{cNN}}{p_{10|1}} - \bar{x}_{aNN}\frac{\pi_{aNN}}{p_{10|1}})ZD(1 - S) \\ (x - \bar{x}_{cNE}\frac{\pi_{cNE}}{p_{00|0}} - \bar{x}_{cNN}\frac{\pi_{cNN}}{p_{00|0}} - \bar{x}_{nNN}\frac{\pi_{nNN}}{p_{00|0}})(1 - Z)(1 - D)(1 - S) \\ (x - \bar{x}_{cNE}\frac{\pi_{cNE}}{p_{11|1}} - \bar{x}_{cEE}\frac{\pi_{cEE}}{p_{11|1}} - \bar{x}_{aEE}\frac{\pi_{aEE}}{p_{11|1}})ZDS \\ (x - \bar{x}_{cNE}\frac{\pi_{cNE}}{p_{11|1}} - \bar{x}_{cEE}\frac{\pi_{cEE}}{p_{11|1}} - \bar{x}_{aEE}\frac{\pi_{aEE}}{p_{11|1}})ZDS \\ x - \sum_{k}\pi_{k}\bar{x}_{k} \end{bmatrix}$$

where  $\{\bar{x}_k\} = \{\bar{x}_{aNN}, \bar{x}_{aEE}, \bar{x}_{nNN}, \bar{x}_{nEE}, \bar{x}_{cNN}, \bar{x}_{cEE}, \bar{x}_{cNE}\}$ . By Law of Iterated Expectation,  $E[g(\{\bar{x}_k\})] = 0$  when evaluated at the true value of  $\{\bar{x}_k\}$ .

Alternatively, it is possible to write the moment function for the proportions of all the strata and then to estimate the model together with the average baseline characteristics simultaneously by GMM. However, such GMM estimators do not behave well in the JC data employed in the dissertation. Thus, in the applications, I first identify the proportions of all the strata, and then estimate all the average baseline characteristics given the identified proportions. In the case of addressing sample selection and noncompliance, as seen in  $g(\{\bar{x}_k\})$ , for each variable, nine equations (eight derived from the conditional expectations defined by  $\{Z, D, S\}$  plus one from the expectation for the entire sample) are used to identify seven means, i.e.,  $\{\bar{x}_k\}$ . Since the standard errors obtained from this GMM model do not take into account the fact that the proportions of the strata are also estimated, I use a 500-repetition bootstrap to calculate the standard errors of the estimated average baseline characteristics.

	Entire Sample			Non-Hispanics		
	cEE	cNE	Diff.	cEE	cNE	Diff.
Female	.396**	.630**	234	.390**	.544**	154
	(.015)	(.165)	(.174)	(.016)	(.134)	(.145)
Age at Baseline	18.44**	19.19**	749	18.39**	19.18**	786
	(.056)	(.699)	(.735)	(.068)	(.592)	(.632)
White, Non-hispanic	.299**	.260**	.039	.369**	.259**	.110
	(.012)	(.126)	(.133)	(.015)	(.126)	(.135)
Black, Non-Hispanic	.445**	.622**	177	.550**	.624**	074
	(.013)	(.158)	(.166)	(.015)	(.136)	(.147)
Has Child	.161**	.229**	068	.151**	.210*	059
	(.011)	(.112)	(.119)	(.012)	(.111)	(.119)
Number of children	.215**	.356*	141	.209**	.280	071
	(.018)	(.187)	(.200)	(.019)	(.179)	(.192)
Personal Education	10.22**	10.34**	123	10.24**	10.27**	036
	(.040)	(.504)	(.529)	(.048)	(.402)	(.434)
Ever Arrested	.230**	.223*	.007	.228**	.292**	064
	(.012)	(.128)	(.136)	(.013)	(.112)	(.121)
At Baseline						
Have job	.241**	.174	.068	.244**	.159	.084
	(.011)	(.108)	(.115)	(.012)	(.102)	(.110)
Weekly hrs. worked	24.07**	25.27**	-1.196	24.05**	25.23**	-1.187
	(.583)	(6.365)	(6.766)	(.613)	(5.768)	(6.160)
Weekly earnings	113.86**	120.08**	-6.219	115.48**	142.57**	-27.09
	(3.987)	(39.90)	(40.90)	(3.500)	(34.31)	(36.51)
Had job, Prev. Yr.	.714**	.585**	.129	.718**	.588**	.130
	(.013)	(.141)	(.151)	(.014)	(.126)	(.136)
Mths. employed, Prev. Yr.	4.346**	3.286**	1.060	4.435**	2.935**	1.500
	(.122)	(1.201)	(1.280)	(.137)	(1.105)	(1.201)
Earnings, Prev.Yr.	3396.2**	3136.2**	260.02	3377.6**	2879.7**	497.88
	(128.63)	(1185.0)	(1250.6)	(128.55)	(1009.5)	(1095.9)

Table C.1: Average Baseline Characteristics for the cEE and cNE Strata

Note: Numbers in parentheses are standard errors. **\*\*** and **\*** denote that estimate is statistically different from 0 at 5% and 10% level, respectively. Computations use design weights. Missing values for each of the baseline variables were imputed with the mean of the variable. The standard error is calculated by a 500-repetition bootstrap.

# C.3 Baseline Characteristic Estimates

To assess Assumption 3.6, Table C.1 shows the average baseline characteristics for the cEE and cNE strata and their difference in Section 4.3. It is obtained by estimating the above GMM function (C.2).